

FRIDAY November 26, 1993

ENVIRONMENTAL FEDERAL REGISTER

Electronic Sub-Set

Note: Agencies currently participating in

this sub-set include: EPA

Contents

Federal Register

Vol. 58, No. 226

Friday, November 26, 1993

Environmental Protection Agency

RULES

Air pollutants, hazardous; national emission standards: State programs approval and Federal authorities delegation, 62262

PROPOSED RULES

Air quality implementation plans; approval and promulgation;

various States:

Colorado, 62304

Delaware, 62307

Illinois, 62309

NOTICES

Environmental statements; availability, etc.:

Agency statements-

Weekly receipts, 62345

Meetings:

Biotechnology issues, agency's current thinking; discussion, 62346

Science Advisory Board, 62346

Pesticide registration, cancellation, etc.:

Air Products & Chemicals et al., 62347

Superfund; response and remedial actions, proposed settlements, etc.:

Beaunit/Circular Knit and Dying Plant Site, FL; correction, 62354

Rock Hill Chemical Co. Site, SC, 62354

CFR PARTS AFFECTED IN THIS ISSUE

A cumulative list of the parts affected this month can be found in the Reader Aids section at the end of this issue.

40 CFR 9	62262
63	62262
Proposed Rules:	
52 (3 documents)	62304
` '	62307 62309

Rules and Regulations

Federal Register

Vol. 58, No. 226

Friday, November 26, 1993

This section of the FEDERAL REGISTER contains regulatory documents having general applicability and legal effect, most of which are keyed to and codified in the Code of Federal Regulations, which is published under 50 titles pursuant to 44 U.S.C. 1510.

The Code of Federal Regulations is sold by the Superintendent of Documents. Prices of new books are listed in the first FEDERAL REGISTER issue of each week.

ENVIRONMENTAL PROTECTION AGENCY

40 CFR Parts 9 and 63

[AD-FRL-4804-7]

Approval of State Programs and Delegation of Federal Authorities

AGENCY: Environmental Protection Agency

(EPA).

ACTION: Final rule.

SUMMARY: The EPA is promulgating regulations to provide guidance, relating to approval of State programs, that the EPA is required to publish under section 112(1) of the Clean Air Act as amended in 1990 (Act). Section 112(1)(2) of the Act requires the EPA to publish guidance useful to States in developing programs for implementing and enforcing emission standards and other requirements for hazardous air pollutants (HAP's) and guidance concerning requirements for the prevention and mitigation of accidental releases of toxic substances into the ambient air. This final rule contains guidance specifically relating to the approval of rules or programs that States can implement and enforce in place of certain Federal section 112 rules, and the partial or complete delegation of Federal authorities and responsibilities associated therewith. Submission of rules or programs by the States under this subpart is entirely voluntary. States seeking to implement and enforce some provisions of their own programs in lieu of federally promulgated hazardous air pollutant standards under section 112 need to obtain approval under this final rule. Once granted approval, State rules and applicable part 70 operating permit conditions resulting from approved State programs would be federally enforceable and would substitute for the otherwise applicable Federal requirements within a State or local jurisdiction.

EFFECTIVE DATE: The guidance announced herein takes effect on December 27, 1993. **ADDRESSES:** *Docket.* Supporting information used in developing the proposed and final rules is contained in Docket No. A–92–46.

The docket is available for public inspection and copying from 8:30 a.m.–12 p.m. and 1:30 p.m.–3:30 p.m., Monday through Friday, at the EPA's Air Docket Section, Waterside Mall, Room M1500, U.S. Environmental Protection Agency, 401 M Street, SW., Washington, DC 20460. A reasonable fee may be charged for copying.

FOR FURTHER INFORMATION CONTACT: For general information on today's final rule, contact Sheila Q. Milliken, Pollutant Assessment Branch, Emission Standards Division (MD–13), U.S. Environmental Protection Agency, Office of Air Quality Planning and Standards, Research Triangle Park, North Carolina 27711, telephone number (919) 541–2625.

SUPPLEMENTARY INFORMATION: The information presented in this preamble is organized as follows:

- I. Background and Purpose
- II. Public Participation
- III. Summary of Final Rule
- IV. Significant Comments and Changes to the Proposed Rule
 - A. Section 63.90—Program Overview
 - B. Section 63.91—Criteria common to all approval options
 - C. Section 63.92—Approval of a State rule that adjusts a section 112 rule
 - D. Section 63.93—Approval of State authorities that substitute for a section 112 rule
 - E. Section 63.94—Approval of a State program that substitutes for section 112 emission
- F. Section 63.95—Additional approval criteria for Federal accidental release prevention programs
- G. Section 63.96—Review and withdrawal of approval
- H. Other Comments
- V. Additional Guidance
- VI. Administrative Requirements
 - A. Docket
 - B. Executive Order 12866
 - C. Paperwork Reduction Act
 - D. Regulatory Flexibility Act
 - E. Review

This preamble provides an overview of criteria and procedures for approval by the EPA of State rules and programs that implement and enforce section 112 of the Act.

The preamble also provides a detailed discussion of the changes made to the proposed regulation. Section I discusses the background and purpose of today's rule. Section II provides information regarding public involvement in the rulemaking during the public comment period following proposal. A summary of today's rule is found in section III which gives a brief overview of the regulatory requirements. A discussion of the significant comments and resulting

regulatory changes from the proposed requirements is detailed in section IV. The discussion of comments and changes to the rule are found in this section in the sequence of the subpart E rule. In the preamble of the proposed rule, the EPA explained the basis for its various proposed positions. Where the proposed regulation has not been changed in the final rule, the EPA continues to rely on the rationale provided in the proposal notice. In addition, clarification or explanation has been included in those places where comments indicated it would be useful. Section V discusses additional guidance required by section 112(1). Finally, section VI covers administrative requirements necessary for promulgation of this rule.

The EPA proposed these regulations to be codified in 40 CFR part 63 on May 19, 1993 (58 FR 29296). The comment period for the proposal ended on July 6, 1993. The EPA received comments from 27 commenters on the proposed rule during the public comment period. The comments have been carefully considered, and where determined to be appropriate by the Administrator, changes have been made in the proposed rule. Copies of these comments appear in the docket for this action.

The major comments and responses are summarized in this preamble. A separate document providing additional responses to comments on the proposal is included in the docket.

I. Background and Purpose

Many States have developed or are developing air toxics programs under State authorities. The Congress was very much aware of the States' air toxics programs in the course of developing the 1990 Amendments. (See, e.g. S. Rep. No. 228, 101st Cong. 1st Sess. 192 (1989) (herein after S. Rep.).) These programs, developed to address specific State needs, may differ from Federal rules being developed by the EPA under section 112 of the 1990 Amendments for the control of emissions of HAP's and other programs. Existing State programs may result in controls that are more stringent than, equivalent to, or less stringent than controls resulting from corresponding Federal standards.

From discussions with States and other interested parties, the EPA has learned that some States want to continue to implement and enforce the requirements of their own air toxics and accidental release prevention programs even though new 1990 Amendments requirements under section 112 relating to hazardous air pollutants will be issued. The prospect of simultaneous implementation and enforcement of both

Federal and State air toxics and accidental release prevention programs in some States has caused concerns to be expressed regarding the possible effects on the States and the regulated community. A primary concern is that section 112 could lead to "dual regulation", a situation in which sources are subject to differing State and Federal program requirements. Dual regulation may burden regulated sources and permitting and enforcement agencies for several reasons. First, permits resulting from dual regulation are necessarily longer and more expensive to develop and approve due to the need to specify separate sets of operating conditions derived from both Federal and State regulations. Second, compliance and enforcement costs may be greater because of two sets of conditions that must be enforced. Third, permit conditions that result from dual regulation may not always be complementary and may even be fundamentally inconsistent in instances where the Federal and State programs may require measures that are technically incompatible. In this latter instance, it may be difficult or impossible for a source to employ simultaneously the control requirements mandated by both Federal and State regulations.

To avoid dual regulation and the attendant complications, as well as to preserve the integrity of their own air toxics and accidental release prevention programs, some States have contended that section 112(l) of the 1990 Amendments authorizes the EPA to delegate authority to the States to implement and enforce their rules or programs in lieu of Federal rules under section 112. Many States have expressed this argument to the EPA through a series of discussions and informal conversations prior to proposal. The EPA agrees that section 112(1) authorizes the EPA to delegate certain section 112 authorities to States. Today's final rule offers guidance intended to assist States (and local agencies) in submitting rules and programs for approval by the EPA. After approval by the EPA, States may implement and enforce their rules and programs in place of certain Federal rules promulgated under section 112, with the EPA approved State rules and programs being federally enforceable. Section 112(1) also provides that any delegation of the EPA's authorities under today's rule shall not include the authority to set standards or other emission limitations or requirements less stringent than those promulgated by the EPA under the 1990 Amendments. The regulation in today's notice, along with guidance for review of high-risk point sources fulfills the requirement for the EPA to publish guidance under section 112(1)(2). See section V of this preamble for further discussion of the highrisk point source program guidance. In addition, today's final rule provides a procedural mechanism for approval and delegation of State requirements that are

exactly as promulgated by the EPA under section 112.

Today's final rule seeks to achieve the goal of allowing the EPA and the States to work together to minimize potential program redundancies and inconsistencies and to reduce the costs and time involved in permit review and issuance. At the same time today's rule will assure that all sources of hazardous air pollutants and hazardous substances listed under section 112(r) meet emission standards and other requirements that are no less stringent than corresponding Federal requirements.

Today's notice also addresses the requirement in section 112(1)(2) that the EPA include as an element of the guidance "an optional program begun in 1986 for the review of high-risk point sources of air pollutants including, but not limited to, hazardous air pollutants listed pursuant to subsection (b)." Pursuant to that provision, the EPA has developed guidance to assist State agencies in establishing a high risk point source program that can work within and beyond the context of section 112. Enabling Guidance to provide further details on the requirements of section 112(1) and information about various technical assistance activities, including an air toxics clearinghouse is published concurrent with promulgation of this rule.

II. Public Participation

A public hearing was held on the proposed rule in Research Triangle Park, North Carolina on June 22, 1993 to provide interested persons the opportunity for oral presentation of data, views, or arguments concerning the proposed rule. This hearing was open to the public, and each attendee was given an opportunity to comment on the proposed rule. The significant changes to the regulations resulting from public comments are described in this preamble. A summary of all public comments and the EPA responses and transcripts of the public hearing are contained in the docket.

III. Summary of Final Rule

Today's final regulations establish guidance for the EPA approval of State (or local, Tribal or Territorial) air toxics control rules (i.e., promulgated regulations) or programs (i.e., any collection of legally enforceable statutes, regulations) that are at least as stringent as otherwise applicable Federal section 112 rules. No State rule or program is federally approved and enforceable unless and until it is approved by the EPA through the full section 112(l) process established in subpart E. After approval, State rules and operating permit conditions (incorporated in a part 70 permit, as applicable,) that result from approved State programs would be federally enforceable and substitute for the otherwise applicable Federal requirements in that State or local jurisdiction.

State and local agencies with approved part 70 operating permit programs have the responsibility under part 70 to begin immediately the implementation and enforcement of all applicable section 112 rules. Authorities granted at the time of part 70 program approval will not by themselves allow for Federal enforceability of a State rule or program that differs in any respect from an existing Federal rule. State rules or programs that differ from the existing Federal rule remain State enforceable until approved under subpart E. Upon the EPA approval of part 70 programs, States may also receive approval under section 112(l) to implement and enforce federal section 112 rules as promulgated for all part 70 sources. Prior to part 70 approval, States seeking delegation of authority to implement and enforce Federal section 112 rules as promulgated for part 70 sources may request approval under subpart

To gain EPA approval of a State rule or program under today's final rule, certain statutory approval criteria contained in section 112 must be met. These criteria require that a submission for approval of a State rule or program must demonstrate adequate authority, adequate resources, an expeditious implementation schedule and an adequate enforcement strategy. In addition, for State rules or programs that differ from Federal requirements, one of three sets of specific criteria must be met to assure adequate stringency. If a State is seeking delegation without changes, these stringency criteria are not necessary. The three sets of specific criteria correspond to three options for requesting approval of such rules or programs: Approval of a state rule that adjusts a section 112 rule, approval of State authorities that substitute for a section 112 rule, and approval of a State program which substitutes for some or all section 112 emission standards. Under the first of these three options, a State rule could be approved that is similar to and at least as stringent as, a Federal rule. The State rule must have undergone a 30-day State notice and public comment period before submission for Federal approval under section 112(1). Under this option, any difference from the Federal rule must have been included in the subpart E list of "adjustments". The Agency believes that those adjustments will result in a rule that is clearly no less stringent than the otherwise applicable Federal rule. There can be no ambiguity regarding the stringency of a rule that differs from the Federal rule by any of the proposed adjustments approved under this option. If the EPA finds that the State request meets the necessary criteria, the State rule with adjustments is approved and becomes Federally enforceable in lieu of the otherwise applicable section 112 rule.

Under the second option, the EPA may approve a State rule (and in certain limited cases, a specific application of broader State authorities) with greater differences from the

Federal rule. This could be the case when a State submits a rule written with a different conformation than a Federal rule or when, for example, a State rule achieves equivalent emission reductions but with a combination of levels of control and compliance and enforcement measures not provided for in the Federal rule.

Under today's final rule, a State must make a detailed demonstration that the State rule results in equal or greater emission reductions (or other measure of stringency such as specified for section 112(r)) for each individual source affected by the Federal section 112 rule. Further discussion of detailed demonstrations can be found in the enabling guidance entitled, "Enabling Guidance for Approval of State Programs and Delegation of Federal Authorities available as described in section V of this preamble. If the EPA finds that the demonstration is satisfactory, subpart A of part 63 would be amended to incorporate the approved State rule. The approved State rule would be federally enforceable and replace the otherwise applicable Federal rule in the relevant State or local jurisdiction.

For requirements for prevention of accidental releases, approval of a State rule which substitutes for the Federal section 112(r) rule must be no less stringent, cover the substances listed pursuant to section 112(r) at or below the threshold quantity, contain accident prevention requirements, facility registration, enforcement provisions which contain an auditing component as well as other measures, and provisions for the disclosure of facility information.

The third option is for approval of a generic State program that substitutes for some or all section 112 emission standards. Under this option, a State program may be approved in place of specific standards and requirements established under sections 112(d), (f), or (h) for incorporation in part 70 permits. For other Federal rules which are not emission standards, for example the requirements of section 112(g), this third option is not available. Rather, approval for State programs with requirements corresponding to Federal requirements other than section 112 (d), (f), or (h) may be sought under options one or two.

For approval under this third option, a State must make a legally-binding commitment to undertake certain actions; the commitment will be adopted under State law. First, the State must commit to regulate every source that would have been regulated by the otherwise applicable Federal section 112 emission standards for which approval is requested. Second, the State must provide assurance that the level of control and compliance and enforcement measures in each part 70 permit for these sources are at least as stringent as those that would have resulted from the otherwise applicable Federal emission standards. Finally, the State must commit to expressing the part 70

operating permit terms and conditions in the form of the otherwise applicable Federal standard. This means that the State must commit to express in the resulting part 70 permit, a level of control in terms of an emission limit, level or reduction, derived from its own program, that is in the same units of measure as the Federal rule and must commit to express other elements of the standard in the same form as the Federal standard. Required compliance provisions must also be in the same form and units of measure as the Federally promulgated compliance provisions. Underlying these commitments is the premise that a State must demonstrate the authority and commitment to permit all of these sources and to require terms and conditions that are no less stringent than would be required under the otherwise applicable Federal standard. If the EPA approves the State program, the EPA would then promulgate a rule amending part 63 to incorporate the State program.

A State may use any one or any combination of these three options in its request for approval of State rules or programs. To illustrate, a State submitting a request under option three, program approval, might not be able to gain approval for regulation of all source categories. In particular, approval under option three may not be granted for area sources which a State has chosen to exempt from part 70 permits. This would not, however, preclude a State from seeking approval under option two of a State rule regulating these area sources.

Regarding the EPA oversight of approved State programs, in receiving approval of a State rule or program, a State has the responsibility to respond in a timely fashion to the EPA requests for information needed to review the adequacy of State implementation and enforcement of an approved rule or program. The EPA will develop guidance for the regular review and intermittent audits of approved State rules and programs.

After approval has been granted, if the EPA finds that an approved rule or program is not being adequately implemented or enforced, the EPA has the authority to withdraw approval of that rule or program. Before approval is withdrawn, however, the State has the opportunity to correct the deficiencies identified in the EPA's review or audit. The EPA would inform the State of changes that need to be made and, if the State does not take adequate action to correct the deficiencies, a public hearing would be held and public comment accepted. The State would then have 90 days to correct the situation. After this process has taken place, if the State does not correct the identified deficiencies, the EPA would then withdraw approval of the rule, the program or part of the rule or program. Upon withdrawal of approval of a State rule or program that is found to be less stringent than Federal requirements, States would be required to

reopen part 70 operating permits according to the provisions in § 70.7(f) and rewrite permit conditions to reflect requirements of the applicable Federal section 112 rule. The federally promulgated section 112 standard is the applicable and federally enforceable standard unless and until a State rule or program is approved by the EPA pursuant to the procedures set forth in this final rule. Once approved, the State rule or program becomes the applicable standard which the EPA has authority to enforce, and the federally promulgated standard is no longer the applicable or enforceable standard. Upon withdrawal of approval of a State rule or program, the federally promulgated standard for which the State rule or program substitutes once again becomes the applicable standard. In the withdrawal notice, the EPA will establish an expeditious schedule for sources to come into compliance with the federally promulgated standard.

Under $\S\S 63.96(b)(4)(v)$ and 63.96(b)(7)(iii), which address withdrawal of approval of State programs either by the EPA or voluntarily by the State, the final rule states that the EPA has authority to enforce the applicable section 112 requirement. This authority is a restatement of section 112(1)(7), which provides that nothing shall prohibit the EPA from enforcing any applicable emissions standard or requirement under section 112. The EPA always has concurrent authority to enforce the applicable section 112 standard, which may be either an approved State standard or a Federal standard, depending upon whether the State standard has been federally approved pursuant to the procedures set forth in this final rule.

Today's rule also provides guidance on the approval of State Accidental Release Prevention (ARP) Programs established under section 112(r). The section 112(r) (3)–(5) "list and threshold" rule was proposed in January 1993 (58 FR 5102). A proposed risk management program rule under section 112(r)(7) was proposed in October 1993.

In order to receive approval and delegation for an ARP program which differs from the Federal section 112(r) rules, a State submission must meet the criteria set out in § 63.91, either § 63.92 or § 63.93, and § 63.95. For approval of State rules or programs to implement and enforce the Federal accidental release prevention program as promulgated without changes, the requirements of this section and § 63.95 must be met.

A State program must demonstrate the authority and resources necessary to implement and enforce regulations which authority covers the regulated substances at or below the thresholds, the accidental release prevention requirements, as well as identify the entity that will be receiving the registration from regulated sources.

In addition, the State submission must include a description of the procedures for registration of sources, receiving and reviewing risk management plans, making the

plans available to the public, and the coordination mechanism the implementing agency will use with the Chemical Safety and Hazard Investigation Board, the State Emergency Response Commission, the Local Emergency Planning Committees, and the air permitting program (if it is not responsible for implementing section 112(r) in the State).

States do have the option of requesting a complete or partial program. Partial delegation in terms of the ARP program here refers to geographic area. This allows delegation of section 112(r) to local agencies, provided that the entire area of the State is subject to the requirements under section 112(r). The Agency believes that the ARP program should not be subdivided into various components based on chemical or industry because this would promote confusion for industry and inhibit the integration of the ARP program into State wide activities. Further, any delegation of the ARP program requires the State program to contain a set of core requirements for all subject sources. This is consistent with the requirements in section 112(1)(5)(A) that requires an approved State program to contain the authorities "to assure compliance by all sources within the State with each applicable standard, regulation, or requirement established by the Administrator under this section." Section 63.95 sets out the core requirements for an approvable State ARP program.

IV. Significant Comments and Changes to the Proposed Rule

This portion of the preamble is organized by each section in subpart E, and discusses the principal regulatory changes made in the final rule in response to public comments. It also discusses some comments that did not result in regulatory changes.

A. Section 63.90—Program Overview

This section provides a brief overview of the subpart. It also establishes subpart definitions, outlines local agency roles and enumerates authorities to be retained by the Administrator.

In response to comments received, and to provide for approval of State programs under additional circumstances, the Agency has amended this section to provide for approval of State rules or programs to implement and enforce Federal section 112 rules without change as promulgated by the EPA. Therefore, this section now provides a mechanism for delegation of Federal standards prior to approval of a State's part 70 operating permit program and for Federal section 112 requirements for sources that are not subject to the requirements of part 70. A State seeking approval for programs to implement and enforce Federal section 112 rules must meet the criteria of section 112(1), as specified in this section, including the requirement for notice and opportunity for public comment. Procedural mechanisms for

delegation will be addressed in the Enabling Guidance, available as described in section V of this preamble.

Part 70—Approval and Delegation Without Changes

One commenter noted that approval of a State part 70 program would not include a review of resources needed to cover the cost of bringing enforcement actions under section 502(b) and yet this cost must be included in an adequate demonstration of resources before approval under this subpart. In addition, the commenter argued that the part 70 program approval process will not be adequate to assure section 112(l)(5) criteria are met before delegating under section 112(l).

The Agency disagrees with these comments. The Agency notes that approval that occurs under any of the three part 63 subpart E options for approving changes to the Federal program will examine these costs in a specific resource review during the approval process under subpart E. The EPA maintains that program review under part 70 will satisfy the adequate resource criterion under section 112(1) and that the section 112 program requirements may be delegated to the States without changes.

Part 70 requires a demonstration that a State has authority to adequately administer and enforce the part 70 program. Several provisions of the part 70 regulations ensure this type of demonstration. For example, the State must demonstrate under § 70.4(b)(3)(vii) adequate authority to enforce all permit terms and conditions and the requirements of the permit program consistent with the civil and criminal authority required by § 70.11. States must also submit pursuant to § 70.4(b)(4)(ii) all relevant guidance used in implementing the program, including criteria for monitoring source compliance such as inspection strategies, and pursuant to § 70.4(b)(5), a complete description of the State's compliance tracking and enforcement program.

States must also submit a detailed statement that adequate personnel and funding have been made available to develop, administer and enforce the program under § 70.4(b)(8). Finally, States are required to annually submit detailed information regarding the State's enforcement activities under § 70.4(b)(9). In addition § 70.6 (a) and (c) require all permits to contain sufficient monitoring, recordkeeping, reporting and compliance certification requirements to ensure that the permit terms and conditions may be adequately enforced.

The commenter is correct in that section 502(b)(3)(A)(ii) does not require a State to collect permit fees to cover "any court costs or other costs associated with any enforcement action." However, this does not mean that the State does not have to adequately enforce the terms and conditions of permits, including bringing judicial

enforcement action where necessary. Rather, it means that the State is not required under title V to collect fees to cover the actual court costs of such enforcement actions. Under section 112(1)(5), the State demonstration must show that adequate resources to implement the program are available; the EPA believes that the requirements under part 70 will meet this requirement as applicable.

In addition, the commenter noted that citizens have not been provided adequate notice that States may receive delegation of section 112 based on their part 70 operating permit program because some States have already begun preparation for submittal of part 70 programs.

The ÉPA disagrees with this comment. In order to obtain approval of a part 70 operating permit program, adequate resources and authority must be demonstrated. In addition, the part 70 operating permits rule provides that States write permits including "all applicable requirements". Part 70 defines applicable requirement to include "(a)ny standard or other requirement under section 112 of the Act." Clearly, this constitutes adequate notice of intent that section 112 requirements must be included in part 70 programs, and this was included in the part 70 regulations when promulgated.

In addition, during the part 70 permit issuance process, "* * * any person may petition the Administrator to veto a permit * * *." § 70.8(d). The objections in the petition must have been previously raised during the public comment period on the permit provided by the State issuance process, unless the petitioner shows that it was impracticable at that time. This provides an additional opportunity for comment on incorporation of particular section 112 requirements in an individual permit.

Clarification of State Rights Under Section 116 and Section 112(d)(7)

Some commenters questioned whether the Act provides authority for the EPA to approve more stringent State standards that are not based on the same considerations the EPA must include when it establishes standards under section 112, for example the cost of achieving emission reduction and any non-air quality health and environmental impacts and energy requirements. See section 112(d)(2). In addition, commenters pointed out that EPA may not approve, under § 70.1(c), State programs that are inconsistent with the Act.

The Agency recognizes the complex interactions that are the consequence of regulation of a community of sources by both the State and Federal governments and that accompany any division of responsibility in such a joint effort. From its inception, the Act has been based on a strategy of air pollution prevention and control at its source that recognizes the States and local governments as bearing the primary responsibility for such prevention and control. Clean Air Act section

101(a)(3). By enacting section 116, Congress also recognized that States and local governments, in responding to concerns within their own jurisdictions, might desire to control air emissions more stringently than would be required by the Federal government on a nationwide basis and might therefore require more stringent limitations on emission of air pollutants from sources within their State. Section 116 explicitly allows such State standards and limitations as long as they are no less stringent than the corresponding Federal standard or limitation.

In enacting the 1990 Amendments, which require the EPA to establish standards for emission of hazardous air pollutants, Congress was aware that many States had already developed active and effective air toxics programs. S. Rep. at 149. Much of the development of these State programs had occurred with the support and encouragement of the EPA, and Congress recognized that existing State programs were a significant component of the nationwide air toxics control strategy. In addition, the preamble to the final rule establishing the part 70 operating permit program recognized that minimizing disruption of existing State programs is an important goal of the Agency's implementation of the Act. (57 FR 32350, 32251, 32263, 32265, 32273 (1992).)

In establishing requirements under the 1990 Amendments, Congress included under the provisions of section 112 a mechanism by which States could seek approval of their air toxics programs and established criteria for determining whether or not a State program was approvable. Section 112(1)(5) requires that the Administrator disapprove a State program if, among other things, the Administrator determines that "the authorities contained in the program are not adequate to assure compliance by all sources within the State with each applicable standard, regulation or requirement established by the Administrator under this section." Section 112(l)(1) requires that a program submitted by a State "shall not include authority to set standards less stringent than those promulgated by the Administrator" under the Act. Therefore, State standards and requirements must be at least as stringent as corresponding Federal standards and requirements.

In addition, section 112(d)(7) reinforces the authority of States to issue standards under State authority specifically in the area of air toxics control. No section 112 standard or other requirement is to be interpreted, construed or applied to diminish or replace the requirements of a standard issued under State authority. Since section 116 precludes a State from adopting or enforcing less stringent standards than those under section 112, section 112(d)(7) thus prohibits interpreting, construing, or applying section 112 standards or requirements to diminish or replace State standards if they are no less stringent than section 112 standards.

The part 70 operating permits program regulations also provide for no less stringent State requirements. Section 70.1(c) states that nothing in part 70 shall prevent a State from establishing additional or more stringent requirements not inconsistent with the Act. In addition, § 70.1(c) also states that no permit can be less stringent than necessary to meet all applicable requirements. Section 70.6(b)(2) requires a State to identify any permit terms and conditions that are not required under the Act or under any of its applicable requirements, and thus States may establish more stringent State-only standards for incorporation in that section of the operating permit. The 1990 Amendments section 506 authorizes States to establish additional permitting requirements as long as they are not inconsistent with the Act, and States are free to establish more stringent permit revision procedures provided the minimum requirements of part 70 are met. 57 FR 32250, 32284 (1992).

Thus, States may establish State requirements, as long as they are no less stringent than corresponding Federal requirements, and may incorporate those requirements into part 70 operating permits according to the requirements of part 70. In addition, section 112(1) places no restrictions on the stringency of approvable State standards, other than that they may not be less stringent than corresponding Federal standards, nor does section 112(1) require consideration of any particular factors in development of an approvable State standard.

Federal Enforceability

Several commenters questioned the basis for the EPA's determination that a State rule or program, once approved according to the requirements of section 112(l), resulted in approved State standards and emission limitations that were federally enforceable. Other commenters requested explanation as to the EPA's delegation authority under section 112. One commenter stated that the EPA's delegation of authority would be unconstitutional under the Appointments Clause of the United States Constitution.

Prior to the enactment of the 1990 Amendments, the Administrator was authorized to delegate her authority to implement and enforce standards promulgated under section 112. When this delegation occurred, a Federal Register notice was published and the delegation authority cited in the Code of Federal Regulations. Delegation procedures were spelled out in an EPA publication, "Good Practices Manual for Delegation of NSPS and NESHAPs". Duplicate delegation authority for new source performance standards resided in section 111 and was unchanged by the 1990 Amendments section 111(c). In the 1990 Amendments, Congress chose a new mechanism for delegation of EPA's authority under section 112, by adding provisions for approval of State programs to the delegation of authorities

and responsibilities that had been present in the pre-1990 section 112. See S. Rep. at 196. The provisions for approval under section 112(l) indicate Congress's view of a dramatically expanded role for the States in regulation of air toxics. For example, Congress recognized that section 112(l) authorities will greatly expand the role of State agencies and stated that "the legislation significantly expands the statutory role for State and local air pollution control agencies in the regulation of air toxics." S. Rep. at 149, 192. In addition, Congress expressly recognized the effectiveness of existing State programs in control of air toxics, e.g., 136 Cong. Rec. S16978 (daily ed. Oct. 27, 1990) (Clean Air Conference Report, Air Toxics); 136 Cong. Rec. S519-20 (daily ed. Jan. 30, 1990) (statement of Sen. Durenberger).

As enacted under the 1990 Amendments, section 112(l) authorizes the Administrator to approve State programs for control of hazardous air pollutants and for prevention and mitigation of accidental releases if the State program meets certain criteria, which are specified in section 112(l)(5).

These criteria require the State program to contain adequate authorities to assure compliance by all sources within the State with each applicable standard, regulation or requirement established by the Administrator under section 112; adequate authority and resources to implement and enforce the program; and an expeditious schedule for implementing the program and assuring compliance by affected sources. Section 112(1)(5)(A)–(C). The program must be in compliance with the guidance issued under section 112(1)(2) and can not be unlikely to satisfy, in whole or in part, the objectives of the Act. Section 112(1)(5)(D). In addition, the program may not include authority to set standards less stringent than Federal standards promulgated under the Act. Section 112(1)(1). Activities under section 112(l) are subject to the provisions of savings clauses for enforcement of section 112 standards and requirements, section 112(1)(7), and authorities and obligations of the Administrator and the State under title V, section 112(1)(9). However, section 112(1) does not directly address the issue of Federal enforceability of State air toxics standards.

Provisions regarding Federal enforcement of section 112 requirements are specified in section 113. In particular, section 113(a)(3) provides for enforcement of any "requirement or prohibition of (title I, including section 112), including, but not limited to, a requirement or prohibition of any rule, plan, order, waiver, or permit promulgated, issued, or approved under (title I, including section 112)." This language was added by the 1990 Amendments, which generally broadened enforcement authorities under section 113. S. Rep. at 358–66.

Under the pre-1990 Amendments, more stringent State standards were not Federally enforceable, since the statute provided for

enforcement only of violations of section 112(c), which clearly applied only to standards promulgated by the EPA. The statute as amended in 1990 does not by its own terms prohibit the violation of State hazardous air pollutant emission standards, and thus, such standards are Federally enforceable only if they constitute a "rule, order, waiver or permit promulgated, issued, or approved" under the Act, that is if such State hazardous air pollutant emission standards included within or adopted into an approved program are "promulgated, issued, or approved" under the Act.

There is no doubt that State standards are "emissions standards" that the State must implement and enforce under section 112(l)(1). Section 302(k) defines "emission standard" to mean "a requirement established by the State or the Administrator which limits the quantity, rate, or concentration of emissions of air pollutants on a continuous basis * * *." Moreover, Congress expressly acknowledged State authority to set emission standards under section 112(l) programs. Section 112(l) prohibits States from submitting programs that include "authority to set standards less stringent than those promulgated by the Administrator" under the Act. This formulation implies that States have authority to set more stringent standards. If they lacked such authority, the prohibition would be unnecessary. While section 112(1)(1) permits States to submit programs for the "implementation and enforcement" of emission standards and other requirements, it does not provide guidance as to the scope of applicability of these State programs. Section 112(l) does not specify what is meant by "partial or complete delegation" of the EPA's authorities and responsibilities and does not provide guidance as to the relationship between existing State standards, previously encouraged and supported by the EPA, and newly-promulgated Federal standards. In particular, it does not provide explicitly for the approval of State emission standards. Because the statute is ambiguous regarding the question of Federal enforceability of approved State standards, the EPA must consider Congress's objectives and policy goals in enacting section 112(1), as well as the overall purposes of the Act. Through this rulemaking the EPA is therefore interpreting the provisions of section 112(1) to authorize approval of State programs and rules that are federally enforceable under section 113. Under the two-step analysis of Chevron U.S.A. Inc. v. Natural Resources Defense Council, 467 U.S. 837 (1984), if Congress has not "directly spoken to the precise question at issue," and if the statute is silent or ambiguous on the issue, then a regulation must be based on a "permissible construction of the statute." Id. at 843. When "competing Congressional goals" are encompassed within the statutory scheme, the

EPA may reconcile these with "a reasonable accommodation of (the) differing policy objectives." Natural Resources Defense Council v. EPA, 822 F.2d 104, 117 (D.C. Cir. 1987). The EPA's interpretation must be "reasonable and consistent with the statute's purpose." Chemical Mfrs. Ass'n v. EPA, 919 F.2d 158, 162-63 (D.C. Cir. 1990). Section 101(a) of the Act recognizes that air pollution prevention and air pollution control at its source is the primary responsibility of States and local governments. In the area of air toxics regulated under section 112, Congress clearly supported States' past efforts to regulate air toxics sources, recognized that some State programs had been developed earlier than the Federal program, and provided a new Federal regulatory scheme under section 112 that included a mechanism for States to maintain existing requirements as long as they were no less stringent than Federal requirements and the State met program approval criteria.

In response to comments, the EPA has added § 63.90(d), Federally enforceable requirements, to clarify that approved rules and requirements are enforceable by the Administrator and citizens under the Act.

However, in exercising EPA's enforcement authority, the Agency would direct its resources towards the provisions of the approved State rule, program, and resulting permit conditions implementing the rule, which the Agency relied on in determining that the State rule or program assured compliance with the Federal requirements. In deciding whether to bring an enforcement action, the Agency would take into account the extent to which any violations implicate the control levels or compliance measures required by the otherwise applicable Federal rule, where the source's compliance with the otherwise applicable Federal standard can be determined. For example, if the Federal standard required a control efficiency of 95 percent and the State rule approved pursuant to section 112(1) required a control efficiency of 99 percent, EPA, in deciding whether to bring an enforcement action, would consider whether the source met the 95 percent control level. If EPA determines that the source was operating equipment that achieved a control level of 96 percent, EPA would not intend to take action against the source for violation of that element of the State rule.

The EPA does not intend to bring an enforcement action against any source covered by a State rule that would not have been covered by the Federal rule. In cases where an alleged violation does not implicate a control requirement of the otherwise applicable Federal rule, EPA will defer to the State to exercise its own enforcement authorities to enforce the more stringent provisions of the approved State rule.

Regarding the constitutionality of delegation under section 112, the EPA notes that delegation of authority is a well-established practice that has long provided a

mechanism for States and local governments to carry out certain provisions of Federal mandates, as authorized by Congress, according to specific criteria and standards, and as overseen by the delegating Federal agency. For example, under the Act, Congress has expressly granted the EPA authority to delegate to a State in section 111(c)(1) (new source performance standards), section 112(l)(1) (hazardous air pollutant and accidental release prevention requirements), and section 328(a)(3) (outer continental shelf activities). Delegation of authority from Federal agencies to State or local governments, including delegation of authority under the Act, has been upheld by the courts. Southern Pacific Transp. Co. v. Watt, 700 F.2d 550 (9th Cir. 1983) cert. denied 464 U.S. 1064 (1984); Nance v. Environmental Protection Agency, 645 F.2d 701, 714-15 (9th Cir. 1981) cert. denied sub nom Crow Tribe of Indians v. Environmental Protection Agency, 454 U.S. 1081 (1981); United States v. Matherson, 367 F.Supp. 779, 782 (E.D.N.Y. 1973) aff'd without op. 493 F.2d 1399 (2d Cir. 1974).

Stringency

Several commenters requested clarification of the measure of stringency that the EPA would use to determine whether a State program or rule was approvable. Commenters asked whether stringency would be measured by emissions reductions and whether the comparison would be made at the emission point, source, or facility level.

As explained below, stringency may be measured by level of control as expressed by emissions reductions, applicability as to the sources subject to requirements, compliance and enforcement measures, such as averaging times, or other measures as determined by the Administrator. Simply put, comparison is made at the point at which the Federal requirement is determined, so that if the Federal requirement is a requirement at the source, so too must the approvable State requirement be at the source; and if the Federal requirement is placed on an emission point, the State requirement must do the same.

In the general description of State Programs under section 112, section 112(1)(1) describes programs that States may develop and submit for approval by the EPA. Section 112(l)(1) prohibits State programs that include "authority to set standards less stringent than those promulgated by the Administrator" under the Act. This prohibition against standards less stringent than Federal standards implies a comparison between the State requirements and the corresponding Federal requirements. Under section 112(1)(5), the EPA must disapprove a State program if the program's authorities are not adequate to assure compliance by all sources within the State with each applicable standard, regulation or requirement

established by the Administrator under section 112.

Taken together, these requirements provide that State programs maintain, as a minimum, all Federal standards, regulations, and requirements as established by the Administrator and ensure that any corresponding State requirements are at least as stringent. Thus, an approvable State standard could not allow, for example, less emissions reductions than the corresponding Federal standard, and the emission reductions would be measured according to the requirements of the Federal standard, that is, if the Federal standard measured emissions reductions at each emission point, a State standard would have to do likewise to be approvable.

Several commenters also felt that the basis for stringency should not be restricted to emission reductions but should instead focus on the impacts that result from emissions. Also, a commenter noted that the basis for stringency should be in accordance with the criteria for establishment of regulatory requirements, e.g. cost, non-air quality health, environmental, and energy impacts of section 112(d) standards.

The EPA recognizes that several provisions under section 112 examine specific impacts to human health and the environment and call for future regulation based on such impacts. The central basis for section 112, however, is the maximum achievable control technology (MACT) program that mandates the installation of controls and the reduction of emissions of listed HAP's regardless of proof of specific resulting impacts. While a reduction of the impacts of HAP emissions to human health and the environment are the central objectives of section 112 of the 1990 Amendments, Congress based the establishment of MACT stringency on "reduction of emissions" and "emission limitation." (See sections 112(d)(2) and 112(d)(3).) This then should also be the primary basis for determining the stringency of State rules and programs to be approved in lieu of Federal section 112 rules. Further, the EPA believes that reliance on a comparison of impacts would be extremely difficult and resource-intensive and such analysis might often require approval decisions in the face of large degrees of uncertainty. Section 63.90 defines "stringency" to be measured by the quantity of emissions or by parameters relating to rule applicability, level of control, and compliance and enforcement measures, or as otherwise determined by the Administrator. Thus determinations that State rules are no less stringent than corresponding Federal rules will typically compare the parameters in the State and Federal rules using the definitional measures or as mandated by a particular Federal standard.

The EPA wishes to clarify that, where Federal emission limitations are expressed as an aggregate total, or as a total of an

aggregate grouping (for example total volatile organic compounds), the stringency comparison is made based upon the aggregation that is identified in the Federal rule. For example, for an emission limitation in a Federal rule expressed as a given pounds per hour of total HAPs, the stringency comparison in a section 112(1) submittal would be made on a total HAP basis. To clarify this point, the definition of "level of control" has been changed in the final rule to explicitly address such situations where the Federal rule provides for emission limitations on an aggregate basis. The change to this definition also reflects a requirement to ensure that, when such aggregate comparisons are made, there would not be an increase in public health risk.

One commenter also indicated that the EPA had not articulated the basis for determining the equivalency of State ARP programs. Since the ARP program is not necessarily based on emissions, determining equivalency based on potential emission reduction can not be done. These requirements are structured as a performance based standard and provide considerable flexibility to the regulated community in terms of compliance. Consequently, State programs may contain different requirements for accidental release prevention which are at least as stringent as the Federal requirement and that may be approvable under the criteria in this rulemaking.

Challenging Mechanism

Several commenters sought clarification of judicial review provisions for approval of State rules and programs. Another commenter asked whether approved State rules and programs would be subject to challenge in State or Federal court.

The EPA will look to the provisions of section 307 of the Act regarding judicial review of this rulemaking and of rulemakings for approval of State rules or programs under subpart E. Challenge to State rules when enacted by the State would be under the requirements of State law. However, approval of State rules or programs under subpart E will be a Federal rulemaking and thereby will be subject to the provisions of section 307.

Public Notice and Comment

One commenter stated that approvals, particularly approvals under § 63.92, should be subject to section 307 notice and comment rulemaking.

Section 307 provides for administrative proceedings and judicial review under the Act. Section 307(d)(1) lists actions to which section 307(d) rulemaking procedures apply, and publication of guidance under section 112(l)(2) is not among them. Although the Administrator may determine under section 307(d)(1)(U) that otherwise unlisted actions are subject to section 307, the Administrator has not done so here. Therefore, the rulemaking provisions of the Administrative

Procedures Act rather than those of section 307(d) are the relevant provisions for this section 112(1) rulemaking. The commenter noted that approvals under section 112(l) effectively constitute promulgation or revision of section 112 standards and are therefore subject to notice and comment requirements. Although the commenter is correct that section 307 requires notice and opportunity for comment for revisions of certain section 112 standards, the EPA believes that Congress's specific provision for notice and comment under section 112(1)(5) rather than the provisions of section 307(d) guides the procedures required under section 112(l). The EPA notes that approvals under section 112(1) are not national in scope like those listed in section 307(d) but are instead limited to a State or local area. Moreover, a State's request for approval may include State standards corresponding to section 112(r), section 112(h), or other section 112 standards that are not listed in section 307 at all, and Congress nowhere indicated that different procedures should be followed depending on the particular section 112 standard for which the State was seeking approval.

Section 112(1)(5) contains procedural requirements that include a requirement for notice and comment. The EPA has revised § 63.91 to clarify that requests for approval, including requests for delegation of unchanged Federal standards, are subject to the notice and comment requirements of section 112(l)(5). Once a State's initial request has been approved, the notice and comment provisions of §§ 63.91, 63.92, 63.93, or 63.94 apply. In the case of requests under § 63.92, i.e. requests for adjustments that are unequivocally no less stringent than the otherwise applicable Federal standard, today's rulemaking along, with the notice and opportunity for comment at the time of the State's initial request fulfills the notice and comment requirement under section 112(1)(5). Because the EPA has determined in this rulemaking, which has provided notice and opportunity for comment, that each of the listed adjustments is unequivocally no less stringent, and because at the time of the State's initial request, the EPA will evaluate the State's program to ensure that it meets the requirements of section 112(1)(5), the requirement for notice and opportunity for comment will be fulfilled both for determination of stringency and for determination of adequacy of the State's program.

Delegation

One commenter noted that the term "'delegation'' should be more clearly defined to explain how it relates to Federal enforceability.

Delegation under section 112(1) means the transfer of authority from the Administrator to a State, according to certain criteria and standards, to implement and enforce the rules or programs approved according to the

requirements of section 112(1). Once approved under the provision of section 112(l), a State rule or program is federally enforceable, which means that the Administrator can enforce the approved State rule or program in Federal court. The State may also enforce approved State standards in State court under State law. In addition, with the exception of requirements designated in the permit as State-enforceable only, and terms and conditions of an approved State rule or program, must be incorporated in the Federally enforceable section of a part 70 permit, and are enforceable according to the provisions of part 70. State law may determine the actual mechanism by which delegation occurs and by which requirements are incorporated in part 70 permits. Also, delegation of authority may occur according to requirements under State law for sources not subject to the requirements of part 70.

Adding Pollutants

In response to EPA's solicitation of comment regarding delegation of authority to add to the list of pollutants under section 112(b), many commenters expressed a view that this was not authorized under section 112(l). Similar comments were received regarding delegation of authority to regulate substances beyond those listed under section 112(r) and to modify the list of source categories under section 112(c). Other commenters feared adverse effects on State programs that contain pollutants other than those specified in section 112(b) if the EPA did not delegate authority under section 112(l) for regulation of additional pollutants.

The EPA notes the many comments regarding delegation of authority to regulate additional pollutants and substances under section 112 (b), (c), and (r). Some commenters noted that these sections contain procedures under which the Administrator may revise the list of pollutants, substances, or source categories and that these procedures are the appropriate mechanism for changes to the lists. The EPA has carefully considered the comments received on this issue and has chosen not to revise the proposal as to delegation of these authorities. Therefore, the EPA retains its authority and will not delegate the authority to add or delete pollutants from the list of hazardous air pollutants established under section 112(b), the authority to add or delete substances from the list of substances established under section 112(r), or the authority to delete source categories from the Federal source category list established under section 112(c)(1) or to subcategorize categories on the Federal source category list after proposal of a relevant emission standard, as was specified in the proposal in § 63.90(c).

The Agency notes that Congress recognized that many State programs prior to enactment of the 1990 Amendments addressed many more pollutants than those finally listed under section 112(b). In fact, Congress explicitly provided for support of

State programs for additional pollutants in requiring the EPA to include the high-risk point source program as an element in the guidance to be published under section 112(l)(2). The EPA is publishing this guidance for high-risk point source programs, available as described in section V of this preamble. Congress also provided for technical assistance and grants, which may include support for high-risk point source review. Section 112(1)(4). These mechanisms provide additional support for broader State programs that address pollutants other than those listed under section 112(b), without requiring approval of State standards for additional pollutants.

B. Section 63.91—Criteria Common to All Approval Options

This section describes the basic process for approval under this subpart, criteria which must be met for all three approval options and discussion of the process employed when previously approved State authorities are later revised.

The EPA has revised this section to incorporate procedures for approval of State programs that contain section 112 rules exactly as promulgated by the EPA. States are likely to seek these approvals prior to receiving approval of their part 70 operating permit programs or for sources not subject to part 70, such as deferred or exempt sources.

In addition, in response to comments received, the EPA has revised this section to delete the reference to a determination by the EPA of whether a State rule or program is likely to satisfy the objectives of the Act in whole or in part. This reference has been deleted because it is not a criterion for approvability to be included in this subpart. Section 112(1)(5) provides that the Administrator must disapprove a State program if the Administrator determines that the program is not in compliance with the guidance issued under section 112(1)(2), that is subpart E, or the program is not likely to satisfy, in whole or in part, the objectives of the Act. Therefore, since the determination as to satisfying the objectives of the Act is separate and distinct from the requirement to comply with subpart E, the EPA has deleted the reference from this section.

The EPA's evaluation of a State's request for approval will necessarily ensure that an approved program is not inconsistent with the objectives of the Act. Consideration of consistency with objectives of the Act is a qualitative judgement implicitly incorporated in the EPA's overall determinations, not only for approval of State programs under section 112(1) but in other determinations that the EPA must make as well, rather than a separate criterion for approval under the guidelines of section 112(l)(5) (A), (B), and (C) and the regulations as promulgated here. The EPA would not and will not approve a State program that is not likely to satisfy, in whole or in part, the objectives of the Act.

Timing for Approvals

Several commenters felt that the 180 days that the EPA is allowed by the statute to approve or disapprove a State rule or program is unnecessarily long.

Submissions for approval under §§ 63.93 and 63.94 require evaluation of the State's submittal and a determination as to the stringency of the State rule or program, as well as notice and opportunity for public comment and a careful consideration by the EPA of those comments prior to approving or disapproving a State submittal. The EPA, therefore, finds 180 days to be an appropriate period to consider State requests made under these sections. As explained in the previous section on public notice and comment, under § 63.92, additional public comment beyond public comments at the State level for each rule submitted under § 63.92 and public comment on this subpart E rulemaking will not be taken. For this reason, the EPA is committing to grant requests for approval under this section within 90 days. Shortening this period for approval will result in less uncertainty for sources and States affected by a request for approval under this section. The rule has been revised accordingly to reflect this shorter review period.

Part 70—Approval as a Precondition for Section 112(1) Approval

Several commenters noted that approval of a State's part 70 program should not be a precondition for approval of a State's request under section 112(1).

Under § 70.4, States must submit to the Administrator a proposed part 70 operating permit program. Elements of the initial program submission are specified in § 70.4(b) and include a complete program description; regulations that comprise the permitting program; a legal opinion from the State Attorney General that the laws of the State provide adequate authority to carry out all aspects of the program including all applicable 112 requirements; a complete description of the State's compliance tracking and enforcement program; a demonstration that permit fees required by the State program are sufficient to cover per program costs; a statement that adequate personnel and funding have been made available to develop, administer, and enforce the program; a commitment from the State to submit information regarding the State's enforcement activities; provisions for adequate, streamlined, and reasonable procedures for expeditious review of permit revisions or modifications; and other information.

Under section 112(l)(5), the Administrator must disapprove a State's program if she determines that the authorities contained in the program are not adequate to assure compliance with each applicable standard, regulation, or requirement established by the EPA under section 112; adequate authority does not exist or adequate resources are not

available to implement the program; the implementation and compliance schedule is not sufficiently expeditious; or the program is not in compliance with the guidance issued under section 112(1)(2) or is not likely, in whole or in part, to satisfy the objectives of the Act.

As outlined above, the information which must be submitted by a State under part 70 encompasses the information required under section 112(1)(5) for approval of State programs that seek only to implement and enforce Federal standards exactly as promulgated. Moreover, the EPA's exercise of its oversight functions under part 70 will help ensure that a State with an approved part 70 program will continue to meet the criteria in section 112(l)(5) for sources subject to the requirements of the part 70 program. Therefore, duplicate applications for such programs would be unnecessary and redundant for any sources a State will permit under part 70. States will need to receive delegation of authorities to implement and enforce section 112 rules and this delegation may take place according to the provisions of the EPA guidance entitled, "Enabling Guidance for Approval of State Programs and Delegation of Federal Authorities".

Part 70 approval also confers approval under section 112(1) for delegation of unchanged Federal standards because part 70 approval suffices to satisfy section 112(l) approval requirements for unchanged section 112 standards. Requirements for part 70 approval, specified in 40 CFR 70.4(b), encompass section 112(l)(5) approval requirements for delegation of section 112 standards as promulgated by the EPA. Section 112(1)(5) requires that the State's program contain adequate authorities, adequate resources for implementation, and an expeditious compliance schedule, which are also requirements under part 70. Therefore, for part 70 sources, part 70 approval also constitutes approval under section 112(1)(5) of the State's programs for delegation of section 112 standards that are unchanged from Federal standards as promulgated.

An approval action under the provisions of part 63 may in fact include the actual delegation of existing standards. It will not, however, include delegation of future standards. Rather, it will make provisionsfor example, as structured in an accompanying MOU—for the delegation of future standards. Such an MOU may allow for automatic delegation, or case-by-case delegation, or automatic delegation except for certain standards, such as the radionuclides standard, or other delegations as appropriate. The provisions of individual approvals and MOUs will differ depending on the authority the State has under State law to accept delegation.

In addition, States may seek approval of State programs prior to receiving approval of their operating permit programs under part 70. In this instance, States must fulfill the requirements of § 63.91, including requirements for notice and comment, even for programs containing only Federal standards exactly as promulgated by the EPA or containing only adjusted rules under § 63.92. The EPA has revised the provisions of § 63.91 to reflect this requirement.

Delegation of section 112 standards is subject to the requirements of section 112(l). Procedurally, implementation of section 112(l) requires submittal of a request for approval, notice in the **Federal Register** that the EPA has received a request, a public comment period of at least 30 days, and notice in the **Federal Register** that the EPA has approved or disapproved the request.

Newly promulgated standards under section 112 must be delegated under the provisions of section 112(1). Delegations of section 112 standards that occurred prior to the 1990 Amendments may remain in effect. Although the EPA could require rescission of these delegations under section 112(1)(1), which provides for review of enforcement delegations previously granted, it is permissible to conclude from that section's provisions and from the savings provisions in section 112(q) that delegations occuring prior to November 15, 1990 remain valid. Nevertheless, the EPA may choose to conduct a review of previously granted delegations under section 112(1)(1), and if the EPA finds, as a result of this review, that the basis for the Agency's determination under pre-1990 section 112(d) of adequacy regarding the State's program is no longer valid, the Agency may require the State to submit a request under section 112(1) to renew its delegation authority.

States may submit requests for section 112(1) programs that would provide for approval of existing standards without the need to repeat section 112(1)(5) notice and comment, as long as the State's law allows such delegation and there is a mechanism to assure that the State continues to meet the approval criteria of section 112(1). A State might be authorized under State law to accept delegation automatically, and as long as the State committed to an adequate funding mechanism, delegation of future standards would be approvable as long as any other section 112(1) requirements were met. If the State for some reason was unable to meet its commitment to provide adequate resources in the future, the auditing and withdrawal mechanism would allow the EPA to withdraw approval, thus providing protection against a State's failure to continue to meet the criteria.

Another procedural streamlining mechanism is the use of direct final rulemaking where appropriate for delegations where there has been no prospective approval like that discussed above. In the instances where the EPA did not expect any comment upon publication of a notice of approval, the notice could specify that the approval would become effective in 30 days unless comments were received. If comments were received,

then the EPA would have to renotice the approval and provide for a 30-day public comment period. The time and resource savings from this use of direct final rulemaking would thus depend on the correctness of the Agency's judgement regarding whether or not comments would be submitted.

For States seeking approval of programs under section 112(l) that will include requirements different from Federal requirements, additional information must be submitted. The requirements for these programs are specified under subpart E and in individual section 112 rules. In some cases, States will obtain approval of part 70 programs before they submit requests under part 63. When this is not the case, under certain circumstances, such as prior to approval of a State's part 70 program, or for a request for approval of standards or requirements for sources not subject to the requirements of part 70, the EPA will review State submissions under § 63.91, § 63.92 or § 63.93 according to the criteria in part 63 and will not require approval of the State's part 70 program as a precondition to approval under part 63. Nevertheless, § 63.94 continues to require part 70 program approval prior to section 112(l) approval (see § 63.94 comments). The EPA reserves the right to establish requirements for delegation under section 112(l) according to the criteria of section 112(1) and under other circumstances which may arise in the future. Because part 70 program approval may not necessarily precede approval under subpart E, the following changes have been made concerning the general criteria for approval:

Section 63.91(b)(5) is amended to state the plan should include 'at a minimum a complete description of the State's compliance tracking and enforcement program, including but not limited to inspection strategies."

Section 63.91(b)(3) is expanded to require the demonstration to include: (i) A description in narrative form of the scope, structure, coverage and processes of the State program; (ii) a description of the organization and structure of the agency or agencies that will have responsibility for administering the program; (iii) a description of the agency staff who will carry out the State program, including the number, occupation, and general duties of the employees. The State need not submit complete job descriptions for every employee carrying out the State program.

Section 63.91(b)(6) is revised to read: "A demonstration of adequate legal authority to assure compliance with the rule or program upon approval. At a minimum, the State must have the following legal authorities concerning enforcement: (i) The State shall have enforcement authorities that meet the requirements of § 70.11; (ii) If a State delegates authorities to a local agency, the State must retain enforcement authority

unless the local agency has authorities that meet the requirements of § 70.11.

The language of § 63.93(b)(4) "whenever they are a part of the rule for which the approved rule would substitute" is deleted.

Section 63.93(b)(4)(iv) is revised to read "The results of all required monitoring or testing must be reported at least every 6 months." Approval of a State part 70 program will substantially meet these requirements.

Objectives of the Act

Nearly every commenter discussed the application of the statutory provision under section 112(1)(5)(D), "not likely to satisfy, in whole or in part, the objectives of the Act." A wide diversity of viewpoints was expressed. Some commenters felt that this requirement gave the EPA the ability to disapprove State rules or programs that, although more stringent than the relevant Federal rule, ran counter to a policy direction the EPA has pursued. Other commenters pointed out that the EPA policies are not themselves objectives of the Act.

The EPA agrees that in application of this provision, the EPA policies do not necessarily represent the only possible way of meeting the objectives of the Act. The EPA policies generally represent the EPA decisions about the means it will use to achieve the Act's objectives. The 1990 Amendments support the adoption of alternative State and local standards that are at least as stringent as Federal standards and section 112(l) itself is structured to provide flexibility and to accommodate differing State and local approaches.

It would be counter to the goals of section 112(l) and the 1990 Amendments, therefore, for the EPA to disapprove a State or local rule or program simply because it perceives its policies to be different than those of the State standard. As previously explained, the EPA has deleted reference to "objectives of the Act" from § 63.91 because this provision is separate from the approval criteria under subpart E.

Opportunity for Public Comment and Review of Permit Modifications

One commenter expressed concern about public opportunity to review and comment on permits which must be updated as a result of an approval under this subpart.

The EPA agrees that public involvement in the review of such permits is appropriate and beneficial to help assure proper implementation. To clarify this position, § 63.91(a)(6) states that newly approved requirements be included in a permit via the process described under § 70.7(f) of this chapter.

Additional Language on Reopening of Permits

Another commenter pointed out that the requirement that language be inserted in each

permit describing permit reopening upon possible withdrawal of approval was unnecessary.

The EPA maintains the need to have a State reopen every permit per the process described in § 70.7(f) upon withdrawal. The EPA feels that such instances are cause for reopening because after withdrawal of the approved State standard, permits containing only the State standard no longer contain applicable requirements.

Compliance Uncertainty

Numerous commenters expressed concerns about the uncertainty sources face in the time period between a State submission for approval and the EPA's decision to approve or disapprove.

Until the EPA approves or disapproves a State submission, sources will remain uncertain about what standards will ultimately apply to them. Several factors may decrease this uncertainty. First, in many cases sources will already be in compliance or soon need to be in compliance with State requirements that are being submitted for approval. It will generally be far more beneficial to such sources to have approval granted, thereby obviating the need for such sources to take further action to comply with the otherwise applicable Federal requirements. Second, any State requirements submitted for approval will have undergone a public comment process at the State level. A third reason tempering concern can be added. While it is true that for any approval under this subpart, a source must always be in compliance with either the underlying Federal rule or the approved State rule or program requirements applicable to that source, this does not mean that sources need be immediately subject to a State rule or program upon approval. It is possible for States to grant additional time to sources to come into compliance with the approved State rule. In their submission to the EPA for approval, a State could set an absolute date for approval or establish a certain period to achieve compliance once a State rule or program is approved. If a State chooses to provide such flexibility, sources must be in compliance with the underlying Federal rule according to any specified compliance timeframes in the interim period.

C. Section 63.92—Approval of a State Rule That Adjusts a Section 112 Rule

This section describes the process and criteria for gaining approval under the first of three approval options. "Rule Adjustment" is the streamlined approval option based on a promulgated list of allowable adjustments to Federal rules that the EPA has determined to result in rules that are categorically no less stringent than the corresponding unchanged Federal rule.

Under each of the three approval options, the EPA will publish the approved rule or program in the **Federal Register** and incorporate the approved rule or program,

directly or by reference, under the appropriate subpart of part 63. Several commenters suggested that incorporation of the approved rule or program under the subpart containing the otherwise applicable Federal rule would simplify sources' understanding of applicable requirements. The EPA agrees and will incorporate rules upon approval into these subparts to the extent appropriate. As approvals under § 63.94 can cover numerous section 112 standards, approvals made under that section as well as approvals of ARP programs may be treated differently. For example, ARP programs may be incorporated under the part containing other accidental release regulations.

Stringency Comparison

A few commenters discussed whether comparisons of stringency are best made at the source level or the level of individual emission points.

Among the criteria for approval under this section, a State rule must be unequivocally no less stringent than the otherwise applicable Federal rule with respect to level of control for each source and emission point. The test for stringency of a State submission under the adjustment approval option is a check to assure that proposed State changes qualify as adjustments under § 63.92 and do not deviate from the Federally promulgated list of allowable adjustments in this section. Once that determination is made, no further judgment is necessary. Therefore, no additional Federal public notice and comment are necessary prior to approval of the adjusted State rule because this rulemaking to establish guidance for approval of State programs under section 112(1) constitutes adequate notice and opportunity for public comment for this approval option. For these reasons there can be no question about stringency of a rule submitted for approval under this section. Therefore, to the extent that there are specific Federal requirements for individual emission points, a State rule must match or exceed stringency at each regulated emission point. If a State seeks to submit a rule that creates opportunities for shifting emissions between emission points within the same source or some other type of averaging scheme, that rule must be submitted under § 63.93 in order that the EPA may evaluate the stringency of the State rule compared to the corresponding Federal rule in detail.

Public Notice and Comment

Several commenters felt that approvals granted under this section should be subject to additional opportunity for Federal public notice and comment. Another commenter stated that for approval under § 63.92, a public comment period at the Federal level is unnecessary if the State or local program is already subject to public participation requirements as stringent as the Federal process.

This section was specifically developed to provide sources and States an opportunity to use a streamlined approval process. The adjustment list was carefully chosen to include only adjustments that are unequivocally no less stringent and has been subject to public comment during this rulemaking under section 112(l). The Agency has considered the public comments regarding the stringency of the adjustments and generally disagrees with those commenters that thought the adjustments could under certain circumstances result in less stringent requirements. If however, a State request for approval under § 63.92 includes an adjustment that as applied in a particular circumstance would not be unequivocally no less stringent, the EPA will disapprove the State's request. The State would be free to resubmit its request under another approval option, such as § 63.93. Therefore, the EPA believes that additional notice and comment for requests under § 63.92 would be unnecessary and redundant and that this rulemaking constitutes adequate notice and opportunity for public comment. First of all, each State rule for which a state seeks approval must have undergone rulemaking at the State level that included public participation equivalent to that required at the Federal level. The regulated community and interested public would have had ample opportunity for comment at that time. Furthermore, this rulemaking for part 63 has provided additional opportunity for comment on the list of adjustments and its application to State rules. Prior to the 1990 Amendments, delegation to implement and enforce unchanged section 112 standards was granted under section 112(d), which did not require opportunity for public notice and comment if the State met "adequate procedures". Those "adequate procedures" specifically called for a demonstration of adequate legal authority, adequate resources, and expeditious compliance. Once a State successfully completed this demonstration, the authority to implement and enforce the unchanged rules was granted.

Under the 1990 Amendments, it is possible for a State to obtain delegation to implement and enforce State rules or programs that may differ from the Federal requirements by submitting them for approval under subpart E. The EPA believes that certain types of changes will clearly result in State rules that are unequivocally no less stringent than the relevant Federal rule. Only a limited set of changes, referred to under this section as adjustments, fall under this definition. These adjustments were included in the proposed subpart E and public comment was sought. A State rule that included only changes from the list of adjustments, therefore, can be approved without additional public notice and comment since the public has had the opportunity through the proposal of this rule to comment on each of the submitted changes. As

discussed above, by streamlining the procedures, the EPA has been able to reduce the normal period of the EPA approval time under this section from 180 days to 90 days and has made this change in § 63.92 in response to comments received. Finally, the EPA anticipates that approvals under § 63.92 will be numerous, routine, and noncontroversial. The Administrator may not approve, under § 63.92, any State program or rule in which "any one of the State adjustments to the Federal rule is in any way ambiguous with respect to the stringency of applicability, the stringency of the level of control, or the stringency of the compliance and enforcement measures for any affected source or emission point." Section 63.92(a)(2). If a State submittal under § 63.92 is in any way ambiguous, the EPA will disapprove the request. The State may then resubmit the request under § 63.93 and the EPA will seek public comment for no less than 30 days. Section 63.93(a). The EPA believes that the regulated community and the interested public will have had sufficient notice and opportunity to comment on the unambiguous, unequivocally no less stringent adjustments listed in § 63.92(b)(3) at the State level and through this rulemaking, and that additional safeguards are provided by the provision that requires a State to seek approval under § 63.93 for nonroutine changes to the Federal requirement.

"Any Other Adjustments"

Some commenters felt that the EPA should include among the list of adjustments one which read, "any other adjustments which are unequivocally no less stringent and which have been approved by the Administrator upon petition by the State."

The EPA believes that additions to the list of adjustments must afford an opportunity for Federal public notice and comment. This would generally amount to an amendment of this regulation and, therefore, such a category should not be included in the listed adjustments under § 63.92. Note, however, that the EPA may propose new adjustments specific to a particular section 112 rule at the time that the Federal section 112 rule is proposed. (See § 63.92(b)(3)(xiii).) The public will have opportunity to comment on such Federal rules when they are proposed.

Adjustment for Additional Pollutants

Several commenters felt that the EPA should include an adjustment allowing for the regulation of pollutants not among those listed under section 112(b).

The EPA has chosen not to include such an adjustment at this time. This situation may be difficult for the many States that regulate pollutants not on the section 112(b) list, as those States may not incorporate requirements that do not relate to Federally listed pollutants in the Federally enforceable section of the part 70 permit and therefore State-only requirements for additional pollutants would

need to be incorporated in a different section of the part 70 permit. See § 70.6(b)(2). Instead, the EPA encourages States with data that indicate a pollutant should be Federally regulated to submit a petition to the EPA to include such pollutants on the section 112(b) list

D. Section 63.93—Approval of State Authorities That Substitute for a Section 112 Rule

This is the second of three approval options under this subpart. Under this option States are given the widest possible range of flexibility in seeking approval of authorities that differ significantly from an otherwise applicable Federal rule. The EPA will make a detailed and thorough evaluation of the State's submittal to ensure that it meets the stringency and other requirements of this section.

Need for Emission Point Basis Rather Than "Affected Source"

One commenter felt that § 63.93 should be deleted from the rule, because it allows stringency to be compared at an "affected source" level rather than for each emission point. The EPA disagrees with the view that this option should be deleted. Under § 63.93, a detailed demonstration is required that will ensure that any approved State alternative will achieve an equal or greater reduction in emissions. This section is further reinforced by requirements to address in detail the effects of alternative enforcement and compliance methods. The final rule continues to address stringency for § 63.93 on a source" basis. The term "affected source" has, however, been deleted from § 63.90(a), because there does not appear to be a compelling need to define the term in both subpart E and also in subpart A "General Provisions" of this part. The final rule will, as a result, rely on the definition in subpart A once it is promulgated. The rule was proposed on August 11, 1993 (58 FR 42760).

For a given source category, the "affected source" definition will be specifically defined when the section 112 standard is promulgated for the category. In making stringency comparisons under § 63.93, the reviewer should therefore consult the appropriate subpart of part 63 for the "affected source" definition for the category in question.

Incorporation of Approved Rules

As discussed in the previous section of this preamble, several commenters suggested incorporation of an approved rule under the subpart containing the otherwise applicable Federal rule. The EPA agrees and will do so as appropriate.

Form of the Standard for Work Practice, Design, Operational or Equipment Standards

For approval under § 63.93, States are required to provide the EPA with a detailed demonstration showing that implementation

and enforcement of State authorities results in as great or greater emission reductions (or other appropriate measures in the case of section 112(r)) for each affected source as the implementation and enforcement of the otherwise applicable Federal rule. In contrast, under the approval option in § 63.94, no detailed demonstration is necessary but States are required to express permit terms and conditions that result from the approved State program requirement in the form of the Federal standard.

Numerous commenters expressed concern about inflexibility that the "form of the standard" requirement imposes. Several commenters had specifically suggested the addition of a provision allowing for a source specific detailed demonstration of stringency in instances where the "form of the standard" requirement severely limited needed flexibility. To address the concerns about this inflexibility, the EPA is broadening the type of State authority that can be approved under § 63.93 for certain types of standards and under certain conditions.

The EPA agrees that under certain conditions, requiring States to write permit conditions in the form of the Federal standard could be unnecessarily inflexible. Those conditions are when the following circumstances exist together: (1) The EPA writes a work practice, equipment, design or operational standard (in other words, when the EPA does not write a standard based on performance, such as control efficiency or an emission rate) and (2) such a Federal standard does not address a State work practice, equipment, design or operational standard as either meeting or failing to meet the Federal standard. These conditions will generally only arise when a State has decided to require control equipment or practices which have been developed since the EPA promulgated the relevant standard. In such cases the State standard may be at least as stringent as the previously promulgated Federal standard but can not be expressed in the form of the Federal standard. The flexibility otherwise provided in § 63.94 here would not allow the State standard to be the basis for the permit in these cases since the State standard would not be able to be expressed in the form of the Federal standard.

In this rule as proposed, States would have had the opportunity to solve this problem by codifying their standard in a source category-specific rule through a State rulemaking and then seeking approval of that rule under § 63.93. Since many States regulate hazardous air pollutants through authorities which do not include source-category-specific rules, pursuit of approval under these circumstances would require a significant employment of resources for largely administrative purposes so that the State standard could be expressed as a source-category specific State rule for which the State could seek approval.

To address this difficulty, the EPA has revised § 63.93 to allow for slightly broader

applicability of this section. Under the narrow circumstances discussed above, the EPA is not requiring the submission of a sourcecategory-specific rule for approval under this section. Instead the EPA is willing to review and to consider for approval a specific application of broader State authorities under certain conditions as explained here. As at proposal, the EPA will only grant approval under this section in lieu of a single specific section 112 rule that specifies work practice or similar requirements. Approval of a specific application of broad State authorities will only be given to a State with a program already approved under § 63.94 to regulate the applicable source category, only for a source category that is not federally regulated by a performance based standard and only where the Federal standard has not addressed the State's particular approach to controlling emissions. In this case, a State need not submit a rule specifically and exclusively addressing the requirements of the Federal rule. Instead a State may submit for approval legally enforceable broader authorities which allow it to regulate the source category in question, identifying the sources in that source category, and specifying proposed section 112 permit terms and conditions (such as the controls that are required by application of those authorities) and the authorities which will assure adequate compliance and enforcement according to the provisions of this subpart and part 70. If the EPA approves a State's request in such a case, only the specific application of the State authorities to a single source category as approved will be Federally enforceable terms and conditions. If a State later revises its authorities to require different controls or compliance and enforcement measures, those changed requirements will not be federally enforceable unless the State submits documentation or a request under § 63.91(c) regarding revisions of State authorities. The source will remain subject to the approved requirements incorporated in its part 70 permit according to the provisions of part 70 unless the EPA disapproves the revision or otherwise finds that the authorities are inadequate and initiates withdrawal proceedings. The EPA has included in § 63.93 the types of "authorities" needed for approval. Authorities submitted under that section must meet the criteria of section 112(1)(5)(A), that is the authorities must be adequate to assure compliance by all sources subject to the request for approval with each applicable Federal standard, regulation or requirement. A threshold requirement for approvability is that State authorities must be legally enforceable by the State under State law. Such legally enforceable authorities may be statutes, rules, regulations, or other instruments that impose legally enforceable requirements.

For example, a State might have a single regulation that assesses risk at facilities that

emit hazardous air pollutants and based on estimates of risk, requires specific emission rates or specific controls at particular facilities which might differ from facility to facility. The State's regulation might apply to a wide range of source categories in the State. If such a State initially received approval under § 63.94 for a State program that included a source category which was later regulated under a Federal equipment standard, approval under § 63.94 might provide very little flexibility to the State to require different types of equipment in lieu of equipment specified by the otherwise applicable Federal requirements. This might be especially problematic when a State sought to require innovative controls not evaluated at the time of Federal standard promulgation because they were not yet developed. If a State can demonstrate to EPA, via the process described in § 63.93 that such controls resulted in emission reductions for all sources in the source category as great or greater than the emissions reductions the Federal standard would achieve, the EPA is willing to consider a request to approve the State authorities requiring such controls as Federally enforceable in lieu of the otherwise applicable Federal standard. Resulting terms and conditions would be incorporated in a part 70 permit. This would be an approval of a specific application of broad State authorities under the narrow circumstances described.

Thus, as described above, provisions for approval under § 63.93 has been revised to allow a State to request approval of a limited application of its general air toxics regulatory authority as that authority applies to a single source category. To do so, the State would need to meet the requirements of this section which call for, among other things, a detailed analysis of emission reductions that would result from both the Federal and State scenarios. Only the terms and conditions to be incorporated in the source's part 70 permit, as approved under this subpart for the single application of authorities for the single source category for which the request was submitted, would be federally enforceable.

Stringency Criterion for Accidental Release Prevention Programs

In terms of the ARP program, one comment indicated that States should not be allowed to submit ARP programs under § 63.93 because approval is unnecessary. Rather, the Federal and current State programs could be easily meshed together and the most stringent requirements of each be included. The commenter also pointed out that the criteria for approval of equivalent State programs are primarily based on the ability of the State program to achieve equivalent or better emission reductions and that this criterion makes little sense in the context of accidental releases.

The Agency disagrees that States do not now, or will not in the future, need the flexibility of submitting ARP programs for approval which differ from the Federal requirements. However, EPA recognizes that the criteria in the proposed rulemaking may not have been sufficiently broad to include all the requirements under the section 112(r) program. Thus, the Agency has added additional approval criteria to § 63.93 which are specific to the ARP program.

E. Section 63.94—Approval of a State Program That Substitutes for Section 112 Emission Standards

This is the third of the three approval options. It allows for a one-time approval of a legally binding commitment adopted through under State law to adequately regulate sources subject to hazardous air pollutant section 112 emission standards as specified under § 63.94(b)(2). This section applies only to sources for which part 70 permits will be issued by the State and which are subject to section 112 requirements expressed as terms and conditions of the part 70 permit. Part 70 permit requirements must be written in the form of the Federal standard which would be otherwise applicable to the source. This section may be used to approve standards corresponding to Federal section 112 (d), (f), or (h) standards only and can not be used to approve infrastructure rules such as those developed under sections 112(g), 112(i)(5) or 112(r).

Infrastructure Rules

Some commenters felt that the EPA could approve under this section requests for approval of infrastructure rules. As mentioned above, stringency requirements for approval under this section require that permit terms and conditions resulting from approval be expressed in the form of the Federal standard. Infrastructure rules may include requirements that can not be simply compared for stringency through a test of emission rates or control efficiencies.

In fact, some infrastructure rules provide guidelines for case-by-case determinations on controls where no simple stringency comparison can be made but rather the determinations must meet criteria specified in the applicable section 112 provisions. For these reasons, State infrastructure rules can only be approved under subpart E when they either adjust the Federal rule per § 63.92 or the State submits a detailed demonstration of stringency according to the provisions of § 63.93.

Stringency Comparison

As in the previous section, commenters questioned whether the basis for stringency comparisons should be at the source or emission point level. Approvals under this section require States to write permits in the form of the Federal standard. If the Federal standard requires for example, specific controls or emission rates from specific emission points, that form must be maintained in permits resulting from an approval under

this section. If, on the other hand, a Federal standard has no requirements that apply at the emission point level, but instead stringency is measured at a more broadly defined source level, a State could choose to express terms and conditions at the source level also, as long as those terms and conditions were no less stringent than the Federal requirements and were expressed in the Federal form. This allows States to express terms and conditions with the same degree of flexibility that is allowed by the Federal standard.

Part 70—Approval Prior to Section 112(l) approval

As discussed earlier, commenters expressed preferences both for and against the requirement of part 70 program approval prior to a State receiving approval under this subpart. The primary Federal determination of stringency under this approval option occurs through EPA review of a part 70 permit with terms and conditions expressed in the form of the Federal standard. The EPA therefore finds that it is necessary for a State to be implementing an approved part 70 program before it would approve a request under this section for different but equally effective State programs. More specifically, approval under this section for this purpose would only apply to those sources for which the State is the part 70 permitting authority. This addresses any potential applicability issues that might arise from a partial or interim approval under part 70.

State Enforceability

Two commenters noted that States with existing hazardous air pollutant programs, which can include State statutes, regulations, or other requirements that limit the emissions of hazardous air pollutants from affected sources and that may be structurally dissimilar from section 112 regulations (e.g. risk-based standards) should be allowed the option to operate independently of and in addition to the Federal MACT standards and programs while at the same time maintaining Federal applicable requirements in part 70 operating permits. These State standards would be State enforceable only.

EPA agrees with this comment. Nothing in today's rule precludes a State from operating existing programs that may differ from federal section 112 emission standards and requirements as long as they are enforceable as State-only requirements. Such State-only requirements may be incorporated in a part 70 permit under § 70.6(b)(2). In addition, as discussed previously, section 116 of the Act preserves the right of States to adopt and enforce standards or limitations as long as they are no less stringent than Federal section 112 standards or limitations.

F. Section 63.95—Additional Approval Criteria for a State Rule That Adjusts or Substitutes for the Federal Accidental Release Prevention Program

Section 63.95 contains specific approval criteria for the approval of State programs which adjust or substitute for the Federal accidental release prevention program.

Section 112(r) Registration of Facilities Under Section 112(1)

One comment disputed whether the proposed rule properly addressed the provision of section 112(1)(2) which directed the EPA to draft guidance under this section which "provides for" the registration of facilities producing, processing, handling, or storing over a threshold quantity of a substance listed under section 112(r). The comment stated that § 63.95 lacked sufficient specificity to fulfill the guidance required by the statute. The commenter suggested specific components of a registration program, including standards for outreach, verification of coverage through database crosschecking, and the specific contents of a registration form. The commenter suggested that by using the phrase "provide for" in the requirement to promulgate guidance, Congress did not intend for the EPA to delay informing affected parties of the minimum requirements of an acceptable program. The EPA generally disagrees with the comment that the EPA has failed to provide sufficient guidance to States as required by section 112(1)(2). However the EPA notes that it has made some modifications to the provisions of §§ 63.93 and 63.95 to clarify the regulatory provisions. Furthermore, the EPA believes that the publication of the proposed rule implementing section 112(r)(7)(B) provides additional guidance to States concerning the specific minimum contents of registration. Together, these actions provide not only for the guidance required by statute but also provide much of the supplementary specifics suggested by the commenter.

Today's rule requires a State agency seeking delegation of the Accidental Release Prevention Program to include procedures for registering stationary sources covered by the section 112(r) rules in a manner consistent with the registration requirements under those rules. Section 63.95(b)(2). In addition, to make clear that a State must include a procedure for registration in its application for delegation of the Accidental Release Prevention Program, § 63.95 requires a State to demonstrate authority and resources to enforce all core release prevention requirements. Furthermore, the rule requires a State to describe its outreach program.

The Agency has modified § 63.95 to clarify that a State seeking delegation of the Accidental Release Prevention Program must identify the State entity with which a source must register. Identifying the entity receiving the registration is a necessary aspect of

describing the procedures by which a State would register subject sources. The Agency has determined that no specific standards for outreach or database crosschecking are appropriate because such activities are extremely State specific. Certain States may maintain computerized Emergency Planning and Community Right-to-know Act Tier 2 databases while others may be able to rely on Standard Industry Classification (SIC) codes, property tax filings, and other information for outreach purposes and for determining whether all covered sources are registered. However, a description of such outreach and oversight activities would be relevant in consideration of the adequacy of program resources.

The Agency set out proposed specific requirements for registration of risk management plans in the proposed rule implementing section 112(r)(7)(B) (proposed § 68.12). The proposed section 112(r)(7)(B)rule, if adopted, would require a stationary source that has over a threshold quantity of a substance listed pursuant to section 112(r)(3) to register with the Administrator within three years of the final rule's publication. The registration would contain identifying information about the source (name, street and mailing addresses, telephone number, contact persons, Dun and Bradstreet number, applicable SIC codes), data on listed substances present in abovethreshold quantities, and a certification by the owner or operator concerning the accuracy of the information submitted and the submission of risk management plans to appropriate local, State and Federal authorities. Such data would need to be updated when it is no longer accurate. The proposed section 112(r)(7)(B) rule does not propose to require additional information concerning the plant's safety programs and surrounding populations because such data would be difficult to standardize for data management purposes. Furthermore, such information does not need to be included in the registration for right-toknow purposes because such information already would be available to the public in the risk management plans filed locally, with the State, and with the Chemical Safety and Hazards Investigation Board. Comments concerning the contents of registration submittal should be directed to the rulemaking docket for the section 112(r)(7)(B) proposed rule.

The Agency believes that today's rule and the discussion herein fulfills its duty to promulgate guidance that provides for registration of facilities that have more than a threshold of a section 112(r) regulated substance. The rule promulgated today unambiguously requires a State seeking delegation of a section 112(r) program to have an element providing for facility registration, which is consistent with the 1989 Senate Environment and Public Works Committee report explaining the guidance requirement. See S. Rep. 1630 at 193. The

Agency interprets the requirement to provide guidance for registration of covered facilities to mean that the Agency must make clear that a State seeking a delegation of the section 112(r) program must have a registration element in its program.

The Agency does not believe that today's rule must detail the substantive data requirements for registration because such detail would be inconsistent with the structure of section 112(1) and section 112(r). Under section 112(r)(7)(B), all stationary sources that have over a threshold quantity of a substance regulated under section 112(r) must prepare a risk management plan. Sources that prepare risk management plans must register such plans with the Administrator. Section 112(r)(7)(B)(iii). The Agency interprets the registration of facilities mentioned in section 112(1)(2) to be the same registration as the registration of stationary sources required under section 112(r). Facilities described in section 112(1)(2) would not have a threshold quantity or more of a chemical unless the quantity would also trigger registration under section 112(r). To interpret section 112(l) to require a different registration than section 112(r) would require States opting to develop a delegated Accidental Release Prevention Program to run two redundant registration programs. Section 112(1) provides a means to delegate the section 112(r) registration requirement as part of a delegation of the Accidental Release Prevention Program. No commenter has suggested otherwise.

The statute provides the Agency with a different and more lengthy time frame to develop the section 112(r) registration requirements than it provides for the development of "guidance * * * useful to the States in developing programs for submittal." Section 112(l)(2). It would be a strained reading of the section 112(l) guidance requirement to say that the Agency must detail the specific registration requirements for State programs that elect to seek delegation of the Accidental Release Prevention Program prior to the Agency developing the actual registration requirement.

The Agency has provided useful guidance to the States concerning the registration requirement by promulgating this rule, discussing registration in today's preamble, and responding to this comment. The rule outlines the minimum content of a State delegation submittal and explicitly provides that such submittal must include a description of the State's registration process. Such description must include an identification of the State entity with which parties must register. Furthermore, as noted above, the Agency has discussed its most current view of the specific details of what registration will entail. Section 112(1) does not require the Agency to provide the specific elements of what information is necessary for facility registration. The 1990 Amendments leave the development of the specific elements of

registration to a future section 112(r)(7) rulemaking.

Section 112(r) Authorities

One commenter indicated that States should be required to obtain the authorities for the general duty and emergency order authority provisions found in section 112(r)(1) and (9) respectively, because State agencies will often receive citizen complaints about hazards and will have more of the expertise necessary to use such authorities properly than the EPA's regional personnel. The Agency believes, however, that States should be given the option to have authorities beyond the core elements necessary to administer the program. While the general duty and emergency order authority provisions could enhance the State program by providing them with additional compliance and enforcement tools, they are not essential elements which would be required to maintain a functioning ARP program at the State level.

Further, many States already have emergency order authorities under other environmental statutes and may not find section 112(r)(9) critical to the administration of their program. In terms of the general duty provisions, some States are prohibited from having general duty authorities.

Section 112(r) Enforcement Authorities

One commenter indicated that the EPA should specifically advise States that they must have the authority to impose the penalties required under the 1990 Amendments for violators of section 112(r). The proposed rulemaking contained specific language which indicates that the State submission would need to contain a demonstration of the State's authority to enforce all accidental release prevention requirements including a risk management plan auditing strategy that is consistent with the proposed section 112(r)(7) rule and this language has been retained in the final rule. Section 63.95(b)(3).

Interface Between Section 112(r) and Part 70

One commenter suggested that the Agency should require States to establish appropriate interagency agreements which would promote the exchange of information between the administering agency and the permitting agency if they are different. The Agency agrees that information flow is critical if the implementing State agency is not the permitting agency. This is particularly important since section 112(r) requires the development, submittal, and implementation of a risk management plan which must be addressed in a part 70 permit for subject sources. Consequently, the Agency has added language to § 63.95 which requires a description of any coordination mechanisms the implementing agency will use with the air permitting program, provided it is not the implementing agency.

G. Section 63.96—Review and Withdrawal of Approval

This section discusses terms for the EPA review of the implementation and enforcement of approved State rules and programs and describes the process and criteria for EPA withdrawal of a State approval.

Source Uncertainty About Withdrawals

Numerous commenters expressed concern over the uncertainty that sources might face when approval of a rule or program to which they are subject is withdrawn.

Generally, there are three reasons upon which the Administrator might base a withdrawal. The State might lack adequate authority or resources, the State might not be implementing or enforcing the rule or program adequately, or the rule or program might be found to be less stringent than the otherwise applicable Federal rule or program, perhaps, for example, as a result of EPA review sometime after approval.

If the Administrator withdraws a program for the third of these three reasons, sources' permits will need to be reopened according to the provisions of part 70 and the underlying Federal standard will become the applicable Federally enforceable requirement again on the date set forth by the Administrator in a compliance schedule published concurrently with the withdrawal.

This withdrawal and permit reopening would be due to the fact that the rule or program was found to be less stringent than the Federal standard and is, therefore, no longer appropriate as a substitute for the Federal standard in the part 70 permit. In this case, the permits of sources subject to the requirement will be reopened according to procedures specified in § 70.7(f) because the withdrawal amounts to a finding by the EPA that the permit no longer assures compliance with the applicable requirement consistent with § 70.7(f)(iv). The withdrawal also results in additional requirements becoming applicable to the source, which triggers a reopening under § 70.7(f)(1)(i). Sources would be required to come into compliance by the date specified in the Federal Register withdrawal notice regardless of whether or not the permit has been reopened. Since the Federal standard is considered a new requirement, the permit shield in § 70.6(f) would not apply. When only the first or second reasons are cited by the Administrator, and the stringency of the State standard is not in question, reopening of part 70 permits will not be required for sources affected by withdrawal of an approval under § 63.92 or § 63.94. Any source that is in compliance with permit conditions established under such approvals will also be in compliance with the underlying Federal standard upon withdrawal, because a source in compliance with a no less stringent State standard that is in the same form as the Federal standard is also in

compliance with the Federal standard. Approved State standards under § 63.92 and permit terms and conditions resulting from an approved State program under § 63.94 are necessarily in the same form as the otherwise applicable Federal standard.

The situation is different for approvals under § 63.93, in particular for work practice, design, operational or equipment standards because such approved State standards would commonly not be in the same form as the otherwise applicable Federal standard. To assure sources of greater certainty, EPA has revised § 63.96 to provide that permits need not be reopened if the Administrator finds at the time of withdrawal that the approved State rule is still demonstrated to be no less stringent than the otherwise applicable Federal standard. Section 63.96(b)(5). In such cases the Administrator will approve as equivalent according to the provisions of the appropriate subpart of part 63 the equipment, design, work practice or operational standard, emission limitation, or other requirement upon which the original approval was based. This is in accordance with the provisions of section 112(h)(3) for alternative standards. Such an equivalence determination was proposed in § 63.6(g) of subpart A of this part (58 FR 42760 August 11, 1993).

To further increase certainty for sources affected by a withdrawal, the EPA will publish an expeditious schedule for compliance by sources for both involuntary and voluntary withdrawals. Included in this schedule are interim emission limits, as appropriate, to limit emissions for the time period between withdrawal and the deadline for the source coming into compliance with the Federal standard. Sources must be operated in a manner consistent with good air pollution control practices for minimizing emissions at all times during this transition period. The schedule will be published in the Federal Register notice withdrawing the approval.

Audits

One commenter noted that the EPA should commit to audits, at least every 3 years, of programs which implement any averaging allowed in approved rules. Today's rule provides that the EPA "may at any time * * * review the adequacy of implementation and enforcement of an approved rule or program * * *". The EPA believes that today's rule provides the appropriate degree of flexibility in performing periodic reviews and allowing the EPA to determine on a case-by-case basis the frequency of those reviews.

One commenter asked the EPA to consider employing an auditing program instead of individual permit reviews. As allowed under § 63.96(a), the EPA intends to establish a program for a review of approved rules and programs and the audit of permits that result from such approvals. Such a program, however, can not replace the EPA's authority

to review and potentially veto any rule or program approved under subpart E if and when the EPA finds such review to be necessary. In addition, § 70.10 provides additional protection through Federal oversight of State part 70 programs.

H. Other Comments

Potential To Emit

The May 19, 1993 proposal, requested comment on the potential to emit definition and how it related to submittals under section 112(1). The potential to emit issue, including concerns raised by comments to the subpart E proposal on this issue will not be addressed here but rather will be addressed in a later rulemaking. The issue was also discussed in the proposal for the General Provisions under subpart A of this part (58 FR 42760, proposed August 11, 1993). Since the potential to emit issue is currently under discussion, the EPA is deferring discussion of that issue at this time in this preamble.

Alternative Equipment Under Section 112(h)(3)

One commenter noted that the form of the standard limitation on State authority imposed by § 63.94 diminishes the flexibility in encouraging alternative technologies. The commenter believes that this approach is inconsistent with the fundamental policy goals of the CAA, including the goal of pollution prevention. The commenter feels that State programs should be allowed by the proposed rule to approve alternative technologies, particularly for equipment standards consistent with section 112(h)(3) of the Act.

Section 112(h) of the Act allows the EPA to promulgate equipment standards, in cases where an emission limitation is not feasible. An example of this type of equipment standard is the standard recently promulgated for perchloroethylene dry cleaning facilities (58 FR 49354, September 22, 1993). Under section 112(h)(3) of the Act, sources may request permission from the EPA for use of an alternative means of control. Procedures for review of these requests by the EPA have been proposed in § 63.6(g) of subpart A of this part (58 FR 42760 August 11, 1993).

The EPA wishes to clarify in this rulemaking, the process for making these section 112(h)(3) equivalency determinations for a State that has an approved program under § 63.94, or for a State that believes a given technology would satisfy the requirements of § 63.93. In the final rule for perchloroethylene dry cleaners (58 FR 49354, 49371, September 22, 1993), the EPA indicated that "Section 112(l) of the Act would allow a State to request approval of a State's program that permits a source to seek permission to use an alternative means of emission limitation under section 112(h)(3), provided that the State demonstrated that its

program would be no less stringent and that certain conditions were met."

The EPA is here in this discussion further clarifying the procedures by which a State may seek and obtain approval of such a program under section 112(1).

It is unlikely that, for an equipment standard promulgated in accordance with section 112(h) of the Act, there would be an alternative means of control that could satisfy the § 63.94 requirement that the program express the limitation in the same "form as the Federal standard." Accordingly, § 63.94 approval by itself would likely not be sufficient to provide for section 112(h)(3) equivalency determinations. There are, however, two other avenues for an equivalency finding that are provided in today's rule.

First, as described above (preamble section IV), § 63.93 has been modified to provide a means for approval of work practice, equipment, or similar standards that do not require the State to submit a category-specific rule. In order to use this process, the State must have prior approval under § 63.94, and must identify in its submittal under § 63.93, (1) the specific work practice, design, equipment or operational standard that would replace the Federal requirement, (2) a specific description of the State authorities that would be exercised, and (3) proposed part 70 permit terms and conditions. Once approved, the State equipment standard would become the applicable requirement.

Second, the State may develop a State rule containing the alternative equipment limitation and provide a detailed demonstration in accordance with § 63.93 that the rule is no less stringent than the Federal standard. A source seeking permission to use an alternative means of emission limitation under section 112(h)(3) would thus first request permission from the State, demonstrating that the subject alternative means is no less stringent than the Federal requirement. The State would then seek approval from the EPA for that kind of equipment or alternative means of emission limitation.

Currently, the EPA does not delegate authority to determine equivalency of emission control technologies to the States. The February 1983 "Good Practices Manual for Delegation of NSPS and NESHAPS", reserved to the EPA the determination of equivalency for design, equipment, or work place standards that will achieve a reduction in emissions as allowed for in section 112(h)(3) of the Act to the Administrator because these determinations require notice and opportunity for comment and impact National consistency of standards. While the EPA continues to retain the authority for this decision process, the EPA is providing as much flexibility as possible to the State and sources to receive approval of an equivalent emission control technology under subparts A and E of part 63.

Guidance on Acceptable Controls

Commenters noted that to reduce the paperwork burden and enhance national consistency, the EPA should provide guidance to States regarding acceptable controls on a source category basis. The EPA will usually provide information regarding acceptable controls in MACT rules (reference control technologies). Therefore, the EPA encourages States to provide comments to EPA on alternative compliance during development of MACT standards so that alternatives will receive the fullest possible consideration.

Approvals Should Be for Entire Rules

Commenters expressed concern that approval under section 112(l) should be for entire rules and not for individual provisions within rules.

In accordance with section 112(l)(1), the EPA has allowed for approval of State provisions which meet the stringency requirements specified in this section 112(l) rule and thus as previously described to become federally enforceable. This subpart provides flexibility to the states in submitting programs for approval and also minimizes dual regulation by providing a mechanism to approve those portions of existing State rules which meet the approval criteria specified in the section 112(l) rule. Determination of stringency as required under section 112(1)(1) is predicated on a corresponding Federal standard, rule, or requirement. As the section 112 regulations are developed, elements relevant to stringency determinations will be included, and State rules submitted for approval may be compared with the corresponding Federal requirements.

Notice of Delegation

One commenter noted that the EPA must provide clear notice of rule applicability, effectiveness, and enforceability by providing a notice of delegation which clearly states which Federal or State rules remain in effect and whether they are enforceable by the EPA or the State.

The EPA agrees with this comment and the **Federal Register** notice of approval will specify this information. In addition, the EPA has added § 63.90(d) to make clear that approved rules and requirements are Federally enforceable.

Exemption of Electric Utility Steam Generating Units Exemption of Sources of Radionuclide Emissions

One commenter noted that electric utility steam generating units are exempt under section 112(n) which provides for such exemption unless and until the EPA finds regulation to be appropriate and necessary after considering the results of a study currently being performed. The commenter stated that therefore the EPA can not enforce State regulation of such units at this time. Additionally, a commenter noted that

radionuclide emissions from facilities licensed by the Nuclear Regulatory Commission (NRC) should be exempted per section 112(d)(9).

The EPA agrees with these exclusions from subpart E regulation at this time according to the provisions of section 112(n) and section 112(d)(9). Under section 112(n), no Federal standard will be promulgated until some future time, and therefore no stringency comparison can be made at this time for a State rule applicable to sources subject to section 112(n).

One commenter argued that the EPA should not enforce State regulations of radionuclide emissions from facilities licensed by the Nuclear Regulatory Commission ("NRC"), citing the intention of Congress in section 112(d)(9) of the 1990 Amendments to avoid duplicative regulation of NRC licensees and the pending EPA proposal to rescind 40 CFR part 61, subpart I, for nuclear power reactors. Under section 112(d)(9), the EPA may decline to regulate radionuclide emissions from a particular category of NRC licensees if it determines by rule that the NRC program for that category of licensees provides an ample margin of safety to protect human health. Thus, before the EPA may decline to regulate radionuclide emissions from a category of NRC licensees, it must examine the NRC regulatory program for that category and determine that it provides a level of protection equal to or greater than would be provided by implementation of the EPA standard. Although the EPA has not yet made the requisite finding for any category of NRC licensees, it has proposed to rescind 40 CFR part 61, subpart I, as applied to nuclear power reactors (56 FR 37196, August 5, 1991) and to NRC licensees other than nuclear power reactors (57 FR 56877, December 1, 1992), and to rescind 40 CFR part 61, subpart T (56 FR 67561), December 31, 1991), which governs inactive uranium mill tailings disposal sites.

The EPA agrees with the commenter that recognition and enforcement by the EPA of State regulations of radionuclide emissions from NRC licensees which differ from the EPA standards is not consistent with the Congressional objective to eliminate unnecessary duplicative regulation of NRC licensees. Acceptance of alternative State standards or programs for such emission could also complicate the rescission of any Federal standards for which the EPA ultimately makes the necessary finding concerning the NRC program. So long as the EPA standards governing radionuclide emissions for NRC licensees are in effect, States may request that the EPA delegate enforcement of such standards pursuant to § 63.91, but the EPA will not grant requests to adjust such standards or to substitute State authorities or programs for such standards pursuant to § 63.92, § 63.93, or § 63.94. The

EPA has added a provision to § 63.90 clarifying this issue.

Of course, in those instances where a State may lawfully adopt differing or more stringent standards regulating radionuclide emissions from NRC licensees under its own authorities, the State may then include terms and conditions implementing such State standards in the State-enforced section of the permit for each facility. Moreover, the decision by the EPA not to adopt or enforce State standards governing radionuclide emissions from NRC licensees does not affect the ability of the States to seek EPA recognition and enforcement of other State standards or programs which may apply to NRC licensees.

High Risk Point Source Program

One commenter pointed out that section 112(l)(2) requires the EPA to publish guidance that includes as an element "an optional program begun in 1986 for the review of high-risk point sources of air pollutants including, but not limited to, hazardous air pollutants listed pursuant to (section 112(b))."

This final rule, along with a guidance document addressing the high-risk point source program described in section 112(1)(2), will fulfill the requirement to "publish guidance useful to the States in developing programs for submittal under (section 112(1))." A summary of the high-risk point source program guidance document, which will be available concurrently with the final rule is discussed below in section V of this preamble.

V. Additional Guidance

As stated previously in this preamble, additional guidance is concurrently published with this final rule. Specifically: (1) Guidance to review high-risk point sources; (2) information about establishing and maintaining various technical assistance activities, including an air toxics clearinghouse; and (3) enabling guidance outlining procedures, roles and responsibilities for section 112(1) approvals. These guidance documents are separate documents which are available with the promulgation of today's rule and may be revised and updated from time to time as appropriate. Each of these documents is discussed here.

A. High-Risk Point Source Guidance

Purpose of High Risk Point Source Guidance

The purpose of the High Risk Point Source (HRPS) guidance is to outline a methodology that State agencies may wish to employ in order to assess the risks from potentially high-risk point sources. The EPA envisions several uses of this program, particularly for those agencies that do not already have comprehensive air toxics programs. First, the guidance can help agencies evaluate and

regulate sources which will not be regulated under the Federal program. As an example, a listed source category may consist of major sources (those that emit greater than 10 tons per year of one hazardous air pollutant (HAP), or 25 tons per year of a combination of HAP's), and area sources (sources of a HAP that are not major). The major sources in the category will be covered by a section 112(d) standard, but the area sources may not be regulated unless the EPA finds that such sources present a threat of adverse effects to human health or the environment warranting Federal regulation under section 112 (section 112(c)(3)) (see source category list (57 FR 31576, July 16, 1992). In this case, the State agency may choose to assess a source to determine whether the State may wish to pursue state mandated controls. States may also undertake such analyses to examine residual risk after installation of Federal controls or risk from pollutants not on the section 112(b) list.

Second, an agency may wish to regulate sources under a faster timetable than the Federal program. Section 112(e) of the 1990 Amendments requires the EPA to regulate source categories on a specific schedule, either within 2, 4, 7, or 10 years after the date of enactment (November 15, 1990). A State may wish to apply the methodology offered in the HRPS guidance to evaluate a source or sources to determine whether early controls required at the State level are warranted. Similarly, States may wish to evaluate sources in order to set residual risk standards sooner than the Federal program. Section 112(f) requires the EPA to address the issue of residual risk eight years after the promulgation of a MACT standard. A State Agency may wish to examine the need for a residual risk examination before the eight year Federal analysis would be conducted.

Third, in response to public concern, agencies may wish to determine the risks associated with sources of air toxics, for a number of reasons, including questions from the public raised during part 70 permit hearings, or in response to public inquiries as to the safety of ambient air. A HRPS program can also increase environmental equity in that it helps an agency address, for example, a single source that may otherwise be missed because it was not in a source category to be regulated under the Federal regulatory program. See 136 Cong. Rec. S16978 (daily ed. Oct. 27, 1990) (Clean Air Conference Report, Air Toxics).

Finally, the methodology and resources presented in the HRPS guidance can add to the available tools States can use to evaluate the potential for adverse health impacts and protect the public health from local sources of HAPs. Information collected from the HRPS evaluations will be useful to the public, the State agencies themselves and to the Federal program.

The use of this guidance does not mandate regulation. It is designed to provide ideas for

developing or expanding upon State high-risk point source programs in keeping with the provisions of section 112(l)(2) that specify the optional nature of the high-risk point source program. Publication of this document fulfills the requirement of section 112(l)(2) to publish guidance that includes as an element the optional high-risk point source program begun in 1986.

Organization of this Document

The document that the EPA developed to fulfill Congress's directive concerning the HRPS program under section 112(1)(2) is based in large part on information and documentation that the EPA has developed from its experience with the program since the program's inception in 1986. See S. Rep. at 193-94 which describes the agency's efforts and support for the HRPS program. The document begins with a discussion of policy issues: how to determine what chemicals to assess, how to choose sources to assess, and how to communicate program objectives and risks to health. The document then outlines a tiered methodology agencies may choose to follow to determine whether the risk from a particular source (or sources) is significant, and worthy of regulation. The reader is then directed to appropriate EPA documents and services to assist in evaluating health effects from High Risk Point Sources.

B. Technical Assistance Activities

Several sources of technical assistance are provided by the EPA to State and local agencies. Each is briefly described.

Air Risk Information Support Center (Air RISC)

Developed to assist State and local air pollution agencies and EPA Regional offices on technical matters pertaining to health, exposure, and risk assessments for toxic air pollutants, the primary goal of Air RISC is to serve as a focal point for obtaining information and, where needed, provide assistance in the review and interpretation of that information.

Emission Measurement Technical Information Center (EMTIC)

Created for the purpose of promoting consistent and accurate emissions test method application in the development and enforcement of national, State, and local emission prevention and control programs, the EMTIC is an information exchange network that communicates the EPA emissions measurement technology to the emissions measurement community. The EMTIC provides information in the form of publications, videos, workshops, computer information databases, and support projects.

National Air Toxics Information Clearinghouse (NATICH)

This clearinghouse is intended to facilitate information exchange among State and local

agencies, and between the EPA and State and local agencies, and to minimize duplication of effort. The Clearinghouse consists of a computerized data base which contains information on potentially toxic air pollutants, hard copy reports of information from the data base, special reports, and a bi-monthly newsletter.

Clearinghouse for Inventories/Emission Factors (CHIEF)

This clearinghouse contains the latest information on air emission inventories and emission factors. It provides access to tools for estimating emission of air pollutants and performing air emission inventories for both criteria and toxic pollutants.

C. Enabling Guidance

The enabling guidance was developed as a further tool to assist State and local agencies interested in receiving approval of State programs under the process described in subpart E of this part. Also included is detailed information for procedures for receiving delegation for unchanged Federal section 112 rules. Included in this document is information on the following: (1) Specific roles and responsibilities of State and the EPA offices; (2) specifics regarding "detailed demonstrations" under § 63.93 submittals and "form of the standard" under § 63.94 submittals; (3) Forms used in submittals; and (4) commonly asked questions regarding section 112(1) submittals.

D. Accessing Additional Guidance

Technology Transfer Network (TTN) Bulletin Board System

This network provides information and technology exchange in different areas or air pollution control, ranging from emission test methods to regulatory air pollution control models. The individual bulletin boards offered with respect to air toxics are: (1) **Emission Measurement Technical** Information Center (EMTIC); (2) National Air Toxics Information Clearinghouse (NATICH); (3) Clean Air Act Amendments (1990 Amendments) and (4) Clearinghouse for Inventories/Emission Factors (CHIEF). The access number to the bulletin board system is 919-541-5742. If problems are encountered accessing the bulletin board, call 919-541-5384.

High Risk Point Source Guidance

The guidance document for the review of high risk point sources is available in Air Docket A–92–46. It can also be found on the EPA Technology Transfer Network (TTN) bulletin board system in the 1990 Amendments section under "Title III Policy and Guidance". The EPA Publication number for this document is EPA–453/R–93–039. To obtain copies, contact the EPA Library in Research Triangle Park, North Carolina at 919–541–2777 or the National Technical Information Service (NTIS) at 800–553–

6847. For other questions regarding this document, contact Kelly Rimer at 919–541–2962.

Air RISC

Air RISC services include a hotline, detailed technical assistance, and general technical assistance. Contact the hotline at 919–541–0888 and requesting the specific type of assistance needed.

Enabling Guidance

The enabling guidance document for Approval of State Programs and Delegation of Federal Authorities is available in Air Docket A–92–46. It can also be found on the TTN Bulletin Board system noted in the above paragraph. The EPA Publication number for this document is EPA–453/R–93–040 and can be obtained from the EPA Library or NTIS. For questions regarding this document, contact Sheila Milliken at 919–541–2625.

E. Grants

Section 112(1)(4) gives the Administrator the discretion to award grants to States to support the development of air toxics programs, including high-risk point source programs and the development and implementation of areawide area source programs pursuant to subsection 112(k).

The EPA has, for a number of years, supported air toxics program activities under State and local assistance grants issued pursuant to section 105 of the Act. The EPA will continue to evaluate, in close cooperation with the States, the types of activities that can and should be supported in this manner. The EPA is currently reviewing the exact mechanisms to be used for this purpose, including any administrative changes that may be required to track the grants pursuant to section 112(1)(4) authority instead of under section 105.

VI. Administrative Requirements

A. Docket

The docket for this regulatory action is A–92–46. The docket is an organized and complete file of all the information submitted to, or otherwise considered by, EPA in the development of this proposed rulemaking. The principal purposes of the docket are:

- (1) To allow interested parties a means to identify and locate documents so that they can effectively participate in the rulemaking process, and
- (2) To serve as the record in case of judicial review. The docket is available for public inspection at the EPA's Air Docket, which is listed under the ADDRESSES section of this document.

B. Executive Order 12866

Under Executive Order 12866 (58 FR 51735, 10/04/94), the Agency must determine whether the regulatory action is "significant" and therefore subject to OMB review and the

- requirements of the Executive Order. The Order defines "significant" regulatory action as one that is likely to lead to a rule that may:
- (1) Have an annual effect on the economy of \$100 million or more, or adversely and materially affecting a sector of the economy, productivity, competition, jobs, the environment, public health or safety, or State, local or tribal governments or communities;
- (2) Create a serious inconsistency or otherwise interfere with an action taken or planned by another agency;
- (3) Materially alter the budgetary impact of entitlements, grants, user fees, or loan programs or the rights and obligation of recipients thereof;
- (4) Raise novel legal or policy issues arising out of legal mandates, the President's priorities, or the principles set forth in the Executive Order.''

Pursuant to the terms of Executive Order 12866, the OMB has notified the EPA that this action is a "significant regulatory action" within the meaning of the Executive Order. For this reason, this action was submitted to the OMB for review. Changes made in response to the OMB suggestions or recommendations will be documented in the public record.

Any written comments from OMB to the EPA and any written EPA response to any of those comments will be included in the docket listed at the beginning of today's notice under **ADDRESSES**. The docket is available for public inspection at the EPA's Air Docket Section, (LE–131), ATTN: Docket No. A–92–46, U.S. Environmental Protection Agency, 401 M Street SW, Washington, DC 20460.

C. Paperwork Reduction Act

Under the Paperwork Reduction Act (44 U.S.C. 3501 et seq.), Federal agencies must obtain the OMB clearance for collection of information from ten (10) or more non-Federal respondents. Under this final rule, each State or other air pollution control agency which elects to develop a section 112(l) program, or to take any other approved actions under section 112(l), shall be required to submit to the Administrator a program, written findings, schedules, plans, statements, and/or other documentation required for approval of the submitted program or action. The effect of this rule is to subject those States and other air pollution control agencies utilizing section 112(l) to the informational requirements of this rule in order to assure that the requirements of a section 112(1) program or approved action have been met under section 112(1)(5) of the Act. These statutory requirements for approval give rise to the informational requirements of this rule.

The Office of Management and Budget (OMB) has approved the information collection requirements contained in this rule under the provisions of the Paperwork Reduction Act, 44 U.S.C. 3501 *et seq.* and

has assigned the OMB control number 2060–0264.

The burden to States and other air pollution control agencies for the collection of information under this rule for the first year is estimated to be a maximum of 1901 hours per State or agency. This estimate includes time for rule interpretation, analysis and/or revision of state or local legislative authority, development of a program and schedule of implementation, as well as demonstrations of adequate resources, compliance and enforcement. Since most of these requirements are not recurring, the burden will decrease significantly in subsequent years.

Send comments regarding the burden estimate or any other aspect of this collection of information, including suggestions for reducing this burden to: Chief, Information Policy Branch (2136), U.S. Environmental Protection Agency, 401 M Street, SW., Washington, DC 20460; and to the Office of Information and Regulatory Affairs, Office of Management and Budget, Washington, DC 20503, marked "Attention: Desk Officer for EPA."

D. Regulatory Flexibility Act

Under the Regulatory Flexibility Act, whenever an Agency publishes any proposed or final rule in the **Federal Register**, it must, except under certain circumstances, prepare a Regulatory Flexibility Analysis (RFA) that describes the impact of the rule on small entities (i.e., small businesses, organizations, and governmental jurisdictions). That analysis is not necessary, however, if an Agency's Administrator certifies that the rule will not have a significant economic impact on a substantial number of small entities.

The EPA believes that there will be no impact on any small entities as a result of the promulgation of this rule since all the entities which would have the authority to accept partial or complete delegation of the Administrator under section 112(1) of the Act are States and other governmental jurisdictions whose populations exceed 50,000 persons. With no impacts expected on entities whose populations are less than 50,000, a RFA is not required by law. What follows is the certification of the Administrator that an RFA is not required with the promulgation of this rule. Pursuant to section 605(b) of the Regulatory Flexibility Act, 5 U.S.C. 605(b), the Administrator certifies that this rule will not have a significant economic impact on a substantial number of small entities.

E. Review

This regulation will be reviewed 9 years from the date of promulgation. This review will include an assessment of such factors including overlap with other programs, the existence of alternative methods, enforceability, and result of section 112 standards review.

List of Subjects

40 CFR Part 9

Reporting and recordkeeping requirements.

40 CFR Part 63

Environmental protection, Administrative practices and procedures, Air pollution control, Hazardous substances, Intergovernmental relations, Reporting and recordkeeping requirements.

Dated: November 15, 1993.

Carol M. Browner,

Administrator.

*

For the reasons set out in the preamble, title 40, chapter I, of the Code of Federal Regulations is amended as follows:

PART 9—[AMENDED]

1. The authority citation for part 9 continues to read as follows:

Authority: 7 U.S.C. 135 et seq., 136–136y; 15 U.S.C. 2001, 2003, 2005, 2006, 2601–2671; 21 U.S.C. 331j, 346a, 348; 31 U.S.C. 9701; 33 U.S.C. 1251 et seq., 1311, 1313d, 1314, 1321, 1326, 1330, 1344, 1345 (d) and (e), 1361; E.O. 11735, 38 FR 21243, 3 CFR, 1971–1975 Comp. p. 973; 42 U.S.C. 241, 242b, 243, 246, 300f, 300g, 300g-1, 300g-2, 300g-3, 300g-4, 300g-5, 300g-6, 300j-1, 300j-2, 300j-3, 300j-4, 300j-9, 1857 et seq., 6901–6992k, 7401–7671q, 7542, 9601–9657, 11023, 11048.

2. Section 9.1 is amended by adding a new entry to the table under the indicated heading to read as follows:

§ 9.1 OMB approvals under the Paperwork Reduction Act.

PART 63—NATIONAL EMISSION STANDARDS FOR HAZARDOUS AIR POLLUTANTS FOR SOURCE CATEGORIES

1. The authority citation for part 63 continues to read as follows:

Authority: 42 U.S.C. 7401, et seq.

2. Part 63 is amended by adding subpart E to read as follows:

Subpart E—Approval of State Programs and Delegation of Federal Authorities

Sec.

63.90 Program overview.

63.91 Criteria common to all approval options.

63.92 Approval of a State rule that adjusts a section 112 rule.

63.93 Approval of State authorities that substitute for a section 112 rule.

Sec.

63.94 Approval of a State program that substitutes for section 112 emission standards.
63.95 Additional approval criteria for accidental release prevention programs.
63.96 Review and withdrawal of approval.

§63.90 Program overview.

The regulations in this subpart establish procedures consistent with section 112(1) of the Clean Air Act (Act) (42 U.S.C. 7401-7671q). This subpart establishes procedures for the approval of State rules or programs to be implemented and enforced in place of certain otherwise applicable section 112 Federal rules, emission standards or requirements (including section 112 rules promulgated under the authority of the Act prior to the 1990 Amendments to the Act). Authority to implement and enforce section 112 Federal rules as promulgated without changes may be delegated under procedures established in this subpart. This subpart also establishes procedures for the review and withdrawal of section 112 implementation and enforcement authorities delegated through this subpart.

(a) *Definitions*. The following definitions apply to this subpart.

Applicability criteria means the regulatory criteria used to define all emission points within all affected sources subject to a specific section 112 rule.

Approval means a determination by the Administrator that a State rule or program meets the criteria of § 63.91 and the additional criteria of either § 63.92, § 63.93 or § 63.94, where appropriate. For accidental release prevention programs, the criteria of § 63.95 must also be met.

Compliance and enforcement measures means requirements within a rule or program relating to compliance and enforcement, including but not necessarily limited to monitoring, test methods and procedures, recordkeeping, reporting, compliance certification, inspection, entry, sampling or accidental release prevention oversight.

Level of control means the degree to which a rule or program requires a source to limit emissions or to employ design, equipment, work practice, operational, accident prevention or other requirements or techniques (including a prohibition of emissions) for:

- (1)(i) Each hazardous air pollutant, if individual pollutants are subject to emission limitations, and
- (ii) The aggregate total of hazardous air pollutants, if the aggregate grouping is subject to emission limitations, provided that the rule or program would not lead to an increase in risk to human health or the environment; and
- (2) each substance regulated under section 112(r).

Local agency means a local air pollution control agency or, for the purposes of § 63.95, any local agency or entity having responsibility for preventing accidental

releases which may occur at a source regulated under section 112(r).

Program means, for the purposes of an approval under § 63.94, a collection of State statutes, rules or other requirements which limits or will limit the emissions of hazardous air pollutants from affected sources.

Stringent or stringency means the degree of rigor, strictness or severity a statute, rule, emission standard or requirement imposes on an affected source as measured by the quantity of emissions, or as measured by parameters relating to rule applicability and level of control, or as otherwise determined by the Administrator.

- (b) Local agency coordination with state and territorial agencies. Local agencies submitting a rule or program for approval under this subpart shall consult with the relevant State or Territorial agency prior to making a request for approval to the Administrator. A State or Territorial agency may submit requests for approval on behalf of a local agency after consulting with that local agency.
- (c) Authorities retained by the Administrator. (1) The following authorities will be retained by the Administrator and will not be delegated:
- (i) The authority to add or delete pollutants from the list of hazardous air pollutants established under section 112(b);
- (ii) The authority to add or delete substances from the list of substances established under section 112(r);
- (iii) The authority to delete source categories from the Federal source category list established under section 112(c)(1) or to subcategorize categories on the Federal source category list after proposal of a relevant emission standard;
- (iv) The authority to revise the source category schedule established under section 112(e) by moving a source category to a later date for promulgation; and
- (v) Any other authorities determined to be nondelegable by the Administrator.
- (2) Nothing in this subpart shall prohibit the Administrator from enforcing any

- applicable rule, emission standard or requirement established under section 112.
- (3) Nothing in this subpart shall affect the authorities and obligations of the Administrator or the State under title V of the Act or under regulations promulgated pursuant to that title.
- (d) Federally-enforceable requirements. All rules and requirements approved under this subpart and all resulting part 70 operating permit conditions are enforceable by the Administrator and citizens under the Act.
- (e) Standards not subject to modification or substitution. With respect to radionuclide emissions from licensees of the Nuclear Regulatory Commission or licensees of Nuclear Regulatory Commission Agreement States which are subject to 40 CFR part 61, subpart I, T, or W, a State may request that the EPA approve delegation of implementation and enforcement of the Federal standard pursuant to § 63.91, but no changes or modifications in the form or content of the standard will be approved pursuant to § 63.92, § 63.93, or § 63.94.

$\S\,63.91$ $\,$ Criteria common to all approval options.

(a) Approval process. To obtain approval under this subpart of a rule or program that is different from the Federal rule, the criteria of this section and the criteria of either § 63.92, § 63.93 or § 63.94 must be met. For approval of State programs to implement and enforce Federal section 112 rules as promulgated without changes (except for accidental release programs), only the criteria of this section must be met. For approval of State rules or programs to implement and enforce the Federal accidental release prevention program as promulgated without changes, the requirements of this section and § 63.95 must be met. In the case of accidental release prevention programs which differ from the Federal accidental release prevention program, the requirements of this section, § 63.95, and either § 63.92 or § 63.93 must be met. For a State's initial request for approval,

- and except as otherwise specified under § 63.92, § 63.93, or § 63.94 for a State's subsequent requests for approval, the approval process will be the following.
- (1) Upon receipt of a request for approval, the EPA will review the request for approval and notify the State within 30 days of receipt whether the request for approval is complete according to the criteria in this subpart. If a request for approval is found to be incomplete, the Administrator will so notify the State and will specify the deficient elements of the State's request.
- (2) Within 45 days after receipt of a complete request for approval, the Administrator will seek public comment for a minimum of 30 days on the State request for approval. The Administrator will require that comments be submitted concurrently to the State.
- (3) If, after review of public comments and any State responses to comments submitted to the Administrator within 30 days of the close of the public comment period, the Administrator finds that the criteria of this section are met, the State rule or program will be approved by the Administrator under this section, published in the **Federal Register**, and incorporated directly or by reference, in the appropriate subpart of part 63. Authorities approved under § 63.95 will be incorporated pursuant to requirements under section 112(r).
- (4) Within 180 days of receiving a complete request for approval, the Administrator will either approve or disapprove the State rule or program.
- (5) If the Administrator finds that; any of the criteria of this section are not met, or any of the criteria of either § 63.92, § 63.93 or § 63.94 under which the request for approval was made are not met, the Administrator will disapprove the State rule or program. If a State rule or program is disapproved, the Administrator will notify the State of any revisions or additions necessary to obtain approval. Any resubmittal by a State of a request for approval will be

considered a new request under this subpart.

- (6) If the Administrator finds that; all of the criteria of this section are met; and all of the criteria of either § 63.92, § 63.93 or § 63.94 are met, the Administrator will approve the State rule or program and thereby delegate authority to implement and enforce the approved rule or program in lieu of the otherwise applicable Federal rules, emission standards or requirements. The approved State rule or program shall be Federally enforceable from the date of publication of approval. When a State rule or program is approved by the Administrator under this subpart, applicable part 70 permits shall be revised according to the provisions of § 70.7(f) of this chapter. Operating permit conditions resulting from any otherwise applicable Federal section 112 rules, emission standards or requirements will not be expressed in the State's part 70 permits or otherwise implemented or enforced by the State or by the EPA unless and until authority to enforce the approved State rule or program is withdrawn from the State under § 63.96. In the event approval is withdrawn under § 63.96, all otherwise applicable Federal rules and requirements shall be enforceable in accordance with the compliance schedule established in the withdrawal notice and relevant part 70 permits shall be revised according to the provisions of § 70.7(f) of this chapter.
- (b) Criteria for approval. Any request for approval under this subpart shall meet all section 112(l) approval criteria specified by the otherwise applicable Federal rule, emission standard or requirements and all of the approval criteria of this section. The State shall provide the Administrator with the following.
- (1) A written finding by the State Attorney General (or for a local agency, the General Counsel with full authority to represent the local agency) that the State has the necessary legal authority to implement and to enforce the State rule or program upon approval and to assure compliance by all sources within the State with each applicable section 112 rule, emission standard or requirement. At a minimum, the State must have the following legal authorities concerning enforcement:
- (i) The State shall have enforcement authorities that meet the requirements of § 70.11 of this chapter.
- (ii) The State shall have authority to request information from regulated sources regarding their compliance status.
- (iii) The State shall have authority to inspect sources and any records required to determine a source's compliance status.
- (iv) If a State delegates authorities to a local agency, the State must retain enforcement authority unless the local agency has authorities that meet the requirements of § 70.11 of this chapter.
- (2) A copy of State statutes, regulations and other requirements that contain the appropriate provisions granting authority to

- implement and enforce the State rule or program upon approval.
- (3) A demonstration that the State has adequate resources to implement and enforce all aspects of the rule or program upon approval, which includes:
- (i) A description in narrative form of the scope, structure, coverage and processes of the State program;
- (ii) A description of the organization and structure of the agency or agencies that will have responsibility for administering the program; and
- (iii) A description of the agency staff who will carry out the State program, including the number, occupation, and general duties of the employees.
- (4) A schedule demonstrating expeditious State implementation of the rule or program upon approval.
- (5) Å plan that assures expeditious compliance by all sources subject to the rule or program upon approval. The plan should include at a minimum a complete description of the State's compliance tracking and enforcement program, including but not limited to inspection strategies.
- (6) A demonstration of adequate legal authority to assure compliance with the rule or program upon approval. At a minimum, the State must have the following legal authorities concerning enforcement:
- (i) The State shall have enforcement authorities that meet the requirements of § 70.11 of this chapter.
- (ii) If a State delegates authorities to a local agency, the State must retain enforcement authority unless the local agency has authorities that meet the requirements of § 70.11 of this chapter.
- (c) Revisions. Within 90 days of any State amendment, repeal or revision of any State rule, program, or other authorities supporting an approval under this subpart, a State must provide the Administrator with a copy of the revised authorities and meet the requirements of either paragraph (c) (1) or (2) of this section.
- (1) (i) The State shall provide the Administrator with a written finding by the State Attorney General (or for a local agency, the General Counsel with full authority to represent the local agency) that the State's revised legal authorities are adequate to continue to implement and to enforce all previously approved State rules and the approved State program (as applicable) and adequate to continue to assure compliance by all sources within the State with approved rules, the approved program (as applicable) and each applicable section 112 rule, emission standard or requirement.
- (ii) If the Administrator determines that the written finding is not adequate, the State shall request approval of the revised rule or program according to the provisions of paragraph (c)(2) of this section.
- (2) The State shall request approval under this subpart of a revised rule or program.

- (i) If the Administrator approves the revised rule or program, the revised rule or program will replace a rule or program previously approved.
- (ii) If the Administrator disapproves the revised rule or program, the Administrator will initiate procedures under § 63.96 to withdraw approval of any previously approved rule or program that may be affected by the revised authorities.
- (iii) Until such time as the Administrator approves or withdraws approval of a revised rule or program, the previously approved rule or program remains Federally enforceable.

§ 63.92 Approval of a State rule that adjusts a section 112 rule.

Under this section a State may seek approval of a State rule with specific adjustments to a Federal section 112 rule.

- (a) Approval process. (1) If the Administrator finds that the criteria of this section and the criteria of § 63.91 are met, the State rule will be approved by the Administrator, published in the **Federal Register** and incorporated, directly or by reference, in the appropriate subpart of this part 63, without additional notice and opportunity for comment. Rules approved under § 63.95 will be incorporated pursuant to requirements under section 112(r).
- (2) If the Administrator finds that any one of the State adjustments to the Federal rule is in any way ambiguous with respect to the stringency of applicability, the stringency of the level of control, or the stringency of the compliance and enforcement measures for any affected source or emission point, the Administrator will disapprove the State rule.
- (3) Within 90 days of receiving a complete request for approval under this section, the Administrator will either approve or disapprove the State rule.
- (b) *Criteria for approval*. Any request for approval under this section shall meet all of the criteria of this section and § 63.91 before approval. The State shall provide the Administrator with:
- (1) A demonstration that the public within the State has had adequate notice and opportunity to submit written comment on the State rule; and
- (2) A demonstration that each State adjustment to the Federal rule individually results in requirements that:
- (i) Are unequivocally no less stringent than the otherwise applicable Federal rule with respect to applicability;
- (ii) Are unequivocally no less stringent than the otherwise applicable Federal rule with respect to level of control for each affected source and emission point;
- (iii) Are unequivocally no less stringent than the otherwise applicable Federal rule with respect to compliance and enforcement measures for each affected source and emission point; and
- (iv) Assure compliance by every affected source no later than would be required by the otherwise applicable Federal rule.

- (3) State adjustments to Federal section 112 rules which may be part of an approved rule under this section are:
- (i) Lowering a required emission rate or de minimis level;
- (ii) Adding a design, work practice, operational standard, emission rate or other such requirement;
- (iii) Increasing a required control efficiency;
- (iv) Increasing the frequency of required reporting, testing, sampling or monitoring;
- (v) Adding to the amount of information required for records or reports;
- (vi) Decreasing the amount of time to come into compliance;
- (vii) Subjecting additional emission points or sources within a source category to control requirements; and
- (viii) Any adjustments allowed in a specific section 112 rule.

§ 63.93 Approval of State authorities that substitute for a section 112 rule.

Under this section a State may seek approval of State authorities which differ in form from a Federal section 112 rule for which they would substitute, such that the State authorities do not qualify for approval under § 63.92.

- (a) Approval process. (1) Within 45 days after receipt of a complete request for approval under this section, the Administrator will seek public comment for a minimum of 30 days on the State request for approval. The Administrator will require that comments be submitted concurrently to the State.
- (2) If, after review of public comments and any State responses to comments submitted to the Administrator within 30 days of the close of the public comment period, the Administrator finds that the criteria of this section and the criteria of § 63.91 are met, the State authorities will be approved by the Administrator under this section and the approved authorities will be published in the **Federal Register** and incorporated directly or by reference, in the appropriate subpart of part 63. Authorities approved under § 63.95 will be incorporated pursuant to requirements under section 112(r).
- (3) If the Administrator finds that any of the requirements of this section or § 63.91 have not been met, the Administrator will disapprove the State authorities.
- (4) Authorities submitted for approval under this section shall include either:
- (i) State rules or other requirements enforceable under State law that would substitute for a section 112 rule; or
- (ii) (A) The specific permit terms and conditions for the source or set of sources in the source category for which the State is requesting approval under this section, including control requirements and compliance and enforcement measures, that would substitute for the permit terms and conditions imposed by the otherwise applicable section 112 rule for such source or set of sources.

- (B) The Administrator will approve authorities specified under paragraph (a)(4)(ii)(A) of this section only when the State submitting the request already has an approved program under § 63.94, the Federal standard for the source category has been promulgated under section 112(h), and the Administrator has not determined the work practice, design, equipment or operational requirements submitted by the State to be inadequate under the provisions of the Federal standard.
- (5) Within 180 days of receiving a complete request for approval under this section, the Administrator will either approve or disapprove the State request.
- (b) Criteria for approval. Any request for approval under this section shall meet all of the criteria of this section and § 63.91 before approval. The State shall provide the Administrator with detailed documentation that the State authorities contain or demonstrate:
- (1) Applicability criteria that are no less stringent than those in the respective Federal rule:
- (2) Levels of control and compliance and enforcement measures that result in emission reductions from each affected source or accidental release prevention program requirements for each affected source that are no less stringent than would result from the otherwise applicable Federal rule;
- (3) A compliance schedule that assures that each affected source is in compliance no later than would be required by the otherwise applicable Federal rule; and
- (4) At a minimum, the approved State authorities must include the following compliance and enforcement measures. (For authorities addressing the accidental release prevention program, minimum compliance and enforcement provisions are described in § 63.95.)
- (i) The approved authorities must include a method for determining compliance.
- (ii) If a standard in the approved authorities is not instantaneous, a maximum averaging time must be established.
- (iii) The authorities must establish an obligation to periodically monitor or test for compliance using the method established per § 63.93(b)(4)(i) sufficient to yield reliable data that are representative of the source's compliance status.
- (iv) The results of all required monitoring or testing must be reported at least every 6 months.

§ 63.94 Approval of a State program that substitutes for section 112 emission standards.

Under this section a State may seek approval of a State program to be implemented and enforced in lieu of specified existing and future Federal emission standards or requirements promulgated under sections 112(d), (f) or (h), for those affected sources permitted by the State under part 70 of this chapter.

- (a) Approval process. (1) Within 45 days after receipt of a complete request for approval under this section the Administrator will seek public comment for a minimum of 30 days on the State request for approval. The Administrator will require that comments be submitted concurrently to the State.
- (2) If, after review of all public comments, and State responses to comments submitted to the Administrator within 30 days of the close of the public comment period, the Administrator finds that the criteria of this section and the criteria of § 63.91 are met, the State program will be approved by the Administrator. The approved State commitment made under paragraph (b)(2) of this section and reference to all documents submitted under § 63.91(b)(2) will be published in the **Federal Register** and incorporated directly or by reference in the appropriate subpart of part 63.

(3) If the Administrator finds that any of the criteria of this section or § 63.91 have not been met, the Administrator will disapprove the State program.

(4) Within 180 days of receiving a complete request for approval under this section, the Administrator will either approve or disapprove the State request.

- (b) Criteria for approval. Any request for approval under this section shall meet all of the criteria of this section and § 63.91 before approval. The State shall provide the Administrator with:
- (1) A reference to all specific sources or source categories listed pursuant to subsection 112(c) for which the State is seeking authority to implement and enforce standards or requirements under this section;
- (2) A legally binding commitment adopted through State law that, after approval:
- (i) For each source subject to Federal section 112 emission standards or requirements for which approval is sought, part 70 permits shall be issued or revised by the State in accordance with procedures established in part 70 of this chapter and in accordance with the schedule submitted under § 63.91(b)(5) assuring expeditious compliance by all sources; and
- (ii) All such issued or revised part 70 permits shall contain conditions that:
- (A) Reflect applicability criteria no less stringent than those in the otherwise applicable Federal standards or requirements;
- (B) Require levels of control for each affected source and emission point no less stringent than those contained in the otherwise applicable Federal standards or requirements;
- (C) Require compliance and enforcement measures for each source and emission point no less stringent than those in the otherwise applicable Federal standards or requirements;
- (D) Express levels of control and compliance and enforcement measures in the same form and units of measure as the otherwise applicable Federal standard or requirement;

(E) Assure compliance by each affected source no later than would be required by the otherwise applicable Federal standard or requirement.

§ 63.95 Additional approval criteria for accidental release prevention programs.

- (a) A State submission for approval of an Accidental Release Prevention (ARP) program must meet the criteria and be in accordance with the procedures of this section, § 63.91, and, where appropriate, either § 63.92 or § 63.93.
- (b) The State ARP program application shall contain the following elements consistent with the procedures in § 63.91 and, where appropriate, either § 63.92 or § 63.93:
- (1) A demonstration of the State's authority and resources to implement and enforce regulations which are at least as stringent as regulations promulgated under section 112(r) that specify substances, related thresholds and a risk management program,
 - (2) Procedures for:
- (i) Registration of stationary sources, as defined in section 112(r)(2)(C) of the Act, which clearly identifies the State entity to receive the registration;
- (ii) Receiving and reviewing risk management plans;
- (iii) Making available to the public any risk management plan submitted to the State pursuant to provisions specified in section 112(r) which are consistent with section 114(c) of the Act; and
- (iv) Providing technical assistance to subject sources, including small businesses:
- (3) A demonstration of the State's authority to enforce all accidental release prevention requirements including a risk management plan auditing strategy;
- (4) A description of the coordination mechanisms the State implementing agency will use with:
- (i) The Chemical Safety and Hazard Investigation Board, particularly during accident investigation; and
- (ii) The State Emergency Response Commission, and the Local Emergency Planning Committees; and
- (iii) The air permitting program with respect to sources subject to both section 112(r) of the Act and permit requirements under part 70 of this chapter.
- (c) A State may request approval for a complete or partial program. A partial accidental release prevention program must include the core program elements listed in paragraph (b) of this section.

§ 63.96 Review and withdrawal of approval.

(a) Submission of information for review of approval. (1) The Administrator may at any time request any of the following information to review the adequacy of implementation and enforcement of an approved rule or program and the State shall provide that information within 45 days of the Administrator's request:

- (i) Copies of any State statutes, rules, regulations or other requirements that have amended, repealed or revised the approved State rule or program since approval or since the immediately previous EPA review;
- (ii) Information to demonstrate adequate State enforcement and compliance monitoring activities with respect to all approved State rules and with all section 112 rules, emission standards or requirements;
- (iii) Information to demonstrate adequate funding, staff, and other resources to implement and enforce the State's approved rule or program;
- (iv) A schedule for implementing the State's approved rule or program that assures compliance with all section 112 rules and requirements that the EPA has promulgated since approval or since the immediately previous EPA review.
- (v) A list of part 70 or other permits issued, amended, revised, or revoked since approval or since immediately previous EPA review, for sources subject to a State rule or program approved under this subpart.
- (vi) A summary of enforcement actions by the State regarding violations of section 112 requirements, including but not limited to administrative orders and judicial and administrative complaints and settlements.
- (2) Upon request by the Administrator, the State shall demonstrate that each State rule, emission standard or requirement applied to an individual source is no less stringent as applied than the otherwise applicable Federal rule, emission standard or requirement.
- (b) Withdrawal of approval of a state rule or program. (1) If the Administrator has reason to believe that a State is not adequately implementing or enforcing an approved rule or program according to the criteria of this section or that an approved rule or program is not as stringent as the otherwise applicable Federal rule, emission standard or requirements, the Administrator will so inform the State in writing and will identify the reasons why the Administrator believes that the State's rule or program is not adequate. The State shall then initiate action to correct the deficiencies identified by the Administrator and shall inform the Administrator of the actions it has initiated and completed. If the Administrator determines that the State's actions are not adequate to correct the deficiencies, the Administrator will notify the State that the Administrator intends to withdraw approval and will hold a public hearing and seek public comment on the proposed withdrawal of approval. The Administrator will require that comments be submitted concurrently to the State. Upon notification of the intent to withdraw, the State will notify all sources subject to the relevant approved rule or program that withdrawal proceedings have been initiated.
- (2) Based on any public comment received and any response to that comment by the State, the Administrator will notify the State

- of any changes in identified deficiencies or actions needed to correct identified deficiencies. If the State does not correct the identified deficiencies within 90 days after receiving revised notice of deficiencies, the Administrator shall withdraw approval of the State's rule or program upon a determination that:
- (i) The State no longer has adequate authorities to assure compliance or resources to implement and enforce the approved rule or program, or
- (ii) The State is not adequately implementing or enforcing the approved rule or program, or
- (iii) An approved rule or program is not as stringent as the otherwise applicable Federal rule, emission standard or requirement.
- (3) The Administrator may withdraw approval for part of a rule, for a rule, for part of a program, or for an entire program.
- (4) Any State rule, program or portion of a State rule or program for which approval is withdrawn is no longer Federally enforceable. The Federal rule, emission standard or requirement that would have been applicable in the absence of approval under this subpart will be the federally enforceable rule, emission standard or requirement.
- (i) Upon withdrawal of approval, the Administrator will publish an expeditious schedule for sources subject to the previously approved State rule or program to come into compliance with applicable Federal requirements. Such schedule shall include interim emission limits where appropriate. During this transition, sources must be operated in a manner consistent with good air pollution control practices for minimizing emissions.
- (ii) Upon withdrawal, the State shall reopen, under the provisions of § 70.7(f) of this chapter, the part 70 permit of each source subject to the previously approved rules or programs in order to assure compliance through the permit with the applicable requirements for each source.
- (iii) If the Administrator withdraws approval of State rules applicable to sources that are not subject to part 70 permits, the applicable State rules are no longer Federally enforceable.
- (iv) If the Administrator withdraws approval of a portion of a State rule or program, other approved portions of the State rule or program that are not withdrawn shall remain in effect.
- (v) Any applicable Federal emission standard or requirement shall remain enforceable by the EPA as specified in section 112(1)(7) of the Act.
- (5) If a rule approved under § 63.93 is withdrawn under the provisions of § 63.96(b)(2) (i) or (ii), and, at the time of withdrawal, the Administrator finds the rule to be no less stringent than the otherwise applicable Federal requirement, the Administrator will grant equivalency to the

previously approved State rule under the appropriate provisions of this part.

- (6) A State may submit a new rule, program or portion of a rule or program for approval after the Administrator has withdrawn approval of the State's rule, program or portion of a rule or program. The Administrator will determine whether the new rule or program or portion of a rule or program is approvable according to the criteria and procedures of § 63.91 and either of § 63.92, § 63.93 or § 63.94.
- (7) A State may voluntarily withdraw from an approved State rule, program or portion of a rule or program by notifying the EPA and

- all affected sources subject to the rule or program and providing notice and opportunity for comment to the public within the State.
- (i) Upon voluntary withdrawal by a State, the Administrator will publish a timetable for sources subject to the previously approved State rule or program to come into compliance with applicable Federal requirements.
- (ii) Upon voluntary withdrawal, the State must reopen and revise the part 70 permits of all sources affected by the withdrawal as provided for in this section and § 70.7(f), and the Federal rule, emission standard, or requirement that would have been applicable
- in the absence of approval under this subpart will become the applicable requirement for the source.
- (iii) Any applicable Federal section 112 rule, emission standard or requirement shall remain enforceable by the EPA as specified in section 112(l)(7) of the Act.
- (iv) Voluntary withdrawal shall not be effective sooner than 180 days after the State notifies the EPA of its intent to voluntarily withdraw.

[FR Doc. 93–28821 Filed 11–24–93; 8:45 am] BILLING CODE 6560–50–P

Proposed Rules

Federal Register

Vol. 58, No. 226

Friday, November 26, 1993

This section of the FEDERAL REGISTER contains notices to the public of the proposed issuance of rules and regulations. The purpose of these notices is to give interested persons an opportunity to participate in the rule making prior to the adoption of the final rules.

ENVIRONMENTAL PROTECTION AGENCY

40 CFR Part 52

[CO20-1-5681; FRL-4798-9]

Clean Air Act Approval and Promulgation of Title V, Section 507, **Small Business Stationary Source Technical and Environmental** Compliance Assistance Program for the State of Colorado

AGENCY: Environmental Protection Agency (EPA).

ACTION: Notice of proposed rulemaking.

SUMMARY: The EPA proposes to approve the State Implementation Plan (SIP) revision submitted by the State of Colorado for the purpose of establishing a Small Business Stationary Source Technical and Environmental Compliance Assistance Program (PROGRAM). The implementation plan was submitted by the State to satisfy the Federal mandate, found in section 507 of the Clean Air Act (CAA), to ensure that small businesses have access to the technical assistance and regulatory information necessary to comply with the CAA. The rationale for the approval is set forth in this proposal; additional information is available at the address indicated below.

DATES: Comments on this proposed action must be received in writing by December 27, 1993.

ADDRESSES: Comments can be mailed to: Douglas M. Skie, Chief, Air Programs Branch, Environmental Protection Agency, Region VIII, Mail Code-8ART-AP, 999 18th Street, Suite 500, Denver, Colorado 80202-2405.

Copies of the State's submittal and EPA's technical support document are available for inspection during normal business hours at the following location: Air Programs Branch, Environmental Protection Agency, Region VIII 999 18th Street, Suite 500, Denver, Colorado 80202-2405.

FOR FURTHER INFORMATION CONTACT: Laura Farris, Mail Code-8ART-AP, Environmental Protection Agency Region

VIII, 999 18th Street, Suite 500, Denver, Colorado 80202-2405, (303) 294-7539.

SUPPLEMENTARY INFORMATION:

I. Background

Implementation of the provisions of the Clean Air Act (CAA), as amended in 1990, will require regulation of many small businesses so that areas may attain and maintain the National ambient air quality standards (NAAQS) and reduce the emission of air toxics. Small businesses frequently lack the technical expertise and financial resources necessary to evaluate such regulations and to determine the appropriate mechanisms for compliance. In anticipation of the impact of these requirements on small businesses, the CAA requires that States adopt a Small Business Stationary Source Technical and Environmental Compliance Assistance Program (PROGRAM), and submit this PROGRAM as a revision to the federally approved SIP. In addition, the CAA directs the Environmental Protection Agency (EPA) to oversee these small business assistance programs and report to Congress on their implementation. The requirements for establishing a PROGRAM are set out in Section 507 of Title V of the CAA. In February 1992, EPA issued Guidelines for the Implementation of Section 507 of the 1990 Clean Air Act Amendments, in order to delineate the Federal and State roles in meeting the new statutory provisions and as a tool to provide further guidance to the States on submitting acceptable SIP revisions.

The State of Colorado has submitted a SIP revision to EPA in order to satisfy the requirements of section 507. In order to gain full approval, the State submittal must provide for each of the following PROGRAM elements:

- (1) The establishment of a Small Business Assistance Program (SBAP) to provide technical and compliance assistance to small businesses;
- (2) The establishment of a State Small Business Ombudsman to represent the interests of small businesses in the regulatory process: and
- (3) The creation of a Compliance Advisory Panel (CAP) to determine and report on the overall effectiveness of the SBAP.

II. Analysis

The State of Colorado has met all of the requirements of section 507 by submitting a SIP revision that implements all required PROGRAM elements. At a public hearing held before the Colorado Air Quality Control Commission (Commission) on October 15, 1992, the plan for the PROGRAM was approved. A notice of approval dated October 30, 1992, was issued by the Technical

Secretary of the Commission adopting the PROGRAM plan. Legal authority to implement the PROGRAM was obtained through Senate Bill 105 and is contained in the Colorado Act (ACT) Section 25–7–109.2. The collection of fees to cover the cost of implementing the PROGRAM is contained in the ACT Section 25-7-114.7.

1. Small Business Assistance Program

Section 507(a) sets forth six requirements 1 that the State must meet to have an approvable SBAP. The State's PROGRAM meets these six requirements which are contained in eight SBAP elements in its proposed SIP revision. It also establishes a Task Force that will discuss these eight SBAP elements and choose some or all of the implementation methods listed for each element, or add additional implementation methods, as required. The Task Force consists of members from the Colorado Department of Health Air Pollution Control Division (Division), the Commission, the Office of Regulatory Reform, the CAP and other State entities which assist small businesses and small business associations.

The first requirement set forth in section 507(a) is to establish adequate mechanisms for developing, collecting and coordinating information concerning compliance methods and technologies for small business stationary sources, and programs to encourage lawful cooperation among such sources and other persons to further compliance with the Act. The State has met this requirement by committing in its proposed SIP revision to implement a SBAP element titled "Dissemination of compliance issues, regulation, monitoring, modeling, and control technology information." The proactive component of this element states the following: "General information regarding the services provided by the PROGRAM shall be developed," "The Division shall develop information regarding existing regulations that affect small business," and "The Division shall coordinate with the Ombudsman to develop a procedure through which the information can be developed as new regulations are proposed by the EPA or the State." The reactive component of this element states that "The SBAP shall refer calls as they are received to appropriate Division staff or other resources, depending on the type and complexity of information being sought."

The second requirement is to establish adequate mechanisms for assisting small business stationary sources with pollution

¹ A seventh requirement of section 507(a), establishment of an Ombudsman office, is discussed in the next section.

prevention and accidental release detection and prevention, including providing information concerning alternative technologies, process changes, products and methods of operation that help reduce air pollution. The State has met this requirement by committing in its proposed SIP revision to implement a SBAP element titled "Pollution prevention and accidental release assistance." This element states that "the Division shall compile a directory of Federal, State, Local, and trade group and other industry resources that can provide assistance in these areas," and "establish a central "clearinghouse" of information from these various entities."

The third requirement is to develop a compliance and technical assistance program for small business stationary sources which assists small businesses in determining applicable requirements and in receiving permits under the Act in a timely and efficient manner. The State has met this requirement by committing in its proposed SIP revision to implement a SBAP element titled "Dissemination of compliance issues, regulation, monitoring, modeling, and control technology information." In addition to the implementation methods listed under the first requirement above, the Division "shall develop information regarding small businesses' requirements for obtaining operating and construction permits," and 'will explain the requirements, ... and how a source can calculate their emissions, and ... determine their compliance status."

The fourth requirement is to develop adequate mechanisms to assure that small business stationary sources receive notice of their rights under the Act in such manner and form as to assure reasonably adequate time for such sources to evaluate compliance methods and any relevant or applicable proposed or final regulation or standards issued under the Act. The State has met this requirement by committing in its proposed SIP revision to implement a SBAP element titled "Notifying small businesses of their rights under the Act." This element states that when Federal or State air regulations are proposed that affect small businesses, the Division and Ombudsman "shall develop brochures and/or presentations for informing small businesses," and "obtain small business comments or to ensure that small businesses submit their comments to the EPA." They will further "develop and implement a procedure that contains specific timeframes for proceeding with development of information regarding a new requirement proposal," and "develop information explaining enforcement procedures and source's rights."

The fifth requirement is to develop adequate mechanisms for informing small business stationary sources of their obligations under the Act, including mechanisms for referring such sources to qualified auditors or, at the option of the State, for providing audits of the operations

of such sources to determine compliance with the Act. The State has met this requirement by committing in its proposed SIP revision to implement a SBAP element titled "Small business source auditor referrals or audit provisions." A list of auditors will be compiled as follows: "The Division shall consult with trade groups and other small business representatives, and shall solicit names of companies that these entities are familiar with and approve of," and "shall add to the list any companies ... dealt with in the past." "When compiling the list, the Division shall consider the company's ability to provide environmental audits, especially in regard to providing pollution prevention assistance." The Division does not plan to directly conduct audits of small businesses.

The sixth requirement is to develop procedures for consideration of requests from a small business stationary source for modification of: (A) Any work practice or technological method of compliance, or (B) the schedule of milestones for implementing such work practice or method of compliance preceding any applicable compliance date, based on the technological and financial capability of any such small business stationary source. The State has met this requirement by committing in its proposed SIP revision to implement a SBAP element titled "Modification of work practices or alternate control method requests." This element states that the Division shall develop a formal procedure for sources to follow when making such requests in cases where a SIP revision is not required. If a SIP revision is required, the Division shall "consult with the EPA Region VIII to establish the least intensive method of obtaining a SIP revision for these requests," and "develop a 'generic' alternative control procedure as part of the State SIP, where possible.'

2. Ombudsman

Section 507(a)(3) requires the designation of a State office to serve as the Ombudsman for small business stationary sources. The State has met this requirement by locating the position of the Small Business Ombudsman in the Office of Regulatory Reform.

3. Compliance Advisory Panel

Section 507(e) requires the State to establish a Compliance Advisory Panel (CAP) that must include two members selected by the Governor who are not owners or representatives of owners of small businesses; four members selected by the State legislature who are owners, or represent owners, of small businesses; and one member selected by the head of the agency in charge of the Air Pollution Permit Program. The State has met this requirement by outlining in its proposed SIP revision how the members will be determined, which is consistent with the method described above.

In addition to establishing the minimum membership of the CAP the CAA delineates

four responsibilities of the Panel: (1) To render advisory opinions concerning the effectiveness of the SBAP, difficulties encountered and the degree and severity of enforcement actions; (2) to periodically report to EPA concerning the SBAP's adherence to the principles of the Paperwork Reduction Act, the Equal Access to Justice Act, and the Regulatory Flexibility Act 2; (3) to review and assure that information for small business stationary sources is easily understandable; and (4) to develop and disseminate the reports and advisory opinions made through the SBAP. The State has met these requirements by outlining in its proposed SIP revision the functions of the CAP which are consistent with those stated above.

4. Eligibility

Section 507(c)(1) of the CAA defines the term "small business stationary source" as a stationary source that:

- (A) Is owned or operated by a person who employs 100 or fewer individuals;
- (B) Is a small business concern as defined in the Small Business Act;
 - (C) Is not a major stationary source;
- (D) Does not emit 50 tons per year (tpy) or more of any regulated pollutant; and
- (E) Emits less than 75 tpy of all regulated pollutants.

The State of Colorado has established a mechanism for ascertaining the eligibility of a source to receive assistance under the PROGRAM, including an evaluation of a source's eligibility using the criteria in Section 507(c)(1) of the CAA. The State of Colorado has also provided for public notice and comment on grants of eligibility to sources that do not meet the provisions of sections 507(c)(1)(C), (D), and (E) of the CAA but do not emit more than 100 tpy of all regulated pollutants, and for exclusion from the small business stationary source definition, after consultation with the EPA and the Small Business Administration Administrator and after providing notice and opportunity for public comment, of any category or subcategory of sources that the State determines to have sufficient technical and financial capabilities to meet the requirements of the CAA.

III. This Action

In this proposal, EPA is proposing to approve the SIP revision submitted by the State of Colorado.

The State of Colorado has submitted a SIP revision implementing each of the required PROGRAM elements required by section 507 of the CAA. The Small Business Ombudsman started in the Office of Regulatory Reform in

² Section 507(e)(1)(B) requires the CAP to report on the compliance of the SBAP with these three Federal statutes. However, since State agencies are not required to comply with them, EPA believes that the State PROGRAM must merely require the CAP to report on whether the SBAP is adhering to the general principles of these Federal statutes.

August, 1992. A schedule for implementation of the PROGRAM submitted in a letter by the State dated January 7, 1993, indicates that members of the CAP will be appointed by July 1, 1993, and a final report of the Task Force on implementation methods for the SBAP is scheduled for completion by October, 1993. EPA is therefore proposing to approve this submittal.

This action has been classified as a Table 2 Action by the Regional Administrator under the procedures published in the Federal Register on January 19, 1989 (54 FR 2214-2225). On January 6, 1989, the Office of Management and Budget (OMB) waived Table 2 and Table 3 SIP revisions from the requirement of section 3 of Executive Order 12291 for a period of two years. U.S. EPA has submitted a request for a permanent waiver for Table 2 and Table 3 SIP revisions. The OMB has agreed to continue the waiver until such time as it rules on U.S. EPA's request. This request continues in effect under Executive Order 12866 which superseded Executive Order 12291 on September 30,

Under the Regulatory Flexibility Act, 5 U.S.C. 600 *et seq.*, EPA must prepare a regulatory flexibility analysis assessing the impact of any proposed or final rule on small entities. 5 U.S.C. 603 and 604. Alternatively, EPA may certify that the rule will not have a significant impact on a substantial number of small entities. Small entities include small businesses, small not-for-profit enterprises, and government entities with jurisdiction over populations of less than 50,000.

In this action, EPA is approving a State program created for the purpose of assisting small businesses in complying with existing statutory and regulatory requirements. The program being approved does not impose any new regulatory burden on small businesses; it is a program under which small businesses may elect to take advantage of assistance provided by the State. Therefore, because the EPA's approval of this program does not impose any new regulatory requirements on small businesses, I certify that it does not have a significant economic impact on any small entities affected.

List of Subjects in 40 CFR Part 52

Environmental protection, Air pollution control, Incorporation by reference, Small business assistance program.

Authority: 42 U.S.C. 7401–7671q. Dated: October 19, 1993.

Patricia D. Hull,

Acting Regional Administrator.
[FR Doc. 93–29047 Filed 11–24–93; 8:45 am]
BILLING CODE 6560–50–F

40 CFR Part 52

[DE 11-1-5979; FRL-4800-1]

Approval and Promulgation of Air Quality Implementation Plans; Delaware; Reasonably Available Control Technology Requirements for Major Sources of Nitrogen Oxides

AGENCY: Environmental Protection Agency (FPA)

ACTION: Proposed rule.

SUMMARY: EPA is proposing a limited approval/limited disapproval of a State Implementation Plan (SIP) revision submitted by the State of Delaware. This revision establishes and requires all major sources of nitrogen oxides (NOx) to implement reasonably available control technology (RACT). The intended effect of this action is to propose approval of Delaware Regulation No. 12, Control of Nitrogen Oxides Emissions for the limited purpose of strenthening the Delaware SIP which currently has no requirements for existing NOx sources. In addition, this action is intended to propose disapproval of Delaware Regulation No. 12 for the limited purpose of allowing Delaware the opportunity to correct the deficiencies in the regulation which result in its failure to meet all requirements of the Clean Air Act (CAA). This action is being taken pursuant to section 110 of the Clean Air Act.

DATES: Comments must be received on or before December 27, 1993.

ADDRESSES: Comments may be mailed to Thomas J. Maslany, Director, Air, Radiation, and Toxics Division, U.S. Environmental Protection Agency, Region III, 841 Chestnut Building, Philadelphia, Pennsylvania 19107.

Copies of the documents relevant to this action are available for public inspection during normal business hours at the Air, Radiation, and Toxics Division, U.S. Environmental Protection Agency, Region III, 841 Chestnut Building, Philadelphia, Pennsylvania 19107; and Delaware Department of Natural Resources & Environmental Control, 89 Kings Highway, P.O. Box 1401, Dover, Delaware 19903.

FOR FURTHER INFORMATION CONTACT: Cynthia H. Stahl, (215) 597–9337, at the EPA Region III address indicated in the

ADDRESSES section.

SUPPLEMENTARY INFORMATION: On January 11, 1993, Delaware Department of Natural Resources and Environmental Control (DNREC) submitted a revision to its State Implementation Plan (SIP) for the control of NOx emissions from major sources. The revision consists of a new regulation which would require sources which emit or have the potential to emit 25 tons or more of NOx per year in New Castle and Kent Counties and 100 tons or more of NOx per year in Sussex County to comply with reasonably available

control technology requirements by May 31, 1995.

Background

Pursuant to section 182 of the Clean Air Act (CAA), Delaware is required to implement RACT for all major NOx sources by no later than May 31, 1995. The major source size in determined by its location, the classification of that area and whether it is located in the ozone transport region (OTR) which is established by the CAA. New Castle and Kent Counties in Delaware are part of the Philadelphia ozone nonattainment area and are classified as severe. Sussex County, Delaware is classified as marginal nonattainment but is located in the OTR on which the CAA imposes certain additional requirements, including the NOx RACT requirement for all major NOx sources. For New Castle and Kent Counties, section 182 requires that, among other requirements, all sources which emit or have the potential to emit 25 tons of NOx or more per year must implement RACT by no later than May 31, 1995. For Sussex County, sections 182 and 184 require that all sources which emit or have the potential to emit 100 tons of NOx or more per year implement RACT by May 31, 1995.

Delaware NOx RACT Regulation

Specific NOx Requirements

Delaware's proposed NOx RACT regulation contains specific emission limits for fuel burning equipment with a rated heat capacity of 100 million BTU (MMBTU)/hour or greater. Gas fired face and tangential units are required to meet an emission limit of 0.2 pounds of NOx per million BTU input. Oil or gas fired face and tangential units are required to meet an emission limit of 0.25 lbs of NOx per million BTU input. Oil or gas fired cyclones are required to meet an emission limit of 0.43 lbs of NOx per million BTU input. Dry bottom coal fired face and tangential units are required to meet an emission limit of 0.38 lbs of NOx per million BTU input. Dry bottom coal fired stokers are required to meet an emission limit of 0.40 lbs of NOx per million BTU input. All emission limits are required to be met on a 24 hour rolling averaging period using continuous emission monitoring (CEM) data approved by the Department. EPA is proposing to approve the above emission limits as RACT for fuel burning equipment with a rated heat input capacity of 100 MMBTU/hour or greater. The numerical emission limits are supported by data gathered by the State and Territorial Air Pollution Program Administrators (STAPPA) and Association of Local Air Pollution Control Officials (ALAPCO). The 24 hour rolling average compliance period is consistent with protection of the short-term ozone National Ambient Air Quality Standard (NAAQS).

EPA is, however, proposing to disapprove Delaware's proposed emission limit of 1.0 lbs of NOx per million BTU for wet bottom coal fired face and tangential units. There are no applicable sources in the State of Delaware, no data to support such an emission limit, and no reason to adopt such an emission standard. Any newly constructed wet bottom coal fired face and tangential units would be subject to the Delaware SIP's new source review requirements.

Delaware's regulation also contains specific emission limits for gas turbines. The emission limit for gas fired simple or combined cycle gas turbines, corrected to 15% oxygen, is 42 parts per million (ppm). The emission limit for liquid fired simple or combined cycle gas turbines, corrected to 15% oxygen, is 88 ppm. Both emission standards require compliance to be demonstrated using a one hour averaging period based on CEM data approved by the Department.

NOx Technology Requirements

The Delaware regulation, in Section 3.3 also requires fuel burning equipment with a rated heat capacity of less than 100 MMBTU/ hr to emit no more NOx than what would be emitted if either low excess air and low NOx burner technology or flue gas recirculation technology is used or some equivalent control technology approved by the Department and EPA. Section 3.4 requires stationary internal combustion engines to limit their emissions to no more than those emitted when using preignition chamber combustion (also referred to as clean burn technology) for gas fired units and those emitted when using lean burn technology for diesel fired units. As in Section 3.3, these sources may also meet an equivalent control technology which is approved by the Department and EPA.

Source-specific Requirements

All other major sources of NOx are required to notify the Department of their subject status, submit a proposal as to what constitutes RACT for the source including technical and economic support documentation, and provide a schedule, acceptable to the Department, to implement RACT no later than May 31, 1995. The notification to the Department, submittal of a RACT proposal and schedule must be made no later than November 15, 1993.

Alternative Requirements

Section 5 of the Delaware regulation allows sources to obtain alternative and equivalent RACT determinations all of which require EPA approval as a SIP revision. Included in these alternative and equivalent RACT determinations is a specific provision to allow averaging of emissions at multiple installations to meet applicable RACT emission limits if those installations have a common owner.

EPA Analysis

EPA is proposing a limited approval of Regulation No. 12 into the Delaware SIP, which was submitted on January 11, 1993. EPA is also proposing to disapprove Delaware Regulation No. 12 for the limited purpose of allowing Delaware the opportunity to correct certain deficiencies in Regulation No. 12. These deficiencies are located in Section 2 (definition of potential to emit), Section 3.2 (emission limit for wet bottom coal fired face and tangential units), Section 4.1 a. through f. (pertaining to exemptions for certain sources and compliance extensions), and Section 7.4 b. (use of alternative test methods). EPA is soliciting public comments on the issues discussed in this document or on other relevant matters. These comments will be considered before taking final action. Interested parties may participate in the Federal rulemaking procedure by submitting written comments to the EPA Regional office listed in the ADDRESSES section of this document.

EPA requires that states determine potential emissions by using the maximum emission rate, production capacity and hours of operation unless the source is restricted by federally enforceable conditions. The Delaware definition of "potential to emit", located in Section 2, is deficient in that it does not require that the emissions limiting conditions be federally enforceable. Therefore, EPA is proposing to disapprove this definition to the extent that it does not require federally enforceable emissions limiting conditions.

EPA is proposing to disapprove the exemptions contained in Section 4.1 a. through f. Section 4.1 a. and b. would allow exemptions of fuel burning equipment used primarily for residential heating and incinerators or oxidizers built before November 15, 1992 and used primarily for the control of air pollution. The exemptions contained in Section 4.1 c. and d. pertain to fuel burning units with a rated heat capacity of less than 15 MMBTU/hr and stationary internal combustion engines with a rated capacity of less than 450 hp of output power, respectively. Delaware has not provided any support information to indicate what population of sources these two exemptions would represent or what the total emissions from these sources are compared to the total NOx emissions in the state. Section 4.1 e. would allow sources which operate exclusively during the months of November through the end of March to be exempt from RACT requirements. There is, however, no requirement in Regulation No. 12, or elsewhere in the Delaware regulations which would apply to this regulation, for these sources to keep the necessary records to prove their exempt status. Therefore, EPA is proposing to disapprove this exemption.

If Delaware wishes to allow exemptions of major NOx sources from RACT requirements, there is a mechanism to do so

under section 182(f) of the Clean Air Act. Under section 182(f), any person, including a state, is allowed to petition EPA so that if the petition were granted, some or all of the Act's Title I NOx requirements would not apply. Under section 182(f)(1)(B), this petition can only be granted if the NOx reductions do not provide a net air quality benefit and if those emission reductions could be considered excess. EPA interprets the term air quality to include both health and welfare effects that may be directly or indirectly related to NOx emissions. Therefore, these impacts include ozone, nitrogen dioxide, and particulate matter formation, visibility impariment, acid deposition, air toxics formation, and nitrogen deposition in nutrient-sensitive areas. The net air quality benefit test requires that a modeling demonstration with and without the proposed NOx emission reductions be performed for the applicable area. The petitioner must show that there is a detrimental air quality effect with the control of NOx emissions, not simply that there is no effect on air quality.

EPA is also proposing to disapprove
Section 7.4 b. which allows alternative test
methods to be approved by the Department
without EPA approval. Since the stringency
of the emission standards and other RACT
requirements are dependent on compliance
using EPA approved test methods, EPA
cannot agree to allow Delaware to approve
alternative test methods without knowing how
those alternative test methods may affect
RACT.

EPA has evaluated Delaware's submitted Regulation No. 12 for consistency with the CAA and EPA regulations, and has found that the revisions will require emission reductions from a population of sources which were previously unregulated. The Delaware regulation, as noted above, also contains certain deficiencies which result in enforceability problems and in the regulation of a smaller population of sources than required by the CAA. Further discussion can be found in the accompanying technical support document.

Because of the deficiencies discussed above, EPA cannot grant full approval of this rule under section 110(k)(3) and part D. Also, because the submitted rule is not composed of separable parts which meet all the applicable requirements of the CAA, EPA cannot grant partial approval of the rule(s) under section 110(k)(3). However, EPA may grant a limited approval of the submitted rule under section 110(k)(3) in light of EPA's authority pursuant to section 301(a) to adopt regulations necessary to further air quality by strengthening the SIP. The approval is limited because EPA's action also contains a simultaneous limited disapproval, due to the fact that the rule does not meet the section 182(a)(2)(A) requirement of part D because of the noted deficiencies. Thus, in order to strengthen the SIP, EPA is proposing a limited approval of Delaware's submitted

Regulation No. 12 under section 110(k)(3) and 301(a) of the CAA.

At the same time, EPA is also proposing a limited disapproval of this rule because it contains deficiencies which result in less than the full population of required NOx sources to be regulated and enforceability problems, and, as such, the rule does not fully meet the requirements of Part D of the Act. Under section 179(a)(2), if the Administrator disapproves a submission under section 110(k) for an area designated nonattainment, based on the submission's failure to meet one or more of the elements required by the Act, the Administrator must apply one of the sanctions set forth in section 179(b) unless the deficiency has been corrected within 18 months of such disapproval. Section 179(b) provides two sanctions available to the Administrator: highway funding and offsets. The 18 month period referred to in section 179(a) will begin at the time EPA publishes final notice of this disapproval. Moreover, the final disapproval triggers the 24-month Federal implementation plan (FIP) requirement under section 110(c).

Proposed Action

For the above stated reasons, EPA is proposing a limited approval/limited disapproval of Delaware Regulation No. 12. In order to correct the deficiencies in Regulation No. 12 which EPA is proposing as a limited disapproval, Delaware must successfully petition EPA under section 182(f) of the Act if it wishes to exempt any major NOx sources from RACT requirements or delete the exemptions, correct the definition of potential to emit to require Federal enforceability of restricting operating conditions, provide adequate technical support for the proposed emission limit for wet bottom coal fired face and tangential units or delete the emission limit, require adequate recordkeeping requirements for all NOx sources, including those which claim exempt status, and require EPA approval for alternative test methods.

Nothing in this action should be construed as permitting or allowing or establishing a precedent for any future request for revision to any State Implementation Plan. Each request for revision to the State Implementation Plan shall be considered separately in light of specific technical, economic, and environmental factors and in relation to relevant statutory and regulatory requirements.

Under the Regulatory Flexibility Act, 5 U.S.C. 600 *et seq.*, EPA must prepare a regulatory flexibility analysis assessing the impact of any proposed or final rule on small entities. 5 U.S.C. 603 and 604. Alternatively, EPA may certify that the rule will not have a significant impact on a substantial number of small entities. Small entities include small businesses, small not-for-profit enterprises, and government entities with jurisdiction over populations of less than 50,000.

SIP approvals under sections 110 and 301, and subchapter I, part D of the CAA do not create any new requirements but simply approve requirements that the State is already imposing. Therefore, because the Federal SIP approval does not impose any new requirements, I certify that it does not have a significant impact on any small entities affected. Moreover, due to the nature of the Federal-State relationship under the CAA, preparation of a flexibility analysis would constitute Federal inquiry into the economic reasonableness of state action. The Clean Air Act forbids EPA to base its actions concerning SIPs on such grounds. Union Electric Co. v. U.S. EPA, 427 U.S. 246, 255-66 (1976); 42 U.S.C. 7410(a)(2).

EPA's disapproval of the State request under section 110 and subchapter I, part D of the CAA does not affect any existing requirements applicable to small entities. Any pre-existing Federal requirements remain in place after this disapproval. Federal disapproval of the state submittal does not affect its state-enforceability. Moreover, EPA's disapproval of the submittal does not impose any new Federal requirements. Therefore, EPA certifies that this disapproval action does not have a significant impact on a substantial number of small entities because it does not remove existing requirements and impose any new Federal requirements.

This action has been classified as a Table 2 Action by the Regional Administrator under the procedures published in the Federal Register on January 19, 1989 (54 FR 2214-2225). On January 6, 1989, the Office of Management and Budget (OMB) waived Table 2 and Table 3 SIP revisions from the requirement of section 3 of Executive Order 12291 for a period of two years. U.S. EPA has submitted a request for a permanent waiver for Table 2 and Table 3 SIP revisions. The OMB has agreed to continue the waiver until such time as it rules on U.S. EPA's request. This request continues in effect under Executive Order 12866 which superseded Executive Order 12291 on September 30, 1993.

The Administrator's decision to approve or disapprove the SIP revision will be based on whether it meets the requirements of Section 110(a)(2)(A)-(K), and part D of the Clean Air Act, as amended, and EPA regulations in 40 CFR part 51.

List of Subjects in 40 CFR Part 52

Environmental protection, Air pollution control, Intergovernmental relations, Nitrogen dioxide, Ozone, Reporting and recordkeeping requirements.

Authority: 42 U.S.C. 7401–7671q Dated: September 16, 1993.

W.T. Wisniewski,

Acting Regional Administrator, Region III. [FR Doc. 93–29045 Filed 11–24–93; 8:45 am] BILLING CODE 6560–50–F

40 CFR Part 52

[IL67-1-5721; FRL-4799-8]

Approval and Promulgation of an Implementation Plan for Vehicle Miles Traveled (VMT) Offsets; Illinois

AGENCY: U.S. Environmental Protection

Agency.

ACTION: Proposed rule.

SUMMARY: The United States Environmental Protection Agency (U.S. EPA) proposes to conditionally approve a request for a State Implementation Plan (SIP) revision, addressing the Chicago ozone nonattainment area, submitted by the State of Illinois for the purpose of offsetting any growth in emissions from growth in vehicle miles traveled (VMT) or number of vehicle trips, and to attain reduction in motor vehicle emissions, in combination with other emission reduction requirements, as necessary to comply with Reasonable Further Progress (RFP) milestones and attainment requirements of the Clean Air Act (Act). The implementation plan request was submitted by the State of Illinois to satisfy the statutory mandate that the State submit a SIP revision which identifies and adopts specific enforceable transportation control measures (TCMs) to offset any growth in emissions from growth in VMT or number of vehicle trips in severe and extreme ozone nonattainment areas. The rationale for the proposed conditional approval is set forth in this document; additional information is available at the address indicated in the ADDRESSES section.

DATES: Comments on this proposed action must be received in writing by December 27, 1993.

ADDRESSES: Written comments should be mailed to: J. Elmer Bortzer, Chief, Regulation Development Section, Regulation Development Branch (AR–18J), U.S. Environmental Protection Agency, Region 5, 77 West Jackson Boulevard, Chicago, Illinois, 60604.

Copies of the State's submittal and U.S. EPA's technical support document are available for inspection during normal business hours at the following location: Regulation Development Section, Regulation Development Branch (AR–18J), U.S. Environmental Protection Agency, Region 5, 77 West Jackson Boulevard, Chicago, Illinois 60604.

FOR FURTHER INFORMATION CONTACT:

Patricia Morris, Regulation Development Section, Regulation Development Branch (AR–18J), U.S. Environmental Protection Agency, Region 5, 77 West Jackson Boulevard, Chicago, Illinois 60604, (312) 353–8656.

SUPPLEMENTARY INFORMATION:

I. Background

Section 182(d)(1)(A) of the Act as amended in 1990 requires States containing

ozone nonattainment areas classified as "severe" pursuant to section 181(a) of the Act to adopt transportation control measures and transportation control strategies to offset any growth in emissions from growth in VMT or number of vehicle trips, and to attain reductions in motor vehicle emissions (in combination with other emission reduction requirements) as necessary to comply with the Act's RFP milestones. The requirements for establishing a VMT Offset program are discussed in the April 16, 1992, General Preamble to Title I of the Act (57 FR 13498), in addition to section 182(d)(1)(A) of the Act.

Section 110(k) of the Act contains provisions governing U.S. EPA's action on VMT Offset SIP submittals. Once found to be complete (or deemed complete by the passage of time), this section authorizes U.S. EPA to take one of three actions on SIP submittals. As provided by section 110(k)(3), if the submittal satisfactorily addresses all of the required VMT Offset elements, the U.S. EPA will grant full approval. U.S. EPA may grant a conditional approval of the submission under section 110(k)(4), however, if the submittal contains: (1) A commitment from the Governor or the Governor's designee to take the required actions; (2) a schedule establishing a date certain for taking the required actions, with the date not being later than one year from the time U.S. EPA will issue a final conditional approval; and (3) evidence that a public hearing was held on the commitments. See July 22, 1992 memorandum from Michael M. Shapiro, Deputy Assistant Administrator for Air and Radiation, entitled "Guidelines for State Implementation Plan (SIP) Submittals Due November 15, 1992." Finally, if the submittal fails to adequately address or commit to address one or more of the mandatory VMT Offset elements, the U.S. EPA must issue a disapproval. A State plan may therefore be approved, conditionally approved or disapproved.

In order to satisfy the applicable requirements, and to gain full approval, section 182(d)(1)(A) requires that a State VMT Offset SIP submittal must:

- (1) Identify and adopt specific enforceable transportation control strategies and TCMs to offset any growth in emissions from growth in VMT or number of vehicle trips;
- (2) Identify and adopt specific enforceable transportation control strategies and TCMs that attain reductions in motor vehicle emissions (in combination with other emission reduction requirements) as necessary to comply with RFP milestones;
- (3) Consider the measures specified in section 108(f) of the Act and choose from among and implement such measures as necessary to demonstrate attainment with the National ambient air quality standards; and
- (4) Ensure adequate access to downtown, other commercial and residential areas, and that emissions and congestion are reduced rather than relocated.

II. Analysis

The following discussion provides more information on the requirements for full approval, and Illinois' efforts to meet these requirements. Section 182(d)(1)(A) sets forth four requirements that must be met by a VMT Offset SIP. The first and second requirements mandate that the State submit a revision, by November 15, 1992 that identifies and adopts specific enforceable transportation control strategies and TCMs to offset any growth in emissions from growth in VMT or number of vehicle trips in affected areas, and attains reduction in motor vehicle emissions as necessary in combination with other emission reduction requirements of Act title I, part D, subpart II to comply with the requirements of section 182(d)(1)(A). These requirements are to ensure that projected motor vehicle volatile organic compound (VOC) emissions will never be greater during the ozone season in any given year than during the preceding year's ozone season. When growth in VMT or vehicle trips would otherwise cause a motor vehicle emissions upturn, that upturn must be prevented. The emissions level at the point of potential upturn becomes a ceiling on motor vehicle emissions. Motor vehicle VOC emission projections for meeting this requirement will be determined consistent with U.S. EPA's section 187 VMT forecasting and tracking guidance.

It should be noted that while the above requirements are simple in concept, their application could encourage areas to delay VMT or emissions reduction measures suitable for use as offsets until the trend in motor vehicle emissions reaches its minimum point and is about to turn upwards. To implement the VMT offset provision while avoiding this counterproductive incentive for delay, U.S. EPA looks for State compliance with the following approach: if projected motor vehicle emissions during the ozone season in one year are not higher than during the ozone season the year before, given the control measures in the SIP, the VMT offset requirement is satisfied. However, if the State plans to implement control measures over and above those specifically required by the Act and those required to demonstrate RFP and attainment, earlier than would be necessary and sufficient to prevent an emissions upturn, a projected subsequent growth-related increase to the level of emissions that would occur if these measures were scheduled later will not be considered to violate the requirement to offset emissions due to growth in VMT or vehicle trips. The latter situation will be viewed as a temporary reduction in emissions to a level below that required by the provision, rather than an increase above the required level, with no effect on emissions at or after the point at which offsetting measures become essential to compliance.

The U.S. EPA will approve a SIP revision as meeting this provision despite a forecasted upturn in vehicle emissions, as long as motor

vehicle VOC emissions in the ozone season of a given year do not exceed a ceiling level which reflects a hypothetical strategy of implementing otherwise specifically required measures on schedule and saving offset measures until the point at which VMT growth would otherwise cause an emission upturn. The ceiling level is therefore defined (up to the point of upturn) as motor vehicle emissions that would occur in the ozone season of that year, with VMT growth, if all measures for that area in that year were implemented as required by the Act. When this curve begins to turn up due to growth in VMT or vehicle trips, the ceiling becomes a fixed value. The ceiling line would include the effects of Federal measures such as new motor vehicle standards, Phase II Reid Vapor Pressure controls, and reformulated gasoline, as well as Act mandated SIP requirements such as enhanced inspection and maintenance, the fleet clean-fuel vehicle program, and the employer commute options program. The ceiling line would also include the effect of forecasted growth in VMT and vehicle trips in the absence of new discretionary measures to reduce them. The ceiling line must, in combination with projected emissions from nonvehicle sources, satisfy the RFP requirements for the area. Any VMT reduction measures or other actions to reduce motor vehicle emissions adopted since November 15, 1990 and not specifically required for the area by another provision of the Act would not be included in the calculation of the ceiling line.

Forecasted motor vehicle emissions must be held at or below the minimum level of the ceiling line after the ceiling line reaches its minimum level. If an area implements offset measures early, the forecasted emissions will be less than the ceiling line, and forecasted motor vehicle emissions could increase from one year to the next, as long as forecasted emissions never exceed the ceiling line.

These emissions offset requirements apply to projected emissions during the years between submission of the SIP revision and the statutory ozone national ambient air quality standard (NAAQS) attainment deadline. The State has not met these requirements but is in the process of forecasting future travel and emissions levels and has committed to implement these requirements by adopting specific enforceable measures within one year of U.S. EPA conditional approval.

The third requirement is that the State shall consider transportation control measures specified in section 108(f) and choose from among and implement such measures as necessary to demonstrate attainment with the NAAQS. The State has not met this requirement, but is assessing TCM strategies to offset any growth in emissions from growth in VMT or number of vehicle trips and has committed to implement this requirement by adopting specific enforceable

measures within one year of U.S. EPA conditional approval.

The fourth requirement is that the State should ensure adequate access to downtown, other commercial, and residential areas and should avoid measures that increase or relocate emissions and congestion rather than reduce them. The State has not met this requirement, but is making progress towards forecasting future emissions that would occur from implementing specific TCMs, and has committed to meet this requirement by adopting specific enforceable measures within one year of U.S. EPA conditional approval.

III. Analysis of State Submittal

The State of Illinois has not submitted a SIP revision implementing all the requirements relevant to the nonattainment classification of the State's nonattainment areas contained in section 182(d)(1)(A) of the Act. Illinois has, however, set forth the process for evaluating and selecting TCMs for implementation. The Illinois Environmental Protection Agency (IEPA) has worked with the Illinois Department of Transportation and the Chicago Area Transportation Study (the metropolitan planning organization (MPO) for the Chicago area) to set up a number of committees and taskforces to review and evaluate TCMs. An Air Quality Executive Committee (chaired by IEPA) is providing overall direction and integration, and seven taskforces are addressing air quality issues, including three new taskforces which have been formed to address environmental quality, conformity and employee commute options.

Illinois describes a 3 step process for evaluating TCMs through coordination of activities as specified by section 174 of the Act for transportation planning activities. The CATS, the IDOT and the IEPA have agreed to coordinate activities. The first step, in the three step process, is to determine the acceptability of a TCM through a series of white papers. The second step is to assess the implementability of the TCM. The third step is to evaluate the results of steps one and two. This process is to be completed by November 15, 1993.

The Illinois submittal contains interim dates for completion of work toward the VMT offset SIP. Specifically, the submittal states: all measures pertaining to step 1, step 2 and step 3 will be completed by November 15, 1992, June 30, 1993 and November 15, 1993. In completion of step one, a document entitled "Transportation Control Measures Committal for the State Implementation Plan" dated November 5, 1992 has been made available to the public.

Illinois has been actively participating in the Lake Michigan Ozone Control Program and in the urban air shed modeling for the area. Because of active participation, U.S. EPA expects the attainment demonstration for the Lake Michigan area and specifically the Chicago nonattainment area to be completed by November 15, 1994 and include TCM's as necessary to meet attainment.

Illinois has made commitments to adopt each of the required elements, and has provided a schedule with specific dates for additional work products and a process that will lead to adoption of enforceable measures, that will not extend over one year from the date of final conditional approval, and has subjected the commitments and schedule to a public hearing on October 27, 1992. Illinois has committed to adopt and submit specific enforceable measures to satisfy the requirements of section 182(d)(1)(A) by November 15, 1994. Illinois committed to this date in a letter dated October 13, 1993, which is part of the administrative record of this approval.

U.S. EPA believes that VMT Offset submittals such as Illinois' are appropriate for conditional approval considerations because section 182(d)(1)(A) of the Act requires that specific, enforceable measures identified and adopted by the State be submitted by November 15, 1992 along with a demonstration that they are adequate to offset any growth in emissions from growth in VMT or number of vehicle trips, which U.S. EPA interprets to mean adequate to hold vehicle emissions within the emissions ceiling described in the General Preamble to Title I, Section III. (A)(5)(d) (57 FR 13498, 13521-13523). The Act also states that these measures, beyond offsetting growth in emissions, shall be sufficient to allow total area emissions to comply with the RFP and attainment requirements. These requirements create a timing problem. Ozone nonattainment areas affected by this provision are not otherwise required to submit SIP demonstrations which would predict attainment of the 1996 RFP milestone until November 15, 1993 and likewise are not required to demonstrate post-1996 RFP and attainment until November 15, 1994. U.S. EPA does not believe that Congress intended the offset growth provisions to advance the dates for these broader submissions. Even without the requirement that the offset growth measures be sufficient to allow overall RFP and attainment in conjunction with other measures, U.S. EPA believes that the November 15, 1992 date might not allow the States sufficient time to develop a set of measures that would comply with the offset growth provisions over the long term.

To solve this timing problem and to allow a more coordinated and comprehensive planning process, U.S. EPA may accept and conditionally approve committal SIP revisions for the offset growth requirements under the authority of section 110(k)(4) of the Act. This could allow States one (1) year from the date of U.S. EPA's final conditional approval of the committal SIP revision, but not beyond November 15, 1994 to submit the full revision containing sufficient measures in specific and enforceable form.

IV. Proposed Rulemaking Action

The U.S. EPA proposes to conditionally approve the requested SIP revision commitment submitted by the State of Illinois

If the U.S. EPA conditionally approves the State's commitment, the State must meet its commitment to adopt the specific enforceable VMT requirements and submit these rules to U.S. EPA by November 15, 1994. If the State fails to adopt or submit any rules to U.S. EPA within this time frame, this conditional approval will become a disapproval upon U.S. EPA notification of the State by letter. At that time, this commitment will no longer be a part of the approved Illinois SIP. The U.S. EPA subsequently will publish a document in the Federal Register indicating that the commitment has been disapproved and removed from the SIP. If the State adopts and submits these rules to U.S. EPA within the applicable time frame, the conditionally approved commitment will remain a part of the SIP until the U.S. EPA takes final action approving or disapproving the new submittal. If the commitment is associated with existing rules and the U.S. EPA disapproves the subsequent submittal, the rules on which the conditional approval was based will also be disapproved at that time. If U.S. EPA approves the subsequent submittal, those newly approved rules will become a part of the SIP.

If the U.S. EPA issues a final disapproval, or if the conditional approval is converted to a disapproval, the sanctions clock under section 179(a) will begin. This clock will begin at the time U.S. EPA issues the final disapproval or at the time U.S. EPA notifies the State by letter that a conditional approval has been converted to a disapproval. If the State does not submit the implementation plan on which the disapproval was based within 18 months of the disapproval, U.S. EPA must impose one of the sanctions under section 179(b)—highway funding restrictions or the offset sanction. In addition, the final disapproval triggers the Federal Implementation Plan (FIP) requirement under section 110(c). Finally, under section 110(m), U.S. EPA has discretionary authority to impose sanctions at any time after a final disapproval or conversion of conditional approval to disapproval. Furthermore, if U.S. EPA does not approve the implementation plan revision request, the 18 month clock will begin where it stopped.

This action has been classified as a Table 2 Action by the Regional Administrator under the procedures published in the **Federal Register** on January 19, 1989 (54 FR 2214–2225). On January 6, 1989, the Office of Management and Budget (OMB) waived Table 2 and Table 3 SIP revisions from the requirement of section 3 of Executive Order 12291 for a period of two years. U.S. EPA has submitted a request for a permanent waiver for Table 2 and Table 3 SIP revisions.

The OMB has agreed to continue the waiver until such time as it rules on U.S. EPA's request. This request continues in effect under Executive Order 12866 which superseded Executive Order 12291 on September 30, 1993.

Under the Regulatory Flexibility Act, 5 U.S.C. 600 *et seq.*, the U.S. EPA must prepare a regulatory flexibility analysis assessing the impact of any proposed or final rule on small entities. 5 U.S.C. 603 and 604. Alternatively, U.S. EPA may certify that the rule will not have a significant impact on a substantial number of small entities. Small entities include small businesses, small not-for-profit enterprises, and government entities with jurisdiction over populations of less than 50,000.

SIP approvals under section 110 and subchapter I, part D of the Act do not create

any new requirements, but simply approve requirements that the State is already imposing. Therefore, because the Federal SIP-approval does not impose any new requirements, I certify that it does not have a significant impact on small entities affected. Moreover, due to the nature of the Federal-State relationship under the Act, preparation of a regulatory flexibility analysis would constitute Federal inquiry into the economic reasonableness of state action. The Act forbids U.S. EPA to base its actions concerning SIPs on such grounds. See *Union Electric CO. v. U.S. E.P.A.*, 427 U.S. 246, 256–66 (S. Ct. 1976); 42 U.S.C. 7410(a)(2).

If the conditional approval is converted to disapproval under section 110(k) of the Act, based upon the State's failure to meet the commitment, it will not affect any existing state requirements applicable to small entities.

Federal disapproval of the State submittal does not affect its state-enforceability. Moreover, U.S. EPA's disapproval of the submittal does not impose a new Federal requirement. Therefore, U.S. EPA certifies that this potential disapproval does not have a significant impact on a substantial number of small entities because it does not remove existing state requirements nor does it substitute a new Federal requirement.

List of Subjects in 40 CFR Part 52

Environmental protection, Air pollution control, Ozone.

Authority: 42 U.S.C. 7401–7671q. Dated: October 21, 1993.

Valdas V. Adamkus,

Regional Administrator.

[FR Doc. 93-29046 Filed 11-24-93; 8:45 am]

BILLING CODE 6560-50-F

Notices

Federal Register

Vol. 58, No. 226

Friday, November 26, 1993

This section of the FEDERAL REGISTER contains documents other than rules or proposed rules that are applicable to the public. Notices of hearings and investigations, committee meetings, agency decisions and rulings, delegations of authority, filing of petitions and applications and agency statements of organization and functions are examples of documents appearing in this section.

ENVIRONMENTAL PROTECTION AGENCY

[ER-FRL-4705-8]

Environmental Impact Statements; Notice of Availability

Responsible Agency: Office of Federal Activities, General Information (202) 260–5076 OR (202) 260–5075. Weekly receipt of Environmental Impact Statements Filed November 15, 1993 Through November 19, 1993 Pursuant to 40 CFR 1506.9.

- EIS No. 930497, Draft Supplement, SFW, Federal Aid in Sports Fish and Wildlife Restoration Programs Operation and Management Programmatic EIS, Updated Information, Implementation and Funding, Due: January 17, 1994, Contact: Columbus H. Brown (703) 358–2156.
- EIS No. 930408, Draft Supplement, AFS, MT, Lewis and Clark National Forest 1986–90 Noxious Weed Control Program, Updated Information, Implementation, several Counties, MT, Due: January 10, 1994, Contact: Bob Casey (406) 791–7700.
- EIS No. 930409, Draft EIS, AFS, MT, Berray Mountain Timber Sale, Harvesting Timber, Kootenai National Forest, Cabinet Ranger District, Sanders County, MT, Due: January 10, 1994, Contact: Dave Clay (406) 827–3533.
- EIS No. 930410, Final EIS, APH, Medfly (Mediterranean Fruit Fly) Cooperative Eradication Program, Implementation AL, AZ, CA, FL, GA, LA, MS, SC and TX, Due: December 27, 1993, Contact: Harold T. Smith (301) 436–8963.
- EIS No. 930411, Final EIS, FRC, CA, Lower Mokelumne River Hydroelectric Project Modifications, Licensing, (FERC. No. 29116–004), Parts of Pardee and Camanhe Dams, Mokelumne River, CA, Due: December 27, 1993, Contact: John A. Schnagl (202) 219–2661.
- EIS No. 930412, Final EIS, USA, NM, AZ, UT, Woodbridge Research Facility (WRF) Electromagnetic Pulse Simulators Relocation Project, Construction and Operation, Sites Selected for Six potential locations, Dugway Proving Ground, UT;

Yuma Proving Ground, AZ; and the remaining (4) in White Sands Missile Range, NM, Due: December 27, 1993, Contact: Eugene Patrick (703) 490–2110.

- EIS No. 930413, Draft Supplement, AFS, CA, Casa-Guard Timber Sale, Timber Harvesting, Updated Information concerning Impact on the California Spotted Owl and Fish Creek Watershed and Reforestation, Sequoia National Forest, Cannel Meadow Ranger District, Tulare County, CA, Due: January 10, 1994, Contact: Sandra Keys (209) 784–1500.
- EIS No. 930414, Draft Supplement, AFS, CA, Cottonwood and Gulf Timber Sales, Timber Harvesting in the Breckenridge Compartment, Updated Information Concerning Withdrawal of the Golf Timber Sale and Impacts on the California Spotted Owl and Reforestation for the Cottonwood Timber Sale, Sequoia National Forest, Greenhorn Ranger District, Kern County, CA, Due: January 10, 1994, Contact: Sandra Kay (209) 784–1500.
- EIS No. 930415, Draft EIS, NRC, LA, Claiborne Uranium Enrichment Center, Construction and Operation, (NUREG– 1482), NPDES Permit and Licensing, Homer, Claiborne Parish, LA, Due: January 10, 1994, Contact: Merri Horn (301) 504–2606.
- EIS No. 930416, Final Eis, IBR, CA, Los Vaqueros Water Quality and Reliability Project, Implementation, Section 10 and 404 Permits and Possible NPDES Permit, Contra Costa Water District, Contra Costa County, CA, Due: December 27, 1993, Contact: Doug Kleinsmith (916) 978–5129.
- EIS No. 930417, Draft EIS, NPS, ID, City of Rocks National Reserve, Comprehensive Management Plan and Development Concept Plan, Implementation, Cassia County, ID, Due: February 1, 1994, Contact: Mary Magee (303) 969–2210.
- EIS No. 930418, Draft EIS, AFS, CO, Illonis Creek Timber Sale, Harvesting Timber, Implementation, Amended Land and Resource Management Plan, Grand Mesa, Uncompander and Gunnison National Forests, Gunnison County, CO, Due: January 14, 1994, Contact: Jeff Ulrich (303) 874–6648.
- EIS No. 930419, Draft EIS, AFS, NM, Diamond Bar Allotment Management Plan, Implementation, Gila National Forest, Mimbres Ranger District, Sierra and Catron Counties, NM, Due: February 28, 1994, Contact: Russell Ward (505) 536–2250.

Amended Notices

EIS No. 930395, Draft EIS, TVA, KY, TN, Land between the Lakes (LBL) Natural Resource Management Plan, Implementation, KY and TN, Due: January 10, 1994, Contact: Dale K. Fowler (614) 632–4223. Published FR—11–12–93—Due Date Correction.

Dated: November 22, 1993.

Richard E. Sanderson.

Director, Office of Federal Activities.

[FR Doc. 93-29012 Filed 11-24-93; 8:45 am]

BILLING CODE 6560-50-U-M

[OPPTS-00144; FRL-4745-7]

Biotechnology; Discussion of Agency's Current Thinking on Biotechnology Issues; Open Meeting

AGENCY: Environmental Protection Agency (EPA).

ACTION: Notice of open meeting.

SUMMARY: EPA's Office of Prevention, Pesticides and Toxic Substances will hold a 1-day meeting to discuss biotechnology issues. This will be an information-gathering meeting wherein the views of individual participants at the meeting will be collected. EPA anticipates that during 1994, it will publish two proposals dealing with regulation of biotechnology products: A proposal addressing microorganisms under the Toxic Substances Control Act (TSCA), and a proposal dealing with pesticidal substances produced by plants under the Federal Insecticide, Fungicide, and Rodenticide Act (FIFRA) and the Federal Food, Drug, and Cosmetic Act (FFDCA). Agency staff will briefly outline EPA's current thinking on these two proposals during the first quarter of the meeting.

DATES: The meeting will be held on Thursday, December 16, 1993, starting at 2 p.m. and ending at 5 p.m.

ADDRESSES: The meeting will be held at the Environmental Protection Agency, 401 M St., SW., Rm. 642 East Tower, Washington, DC.

FOR FURTHER INFORMATION CONTACT: Ms. Creavery Lloyd, Science and Policy Staff (MC–7100), Office of Prevention, Pesticides and Toxic Substances, Environmental Protection Agency, 401 M St., SW., Rm. 627B East Tower, Washington, DC 20460, Telephone: (202) 260–6900.

SUPPLEMENTARY INFORMATION: Attendance by the public will be limited to available space. Persons interested in attending the meeting should contact Creavery Lloyd at 202–260–6900 or send a FAX to 202–260–0951 no later than Wednesday, December 8, 1993.

Dated: November 19, 1993.

Lvnn R. Goldman,

Assistant Administrator for Prevention, Pesticides and Toxic Substances.

[FR Doc. 93–29016 Filed 11–24–93; 8:45 am] **BILLING CODE 6560–50–F**

[FRL-4807-1]

Science Advisory Board; Notification of Public Advisory Committee Meeting

SUMMARY: Pursuant to the Federal Advisory Committee Act, Public Law 92-463, notice is hereby given that the Science Advisory Board's (SAB) ad hoc Industrial Excess Landfill (IEL) Advisory Panel will meet on December 14, 1993 at the Uniontown United Methodist Church, 13370 Cleveland Avenue, NW., Uniontown, Ohio 44685. The meeting will begin at 9 a.m. and adjourn no later than 5 p.m. This is the third meeting of the IEL Panel. Information relevant to the history of the request to the SAB and the charge to the Panel are contained in a previous **Federal** Register notice (58 FR (129):36686–36687, dated 8 July 1993). The primary purpose of this meeting is for the Panel to discuss the materials they have received concerning the IEL site and to formulate responses to the questions outlined in the Charge to the Panel. There will be a brief public comment period. The meeting is open to the public, however, due to limited space, seating will be on a first-come basis.

FOR FURTHER INFORMATION CONTACT: For details concerning this review, including a draft meeting agenda and Panel roster, or the overall activities of this Panel or the Science Advisory Board, please contact the Designated Federal Official for this review: Mr. Robert Flaak, Assistant Staff Director, Science Advisory Board (1400F), U.S. Environmental Protection Agency, 401 M Street, SW., Washington, DC 20460. Telephone: (202) 260–6552 and FAX: (202) 260–7118. Members of the public who wish to make a brief oral presentation to the Panel must contact Mr. Flaak in writing (by letter or by fax) no later than 12 noon Eastern Time, Friday, December 10, 1993 in order to be included on the Agenda.

PROVIDING ORAL OR WRITTEN COMMENTS:

The Science Advisory Board expects that public statements presented at its meetings will not be repetitive of previously submitted oral or written statements. In general, each individual or group making an oral presentation will be limited to a total time of five minutes. Written comments (at least 35 copies) received in the SAB Staff Office sufficiently prior to the meeting date may be mailed to the Panel prior to its meeting; comments received too close to the meeting date will normally be provided to the Panel at its meeting. Written comments may be provided to the Panel up until the time of the meeting.

Dated: November 18, 1993.

A. Robert Flaak.

Acting Staff Director, Science Advisory Board. [FR Doc. 93–29050 Filed 11–24–93; 8:45 am]

BILLING CODE 6560-50-P

[OPP-60049; FRL-4741-2]

Intent to Suspend Certain Pesticide Registrations

AGENCY: Environmental Protection Agency (EPA).

ACTION: Notice of issuance of notices of intent to suspend.

SUMMARY: This Notice, pursuant to section 6(f)(2) of the Federal Insecticide, Fungicide, and Rodenticide Act (FIFRA), 7 U.S.C. 136 et seg., announces that EPA has issued Notices of Intent to Suspend pursuant to sections 3(c)(2)(B) and 4 of FIFRA. The Notices were issued following issuance of Section 4 Reregistration Requirements Notices by the Agency and the failure of registrants subject to the Section 4 Reregistration Requirements Notices to take appropriate steps to secure the data required to be submitted to the Agency. This Notice includes the text of a Notice of Intent to Suspend, absent specific chemical, product, or factual information. Table A of this Notice further identifies the registrants to whom the Notices of Intent to Suspend were issued, the date each Notice of Intent to Suspend was issued, the active ingredient(s) involved, and the EPA registration numbers and names of the registered product(s) which are affected by the Notices of Intent to Suspend. Moreover, Table B of this Notice identifies the basis upon which the Notices of Intent to Suspend were issued. Finally, matters pertaining to the timing of requests for hearing are specified in the Notices of Intent to Suspend and are governed by the deadlines specified in section 3(c)(2)(B). As required by section 6(f)(2), the Notices of Intent to Suspend were sent by certified mail, return receipt requested, to each affected registrant at its address of record.

FOR FURTHER INFORMATION CONTACT:

Stephen L. Brozena, Office of Compliance Monitoring (7204), Laboratory Data Integrity Assurance Division, Environmental Protection Agency, 401 M St., SW., Washington, DC 20460, (703) 308–8267.

SUPPLEMENTARY INFORMATION:

I. Text of a Notice of Intent to Suspend

The text of a Notice of Intent to Suspend, absent specific chemical, product, or factual information, follows:

United States Environmental Protection Agency

Office of Prevention, Pesticides and Toxic Substances

Washington, DC 20460

Certified Mail

Return Receipt Requested

SUBJECT: Suspension of Registration of Pesticide Product(s) Containing ______ for Failure to Comply with the Section 4 Phase 5 Reregistration Eligibility Document Data Call-In Notice for ______ Dated

Dear Sir/Madam:

This letter gives you notice that the pesticide product registrations listed in Attachment I will be suspended 30 days from your receipt of this letter unless you take steps within that time to prevent this Notice from automatically becoming a final and effective order of suspension. The Agency's authority for suspending the registrations of your products is sections 3(c)(2)(B) of the Federal Insecticide, Fungicide, and Rodenticide Act (FIFRA). Upon becoming a final and effective order of suspension, any violation of the order will be an unlawful act under section 12(a)(2)(J) of FIFRA.

You are receiving this Notice of Intent to Suspend because you have failed to comply with the terms of the Phase 5 Registration Eligibility Document Data Call-In Notice imposed pursuant to section 4(g)(2)(b) and section (3)(2)(B) of FIFRA.

The specific basis for issuance of this Notice is stated in the Explanatory Appendix (Attachment III) to this Notice. Affected products and the requirements which you failed to satisfy are listed and described in the following three attachments:

Attachment I Suspension Report - Product List

Attachment II Suspension Report - Requirement List

Attachment III Suspension Report -Explanatory Appendix

The suspension of the registration of each product listed in Attachment I will become final unless at least one of the following actions is completed.

1. You may avoid suspension under this Notice if you or another person adversely affected by this Notice properly request a hearing within 30 days of your receipt of this Notice. If you request a hearing, it will be conducted in accordance with the requirements of section 6(d) of FIFRA and the Agency's procedural regulations in 40 CFR part 164.

Section 3(c)(2)(B), however, provides that the only allowable issues which may be addressed at the hearing are whether you have failed to take the actions which are the bases of this Notice and whether the Agency's decision regarding the disposition of existing stocks is consistent with FIFRA. Therefore, no substantive allegation or legal argument concerning other issues, including but not limited to the Agency's original decision to

require the submission of data or other information, the need for or utility of any of the required data or other information or deadlines imposed, and the risks and benefits associated with continued registration of the affected product, may be considered in the proceeding. The Administrative Law Judge shall by order dismiss any objections which have no bearing on the allowable issues which may be considered in the proceeding.

Section 3(c)(2)(B)(iv) of FIFRA provides that any hearing must be held and a determination issued within 75 days after receipt of a hearing request. This 75–day period may not be extended unless all parties in the proceeding stipulate to such an extension. If a hearing is properly requested, the Agency will issue a final order at the conclusion of the hearing governing the suspension of your products.

A request for a hearing pursuant to this Notice must (1) include specific objections which pertain to the allowable issues which may be heard at the hearing, (2) identify the registrations for which a hearing is requested, and (3) set forth all necessary supporting facts pertaining to any of the objections which you have identified in your request for a hearing. If a hearing is requested by any person other than the registrant, that person must also state specifically why he asserts that he would be adversely affected by the suspension action described in this Notice. Three copies of the request must be submitted to: Hearing Clerk, 1900, U.S. Environmental Protection Agency, 401 M St., SW., Washington, DC 20460, and an additional copy should be sent to the signatory listed below. The request must be received by the Hearing Clerk by the 30th day from your receipt of this Notice in order to be legally effective. The 30-day time limit is established by FIFRA and cannot be extended for any reason. Failure to meet the 30-day time limit will result in automatic suspension of your registration(s) by operation of law and, under such circumstances, the suspension of the registration for your affected product(s) will be final and effective at the close of business 30 days after your receipt of this Notice and will not be subject to further administrative review.

The Agency's Rules of Practice at 40 CFR 164.7 forbid anyone who may take part in deciding this case, at any stage of the proceeding, from discussing the merits of the proceeding *ex parte* with any party or with any person who has been connected with the preparation or presentation of the proceeding as an advocate or in any investigative or expert capacity, or with any of their

representatives. Accordingly, the following EPA offices, and the staffs thereof, are designated as judicial staff to perform the judicial function of EPA in any administrative hearings on this Notice of Intent to Suspend: The Office of the Administrative Law Judges, the Office of the Judicial Officer, the Administrator, the Deputy Administrator, and the members of the staff in the immediate offices of the Administrator and Deputy Administrator. None of the persons designated as the judicial staff shall have any ex parte communication with trial staff or any other interested person not employed by EPA on the merits of any of the issues involved in this proceeding, without fully complying with the applicable regulations.

2. You may also avoid suspension if, within 30 days of your receipt of this Notice, the Agency determines that you have taken appropriate steps to comply with the section 4 Phase 5 Reregistration Eligibility Document Data Call-In Notice requirements. In order to avoid suspension under this option, you must satisfactorily comply with Attachment II, Requirement List, for each product by submitting all required supporting data/information described in Attachment II and in the Explanatory Appendix (Attachment III) to the following address (preferably by certified mail):

Office of Compliance Monitoring (7204), Laboratory Data Integrity Assurance Division, U.S. Environmental Protection Agency, 401 M St., SW., Washington, DC 20460.

For you to avoid automatic suspension under this Notice, the Agency must also determine within the applicable 30–day period that you have satisfied the requirements that are the bases of this Notice and so notify you in writing. You should submit the necessary data/information as quickly as possible for there to be any chance the Agency will be able to make the necessary determination in time to avoid suspension of your product(s).

The suspension of the registration(s) of your company's product(s) pursuant to this Notice will be rescinded when the Agency determines you have complied fully with the requirements which were the bases of this Notice. Such compliance may only be achieved by submission of the data/information described in the attachments to the signatory below.

Your product will remain suspended, however, until the Agency determines you are in compliance with the requirements which are the bases of this Notice and so informs you in writing. After the suspension becomes final and effective, the registrant subject to this Notice, including all supplemental registrants of product(s) listed in Attachment I, may not legally distribute, sell, use, offer for sale, hold for sale, ship, deliver for shipment, or receive and (having so received) deliver or offer to deliver, to any person, the product(s) listed in Attachment I.

Persons other than the registrant subject to this Notice, as defined in the preceding sentence, may continue to distribute, sell, use, offer for sale, hold for sale, ship, deliver for shipment, or receive and (having so received) deliver or offer to deliver, to any person, the product(s) listed in Attachment I.

Nothing in this Notice authorizes any person to distribute, sell, use, offer for sale, hold for sale, ship, deliver for shipment, or receive and (having so received) deliver or offer to deliver, to any person, the product(s) listed in Attachment I in any manner which would have been unlawful prior to the suspension.

If the registrations of your products listed in Attachment I are currently suspended as a result of failure to comply with another section 4 Data Requirements Notice or section 3(c)(2)(B) Data Call-In Notice, this Notice, when it becomes a final and effective order of suspension, will be in addition to any existing suspension, i.e., all requirements which are the bases of the suspension must be satisfied before the registration will be reinstated.

You are reminded that it is your responsibility as the basic registrant to notify all supplementary registered distributors of your basic registered product that this suspension action also applies to their supplementary registered products and that you may be held liable for violations committed by your distributors. If you have any questions about the requirements and procedures set forth in this suspension notice or in the subject section 4 Data Requirements Notice, please contact Stephen L. Brozena at (703) 308–8267.

Sincerely yours,

Director, Office of Compliance Monitoring Attachments:

Attachment I - Product List Attachment II - Requirement List Attachment III - Explanatory Appendix

II. Registrants Receiving and Affected by Notices of Intent to Suspend; Date of Issuance; Active Ingredient and Products Affected

The following is a list of products for which a letter of notification has been sent:

TABLE A.—LIST OF PRODUCTS

Registrant Affected	EPA Registration Number	Active Ingredient	Name of Product	Date Issued
Air Products and Chemicals	02847200009	Ethylene	Ethylene	10/8/93

TABLE A.—LIST OF PRODUCTS—Continued

Registrant Affected	EPA Registration Number	Active Ingredient	Name of Product	Date Issued
Angus Chemical Corporation	01136400005	Sodium Hydroxide	Angus Hot Rod	10/8/93
Brooker Chemical Corporation	00828100001	Indole-3-Butyric Acid	Hormex Rooting Powder No. 8	10/8/93
	00828100002	Indole-3-Butyric Acid	Hormex Rooting Powder No. 16	10/8/93
	00828100003	Indole-3-Butyric Acid	Hormex Rooting Powder No. 3	10/8/93
	00828100004	Indole-3-Butyric Acid	Hormex Rooting Powder No. 30	10/8/93
	00828100005 00828100006	Indole-3-Butyric Acid Indole-3-Butyric Acid	Hormex Rooting Powder No. 45 Hormex Rooting Powder No. 1	10/8/93 10/8/93
	00828100007	Indole-3-Butyric Acid	Hormex	10/8/93
Cotey Chemical Company	00094290002	Sodium Hydroxide	Welgicide Cleaner	10/8/93
Earth Science Products Corporation	04390500001	Indole-3-Butyric Acid	Wood's Rooting Compound	10/8/93
Green Light Company	00086900060	Indole-3-Butyric Acid	Green Light Root Stimulator and Starter Solution	10/8/93
Hortus USA Corporation	06331000008	Indole-3-Butyric Acid	Rhizopon AA Water Soluble Tablets	10/8/93
Iron Out, Inc.	00099020001	Sodium Hydroxide	Klean Em All Root Destroyer	10/8/93
National Research & Chemical Company	04219800006	Indole-3-Butyric Acid	Grow More Rapid Root	10/8/93
Plantabbs Corp.	00533200010	Indole-3-Butyric Acid	ole-3-Butyric Acid Plantabbs Rooting Powder	
Riverside/Terra Corporation	00977900313	Indole-3-Butyric Acid	Crop Booster Plus 12-8-8	10/8/93
	00977900321	Indole-3-Butyric Acid	Maxon II	10/8/93
Rockland Corporation	00057200333	Indole-3-Butyric Acid	Hormo-Root "B"	10/8/93
	00057200334	Indole-3-Butyric Acid	Hormo-Root "C"	10/8/93
	00057200335	Indole-3-Butyric Acid	Hormo-Root "A"	10/8/93
	00057200336 00057200337	Indole-3-Butyric Acid	Hormo-Root 4 Hormo-Root 1	10/8/93 10/8/93
	00057200337	Indole-3-Butyric Acid Indole-3-Butyric Acid	Hormo-Root 2; Hormo-Root 3	10/8/93
	00057200339	Indole-3-Butyric Acid	Hormo-Root 3	10/8/93
Rooto Corp.	00813200003	Sodium Hydroxide	Rooto No. 2	10/8/93
	00828100003	Indole-3-Butyric Acid	Hormex Rooting Powder No. 3	10/8/93
Syntex, S.A. DE C.V. Division Quimica	04247300001	Indole-3-Butyric Acid	Indole-3-Butyric Acid	10/8/93
Voluntary Purchasing Group, Inc.	00740100343	Indole-3-Butyric Acid	Ferti-Lome Rooting Powder	10/8/93

III. Basis for Issuance of Notice of Intent; Requirement List

The following companies failed to submit the following required data or information:

TABLE B.—REQUIREMENT LIST

Active Ingredient	Registrant Affected	Requirement Name	Guideline Ref- erence No.	Original Due-Date
Indole-3-Butyric Acid	Riverside/Terra Corporation	Product Identity	151-10	5/10/93
•	·	Manufacturing Process	151-11	5/10/93
		Discussion of Formation of Impurities	151-12	5/10/93
		Analysis of Samples	151-13	5/10/93
		Certification of Limits	151-15	5/10/93
		Analytical Methods	151-16	5/10/93
		Physical and Chemical Properties	151-17	5/10/93
		Acute Oral Toxicity - Tier I	152-10	5/10/93
		Acute Dermal Toxicity - Tier I	152-11	5/10/93
		Acute Inhalation - Tier I	152-12	5/10/93
		Primary Eye Irritation - Tier I	152-13	5/10/93
		Primary Dermal Irritation - Tier I	152-14	5/10/93
		Hypersensitivity Study - Tier I	152-15	5/10/93
		Hypersenstivity Incidents - Tier I	152-16	5/10/93
		Confidential Statement of Formula (CSF) Form		5/10/93
		Certification Form: Data Citation		5/10/93
		Certification Form: Offer to Cost Share		5/10/93
		Certification Form: Data Compensation		5/10/93
	Green Light Company	Product Identity	151-10	5/10/93
	J. Son Eight Company	Manufacturing Process	151-11	5/10/93
		Discussion of Formation of Impurities	151-12	5/10/93
		Analysis of Samples	151-13	5/10/93

TABLE B.—REQUIREMENT LIST—Continued

Active Ingredient	Registrant Affected	Requirement Name	Guideline Ref- erence No.	Original Due-Date
		Certification of Limits Analytical Methods Physical and Chemical Properties	151-15 151-16 151-17	5/10/93 5/10/93 5/10/93
		Acute Oral Toxicity - Tier I	152-10	5/10/93
		Acute Dermal Toxicity - Tier I Acute Inhalation - Tier I	152-11 152-12	5/10/93 5/10/93
		Primary Eye Irritation - Tier I	152-13	5/10/93
		Primary Dermal Irritation - Tier I	152-14	5/10/93
		Hypersensitivity Study - Tier I Hypersenstivity Incidents - Tier I	152-15 152-16	5/10/93 5/10/93
		Confidential Statement of Formula (CSF)	132-10	5/10/93
		Certification Form: Data Citation		5/10/93
		Certification Form: Offer to Cost Share		5/10/93
	B	Certification Form: Data Compensation	454.40	5/10/93
	Plantabbs Corp.	Product Identity Manufacturing Process	151-10 151-11	5/10/93 5/10/93
		Discussion of Formation of Impurities	151-11	5/10/93
		Analysis of Samples	151-13	5/10/93
		Certification of Limits	151-15	5/10/93
		Analytical Methods Physical and Chemical Properties	151-16 151-17	5/10/93 5/10/93
		Acute Oral Toxicity - Tier I	152-10	5/10/93
		Acute Dermal Toxicity - Tier I	152-11	5/10/93
		Acute Inhalation - Tier I	152-12	5/10/93
		Primary Eye Irritation - Tier I Primary Dermal Irritation - Tier I	152-13 152-14	5/10/93 5/10/93
		Hypersensitivity Study - Tier I	152-15	5/10/93
		Hypersenstivity Incidents - Tier I	152-16	5/10/93
		Confidential Statement of Formula (CSF) Form		5/10/93
		Certification Form: Data Citation		5/10/93
		Certification Form: Offer to Cost Share Certification Form: Data Compensation		5/10/93 5/10/93
	Brooker Chemical Corpora-	Product Identity	151-10	5/10/93
	tion	 Manufacturing Process	151-11	5/10/93
		Discussion of Formation of Impurities	151-12	5/10/93
		Analysis of Samples Certification of Limits	151-13 151-15	5/10/93 5/10/93
		Analytical Methods	151-16	5/10/93
		Physical and Chemical Properties	151-17	5/10/93
		Acute Oral Toxicity - Tier I	152-10	5/10/93
		Acute Dermal Toxicity - Tier I	152-11 152-12	5/10/93 5/10/93
		Acute Inhalation - Tier I Primary Eye Irritation - Tier I	152-12	5/10/93
		Primary Dermal Irritation - Tier I	152-14	5/10/93
		Hypersensitivity Study - Tier I	152-15	5/10/93
		Hypersenstivity Incidents - Tier I Confidential Statement of Formula (CSF)	152-16	5/10/93 5/10/93
		Form Certification Form: Data Citation		5/10/93
		Certification Form: Offer to Cost Share		5/10/93
		Certification Form: Data Compensation		5/10/93
	Rockland Corporation	Confidential Statement of Formula (CSF) Form		5/10/93
		Certification Form: Data Citation		5/10/93
		Certification Form: Offer to Cost Share		5/10/93
		Certification Form: Data Compensation Acute Oral Toxicity - Tier I	152-10	5/10/93 5/10/93
		Acute Dermal Toxicity - Tier I	152-11	5/10/93
		Acute Inhalation - Tier I	152-12	5/10/93
		Primary Eye Irritation - Tier I	152-13 152-14	5/10/93 5/10/93
		Primary Dermal Irritation - Tier I Hypersenstivity Incidents - Tier I	152-14	5/10/93
	National Research & Chemical Company	Confidential Statement of Formula (CSF)		5/10/93
	ical Company	Form Certification Form: Data Citation		5/10/93
		Certification Form: Offer to Cost Share Certification Form: Data Compensation		5/10/93 5/10/93

TABLE B.—REQUIREMENT LIST—Continued

Active Ingredient	Registrant Affected	Requirement Name	Guideline Ref- erence No.	Original Due-Date
	Syntex, S.A. De C.V. Division Quimica	Confidential Statement of Formula (CSF)		5/10/93
	sion Quimica	Form Certification Form: Data Citation		5/10/93
		Certification Form: Offer to Cost Share		5/10/93
		Certification Form: Data Compensation		5/10/93
		Product Identity	151-10	5/10/93
		Manufacturing Process	151-11	5/10/93
		Discussion of Formation of Impurities	151-12	5/10/93
		Analysis of Samples	151-13	5/10/93
		Certification of Limits Analytical Methods	151-15 151-16	5/10/93 5/10/93
		Physical and Chemical Properties	151-10	5/10/93
		Primary Eye Irritation - Tier I	152-13	5/10/93
		Primary Dermal Irritation - Tier I	152-14	5/10/93
	Earth Science Products	Confidential Statement of Formula (CSF)		5/10/93
	Corporation	Form Certification Form: Data Citation		5/10/93
		Certification Form: Offer to Cost Share		5/10/93
		Certification Form: Data Compensation		5/10/93
		Product Identity	151-10	5/10/93
		Manufacturing Process	151-11	5/10/93
		Discussion of Formation of Impurities	151-12	5/10/93
		Analysis of Samples	151-13	5/10/93
		Certification of Limits	151-15	5/10/93
		Analytical Methods Physical and Chemical Properties	151-16 151-17	5/10/93 5/10/93
		Acute Oral Toxicity - Tier I	152-10	5/10/93
		Acute Oral Toxicity - Tier I	152-10	5/10/93
		Acute Inhalation - Tier I	152-12	5/10/93
		Primary Eye Irritation - Tier I	152-13	5/10/93
		Primary Dermal Irritation - Tier I	152-14	5/10/93
		Hypersensitivity Study - Tier I	152-15	5/10/93
		Hypersenstivity Incidents - Tier I	152-16	5/10/93
	Hortus USA Corporation	Confidential Statement of Formula (CSF) Form		5/10/93
		Certification Form: Data Citation		5/10/93
		Certification Form: Offer to Cost Share		5/10/93
		Certification Form: Data Compensation		5/10/93
		Product Identity	151-10	5/10/93
		Certification of Limits	151-15	5/10/93
		Analytical Methods	151-16	5/10/93
		Physical and Chemical Properties	151-17	5/10/93 5/10/93
		Acute Oral Toxicity - Tier I Acute Dermal Toxicity - Tier I	152-10 152-11	5/10/93
		Acute Inhalation - Tier I	152-11	5/10/93
		Primary Eye Irritation - Tier I	152-13	5/10/93
		Primary Dermal Irritation - Tier I	152-14	5/10/93
		Hypersenstivity Incidents - Tier I	152-16	5/10/93
	Voluntary Purchasing	Product Identity	151-10	5/10/93
	Group, Inc.	Manufacturing Process	151-11	5/10/93
		Discussion of Formation of Impurities	151-12	5/10/93
		Analysis of Samples	151-13	5/10/93
		Certification of Limits	151-15	5/10/93
		Analytical Methods	151-16	5/10/93
		Physical and Chemical Properties	151-17	5/10/93
		Acute Oral Toxicity - Tier I	152-10	5/10/93
		Acute Dermal Toxicity - Tier I	152-11	5/10/93
		Acute Inhalation - Tier I Primary Eye Irritation - Tier I	152-12 152-13	5/10/93 5/10/93
		Primary Eye imation - Her I	152-13	5/10/93
		Hypersensitivity Study - Tier I	152-14	5/10/93
		Hypersenstivity Incidents - Tier I	152-16	5/10/93
		Confidential Statement of Formula (CSF)		5/10/93
		Form Certification Form: Data Citation		5/10/93
		Certification Form: Offer to Cost Share		5/10/93
		Certification Form: Data Compensation		5/10/93
dium Hydroxide	Cotey Chemical Company	Confidential Statement of Formula (CSF)		5/30/93
	Colo, Chomical Company	Form		

TABLE B.—REQUIREMENT LIST—Continued

Active Ingredient	Registrant Affected	Requirement Name	Guideline Ref- erence No.	Original Due-Date
		Chemical Identity Beginning Materials & Manufacturing Process	61-1 61-2	5/30/9 5/30/9
		Discussion of Formation of Impurities	61-3	5/30/9
		Preliminary Analysis of Product Samples	62-1	5/30/9
		Certification of Ingredient Limits	62-2	5/30/9
		Analytical Method to Verify Certified Limits	62-3	5/30/9
		Odor	63-4	5/30/9
		Melting Point	63-5 63-6	5/30/9
		Boiling Point Density, Bulk Density, or Specific Gravity	63-7	5/30/9 5/30/9
		Solubility Solubility	63-8	5/30/9
		Vapor Pressure	63-9	5/30/9
		Dissociation Constant	63-10	5/30/9
		Octanol/Water Partition Coefficient	63-11	5/30/9
		pH	63-12	5/30/9
		Stability	63-13	5/30/9
		Oxidizing/Reducing Action	63-14	5/30/9
		Flammability Explodability	63-15 63-16	5/30/9 5/30/9
		Storage Stability	63-17	5/30/9
		Viscosity	63-17	5/30/9
		Miscibility	63-19	5/30/9
		Corrosion Characteristics	63-20	5/30/9
		Dielectric Breakdown Voltage	63-21	5/30/9
		Certification Form: Data Citation		5/30/9
		Certification Form: Offer to Cost Share		5/30/9 5/30/9
	Iron Out, Inc.	Certification Form: Data Compensation Confidential Statement of Formula (CSF)		5/30/9
		Form		- /o o /o
		Chemical Identity	61–1	5/30/9
		Beginning Materials & Manufacturing Process	61-2	5/30/9
		Discussion of Formation of Impurities	61-3	5/30/9
		Preliminary Analysis of Product Samples Certification of Ingredient Limits	62-1 62-2	5/30/9 5/30/9
		Analytical Method to Verify Certified Limits	62-3	5/30/9
		Odor	63-4	5/30/9
		Melting Point	63-5	5/30/9
		Boiling Point	63-6	5/30/9
		Density, Bulk Density, or Specific Gravity	63-7	5/30/9
		Solubility	63-8	5/30/9 5/30/9
		Vapor Pressure Dissociation Constant	63-9 63-10	5/30/9
		Octanol/Water Partition Coefficient	63-10	5/30/9
		pH	63-12	5/30/9
		Stability	63-13	5/30/9
		Oxidizing/Reducing Action	63-14	5/30/9
		Flammability	63-15	5/30/9
		Explodability	63-16	5/30/9
		Storage Stability Viscosity	63-17 63-18	5/30/9 5/30/9
		Miscibility	63-19	5/30/9
		Corrosion Characteristics	63-20	5/30/9
		Dielectric Breakdown Voltage	63-21	5/30/9
		Certification Form: Data Citation		5/30/9
		Certification Form: Offer to Cost Share		5/30/9
	Angus Chemical Corpora-	Certification Form: Data Compensation Confidential Statement of Formula (CSF)		5/30/9 5/30/9
	tion	Form		5,5373
		Chemical Identity Beginning Materials & Manufacturing	61-1 61-2	5/30/9 5/30/9
		Process Discussion of Formation of Impurities	61-3	5/30/9
		Preliminary Analysis of Product Samples	62-1	5/30/9
		Certification of Ingredient Limits	62-2	5/30/9
		Analytical Method to Verify Certified Limits	62-3	5/30/9
		Odor	63-4	5/30/9
		Melting Point	63-5	5/30/9
	1	Boiling Point	63-6	5/30/9
		Density, Bulk Density, or Specific Gravity	63-7	5/30/9

TABLE B.—REQUIREMENT LIST—Continued

Active Ingredient	Registrant Affected	Requirement Name	Guideline Ref- erence No.	Original Due-Date
		Solubility	63-8	5/30/93
		Vapor Pressure	63-9	5/30/93
		Dissociation Constant	63-10	5/30/93
		Octanol/Water Partition Coefficient	63-11	5/30/93
		pH	63-12	5/30/93
		Stability	63-13	5/30/93
		Oxidizing/Reducing Action	63-14	5/30/93
		Flammability	63-15	5/30/93
			63-16	5/30/93
		Explodability	63-17	
		Storage Stability		5/30/93
		Viscosity	63-18	5/30/93
		Miscibility	63-19	5/30/93
		Corrosion Characteristics	63-20	5/30/93
		Dielectric Breakdown Voltage	63-21	5/30/93
		Certification Form: Data Citation		5/30/93
		Certification Form: Offer to Cost Share		5/30/93
		Certification Form: Data Compensation		5/30/93
	Rooto Corp.	Confidential Statement of Formula (CSF) Form		5/30/93
		Chemical Identity	61-1	5/30/93
		Beginning Materials & Manufacturing	61-2	5/30/93
		Process		= /0.0/0.0
		Discussion of Formation of Impurities	61-3	5/30/93
		Preliminary Analysis of Product Samples	62-1	5/30/93
		Certification of Ingredient Limits	62-2	5/30/93
		Analytical Method to Verify Certified Limits	62-3	5/30/93
		Odor	63-4	5/30/93
		Melting Point	63-5	5/30/93
		Boiling Point	63-6	5/30/93
		Density, Bulk Density, or Specific Gravity	63-7	5/30/93
		Solubility	63-8	5/30/93
		Vapor Pressure	63-9	5/30/93
		Dissociation Constant	63-10	5/30/93
		Octanol/Water Partition Coefficient	63-10	5/30/93
		pH	63-12	5/30/93
		Stability	63-13	5/30/93
		Oxidizing/Reducing Action	63-14	5/30/93
		Flammability	63-15	5/30/93
		Explodability	63-16	5/30/93
		Storage Stability	63-17	5/30/93
		Viscosity	63-18	5/30/93
		Miscibility	63-19	5/30/93
		Corrosion Characteristics	63-20	5/30/93
		Dielectric Breakdown Voltage	63-21	5/30/93
		Certification Form: Data Citation		5/30/93
		Certification Form: Offer to Cost Share		5/30/93
		Certification Form: Data Compensation		5/30/93
Ethylene	Air Products and Chemicals, Inc.	Certification Form: Data Citation		5/28/93
	IIIO.	Certification Form: Offer to Cost Share		5/28/93
		Certification Form: Data Compensation		5/28/93
		Product Identity	151-10	5/28/93
		Manufacturing Process	151-11	5/28/93
		Discussion of Formation of Impurities	151-12	5/28/93
		Analysis of Samples	151-12	5/28/93
		Certification of Limits	151-15	5/28/93
		Analytical Methods	151-16	5/28/93
		Physical and Chemical Properties	151-17	5/28/93

IV. Attachment III Suspension Report— Explanatory Appendix

A discussion of the basis for the Notice of Intent to Suspend follows:

A. Ethylene

On September 30, 1992, EPA issued the Phase 5 Reregistration Data Requirements

Notice imposed pursuant to section 4 of FIFRA which required registrants of products containing ethylene to develop and submit certain data/information. These data/information were determined to be necessary to satisfy reregistration data requirements of section 4(g)(2)(B) of FIFRA.

The Ethylene Phase 5 Reregistration Data Requirements Notice dated September 30, 1992, required each affected registrant to submit materials relating to the election of the options to address each of the data requirements. The Notice further required that data/information be submitted by deadlines noted for the subject data/information

requirements on Attachment II. These deadlines have passed and to date the Agency has not received adequate data/information to satisfy these data/information requirements. Because you have failed to provide an appropriate or adequate response within the time provided for data/information requirements listed on Attachment II, the Agency is issuing this Notice of Intent to Suspend.

B. Indole-3-Butyric Acid

On September 10, 1992, EPA issued the Phase 5 Reregistration Data Requirements Notice imposed pursuant to section 4 of FIFRA which required registrants of products containing indole-3-butyric acid to develop and submit certain data. These data were determined to be necessary to satisfy reregistration data requirements of section 4(g)(2)(B) of FIFRA.

The Indole-3-Butyric Acid Phase 5 Reregistration Data Requirements Notice dated September 10, 1992, required each affected registrant to submit materials relating to the election of the options to address each of the data requirements. That submission was required to be received by the Agency within 90 days of the registrant's receipt of the Notice. The Agency received a response from you in which you committed to undertake the required testing. The Notice further required that data be submitted by deadlines noted for the subject data requirements on Attachment II. These deadlines have passed and to date the Agency has not received adequate data to satisfy these data requirements. Because you have failed to provide an appropriate or adequate response within the time provided for data requirements listed on Attachment II, the Agency is issuing this Notice of Intent to Suspend.

C. Sodium Hydroxide

On September 30, 1992, EPA issued the Phase 5 Reregistration Data Requirements Notice imposed pursuant to section 4 of FIFRA which required registrants of products containing sodium hydroxide to develop and submit certain data. These data were determined to be necessary to satisfy reregistration data requirements of section 4(g)(2)(B) of FIFRA.

The Sodium Hydroxide Phase 5 Reregistration Data Requirements Notice dated September 30, 1992, required each affected registrant to submit materials relating to the election of the options to address each of the data requirements. That submission was required to be received by the Agency within 90 days of the registrant's receipt of the Notice. The Agency received a response from you in which you committed to undertake the required testing. The Notice further required that data be submitted by deadlines noted for the subject data requirements on Attachment II. These deadlines have passed and to date the Agency has not received adequate data to satisfy these data requirements. Because you have failed to provide an appropriate or adequate response within the time provided for data requirements listed on Attachment II, the Agency is issuing this Notice of Intent to Suspend.

V. Conclusions

EPA has issued Notices of Intent to Suspend on the dates indicated. Any further information regarding these Notices may be obtained from the contact person noted above.

List of Subjects

Environmental protection.

Dated: November 16, 1993.

Michael M. Stahl,

Director, Office of Compliance Monitoring. [FR Doc. 93–29018 Filed 11–24–93; 8:45 am] BILLING CODE 6560–50–F

[FRL-4806-4]

Beaunit/Circular Knit and Dying Plant Site, Notice of Proposed Settlement; Correction

AGENCY: Environmental Protection Agency. **ACTION:** Correction of the notice of proposed settlement.

SUMMARY: In notice document 93–26427 beginning on page 57828 in the issue of Wednesday, October 27, 1993, make the following correction:

On page 57828 in the first column, the city and state of the Beaunit Circular Knit and

Dying Plant Site was previously listed as Baldwin, Florida. This should be changed to read Fountain Inn, South Carolina.

EPA will consider public comments on the proposed settlement for thirty (30) days from the date of the publication of this Notice Correction

Dated: November 10, 1993.

Richard D. Green.

Acting Director, Waste Management Division. [FR Doc. 93–29044 Filed 11–24–93; 8:45 am]

BILLING CODE 6560-50-M

[FRL-4906-5]

Rock Hill Chemical Company Site, Rock Hill, SC; Notice of Proposed Settlement

AGENCY: Environmental Protection Agency. **ACTION:** Notice of proposed settlement.

SUMMARY: Under section 122(h) of the Comprehensive Environmental Response, Compensation and Liability Act of 1980 (CERCLA), as amended 42 U.S.C. 9601 et seq., the Environmental Protection Agency (EPA) has agreed to a partial, de minimis settlement with the First Union National Bank of South Carolina for claims of response costs at the Rock Hill Chemical Company Site. EPA will consider public comments on the proposed settlement for thirty (30) days. EPA may withdraw from or modify the proposed settlement should such comments disclose facts or considerations which indicate the proposed settlement is inappropriate, improper, or inadequate. Copies of the proposed settlement are available from: Ms. Carolyn McCall, Waste Programs Branch, Waste Management DIvision, U.S. EPA, Region IV, 345 Courtland Street, NE., Atlanta, Georgia 30365, 404-347-5059.

Written comments must be submitted to the person above by thirty days from the date of publication.

Dated: November 10, 1993.

Richard D. Green,

Acting Director, Waste Management Division. [FR Doc. 93–29043 Filed 11–24–93; 8:45 am]

BILLING CODE 6560-50-M